U.S. SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

Form 10-QSB

(Mark One)

[X] QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended June 30, 2004

[_] TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from _____ to _____

Commission file number 0-21467

ACCESSITY CORP.

(Name of small business issuer in its charter)

New York

(State or other jurisdiction of incorporation or organization)

12514 West Atlantic Boulevard Coral Springs, Florida 33071

(Address of principal executive offices)

11-2750412

(IRS Employer Identification No.)

(954-752-6161)

(Issuer's telephone number)

Securities registered under Section 12(b) of the Exchange Act: None

Securities registered under Section 12(g) of the Exchange Act: Common Stock par value \$.015 per share Preferred Stock Purchase Rights par value \$.01 per share

Check whether the issuer (1) has filed all reports required to be filed by Section 13 or 15(d) of the Exchange Act during the past 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes[X] No[_]

As of August 10, 2004 the issuer had outstanding a total of 2,237,414 shares of common stock.

Transitional Small Business Format (check one) Yes [_] No [X]

ACCESSITY CORP.

CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

SIX MONTHS ENDED JUNE 30, 2004

CONTENTS

PAGE

Part I. FINANCIAL INFORMATION

Item 1. Financial Statements

Condensed Consolidated Balance Sheet As of June 30, 2004 (Unaudited) 3

Condensed Consolidated Statements of Operations and Comprehensive Loss (Unaudited) for the

	Three Months ended June 30, 2004 and 2003	4
	Condensed Consolidated Statements of Operations and Comprehensive Loss (Unaudited) for the Six Months ended June 30, 2004 and 2003	5
	Condensed Consolidated Statements of Cash Flows (Unaudited) for the Six Months ended June 30, 2004 and 2003	6
	Notes to Condensed Consolidated Financial Statements	7
Item 2.	Management's Discussion and Analysis or Plan of Operation 1	6
Item 3.	Controls and Procedures 2	0
Part II.	OTHER INFORMATION 2	1
Item 6.	Exhibits and Reports on Form 8-K 2	:1
	Certifications 2	:3

Item 1. Financial Statements

	ACCESSITY CORP.						
CONDENSED	CONSOLIDATED BALANCE	SHEET					
	JUNE 30, 2004						
(UNAUDITED)							

ASSETS

Current assets: Cash and cash equivalents Accounts receivable Notes receivable (Note 5) Investments Prepaid expenses and other current assets	\$ 230,886 128,535 181,358 3,543,955 125,230
Total current assets	4,209,964
Property and equipment, net of accumulated depreciation Restricted certificate of deposit Security deposits and other assets	324,333 300,000 23,924
Total assets	\$ 4,858,221
LIABILITIES AND SHAREHOLDERS' EQUITY	
Current liabilities: Accounts payable Accrued expenses and other current liabilities Capital lease obligation	\$ 41,085 373,419 5,458
Total current liabilities, other than shares	419,962
Redeemable preferred stock, \$.01 par value, authorized 1,000,000 shares; 1,000 issued and outstanding; subject to mandatory redemption of \$350,000 (Note 4)	350,000
Total current liabilities	769,962
<pre>Shareholders' equity: Common stock, \$.015 par value, authorized 30,000,000 shares; issued 2,419,398 Additional paid-in capital Accumulated other comprehensive loss, unrealized holding loss on investment securities Deficit Less common stock held in treasury, at cost, 181,984 shares</pre>	36,291 10,751,188 (10,399) (4,959,379) 5,817,701 1,729,442
Total shareholders' equity	4,088,259

See notes to condensed consolidated financial statements.

3

ACCESSITY CORP. CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS AND COMPREHENSIVE LOSS (UNAUDITED)

<table><cap< th=""><th>TION></th><th>Three Mont</th><th>chs Ended,</th></cap<></table>	TION>	Three Mont	chs Ended,
		June 30 2004	June 30 2003
<s></s>			 <c></c>
Revenue:	Collision repairs, fees and royalties Hospital fees	\$ 54,657 144,580	\$ 49,006 62,732
	Total revenues	199,237	111,738
Operating e	xpenses:		
-1) -	Collision repair expenses		3,483
	Sales and marketing	116,644	109,892
	General and administrative	415,660	433,268
	Depreciation and amortization	59,636	85,123
	Total operating expenses	591,940	631 , 766
		(392,703)	(520,028)
Investment	and other income, net of interest expense (Note 5 and 11)	235,865	44,396
	continuing operations before provision for income taxes for income tax benefit	(156,838) 11,525	(475,632)
Loss from c	continuing operations	(145,313)	(475,632)
	d operations (Note 7):		17 220
11	come from affinity services subsidiary (no tax effect)		47,228
Net loss		(145,313)	(428,404)
Other compr	rehensive - unrealized (loss) on marketable securities (Note 11)	(10,399)	(1,275)
Comprehensi	ve loss	\$ (155,712) =========	\$ (429,679) ========
Basic and d	liluted earnings (loss) per common share:		
	Continuing operations	\$ (0.06)	\$ (0.22)
	Discontinued operations	0.00	0.02
	Total	\$ (0.06)	\$ (0.20) ======
-	verage number of common shares outstanding Bilutive securities, stock options and warrants	2,237,414	2,173,885
Basic and d	liluted weighted average number of common shares outstanding	2,237,414	2,173,885

</TABLE>

See notes to condensed consolidated financial statements.

4 ACCESSITY CORP. CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS AND COMPREHENSIVE LOSS (UNAUDITED)

<TABLE><CAPTION>

		Six Mon	ths Ended,
		June 30 2004	June 30 2003
<s></s>		 <c></c>	<c></c>
Revenue:	Collision repairs and fees Hospital fees	\$ 95,234 330,254 	\$ 190,266 62,732
	Total revenues	425,488	252,998

Operating expenses: Collision repair expenses		100,996
Sales and marketing	222,883	
General and administrative	941,026	248,969 930,352
Depreciation and amortization	118,556	169,385
Total operating expenses	1,282,465	1,449,702
	(856,977)	(1,196,704)
Investment and other income, net of interest expense (Note 5 and 11)	276,150	86,559
Loss from continuing operations before provision for income taxes	(580,827)	(1,110,145)
Provision for income (tax) benefit	11,525	
Loss from continuing operations		(1,110,145)
Discontinued operations (Note 7):		
Income from affinity services subsidiary		188,306
Net loss	(569,302)	(921,839)
Other comprehensive - unrealized (loss) on marketable securities (Note 11)		
Marketable Securities (Note 11)	(10, 399)	(20,957)
Comprehensive loss	\$ (579,701) ========	\$ (942,796)
Basic and diluted earnings (loss) per common share: Continuing operations	\$ (0.25)	\$ (0.51)
Discontinued operations	\$ (0.23) 0.00	\$ (0.31) 0.09
bisconclinaca operacions		
Total	\$ (0.25)	\$ (0.42)
Weighted average number of common shares outstanding Effect of dilutive securities, stock options and warrants	2,237,414	2,173,850
-		
Weighted average diluted common shares outstanding	2,237,414	2,173,850

		See notes to condensed consolidated financial statements.		
5				
ACCESSITY CORP.				
CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED)				
		Six Months E		

<table><caption></caption></table>	Six Months Ended			
	June 30 2004	June 30 2003		
<\$>	 <c></c>	 <c></c>		
Cash flows provided by (used in) operating activities:				
Net income (loss)	\$ (569,302)			
Adjustments to reconcile net income (loss) to net cash provided by (used in) operating activities: Depreciation and amortization (including bond premium				
amortization)	118 556	188,912		
Loss on sale of investments	11,669	,		
Impairment losses on marketable securities	72,750			
Options granted for services Changes in assets and liabilities:		8,955		
Accounts receivable	27,561	(81,198)		
Prepaid expenses and other assets	18,695	226,244		
Accounts payable	(11,112)	(128,166)		
Accrued expenses and other current liabilities	(45,480)	(464,695)		
Total adjustments		(235,029)		
Net cash provided by (used in) operating activities	(376,663)	(1,156,868)		
Cash flows provided by (used in) investing activities: Notes receivable Purchase of property and equipment Proceeds from sale of investments Purchase of investments	(181,358) (3,867) 800,000	(28,211) 6,349,183 (4,949,302)		

Net cash provided by (used in) investing activities	526 , 902	1,371,670
Cash flows provided by (used in) financing activities: Payments under capital lease Proceeds from sales of common stock	(14,928)	(15,516) 27,188
Net cash provided by (used in) financing activities	(14,928)	11,672
Net increase (decrease) in cash and cash equivalents	135,311	226,474
Cash and cash equivalents at beginning of period	95,575	908,655 ======
Cash and cash equivalents at end of period	\$ 230,886	\$ 1,135,129 =======
Supplemental disclosure of cash flow information: Cash paid during the period for interest	\$ 351	\$ 2,720
Reclassification of preferred stock from equity to liability resulting from repurchase agreement	\$ 350,000	\$

</TABLE>

See notes to condensed consolidated financial statements.

6 ACCESSITY CORP.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

SIX MONTHS ENDED JUNE 30, 2004 (UNAUDITED)

1. BASIS OF PRESENTATION

The information contained in the condensed consolidated financial statements for the six months ended June 30, 2004 and 2003 is unaudited, but includes all adjustments, consisting of normal recurring adjustments, which the Company considers necessary for a fair presentation of the financial position and the results of operations for these periods.

The financial statements and notes are presented in accordance with the requirements of Form 10-QSB, and do not contain certain information included in the Company's annual statements and notes. These financial statements should be read in conjunction with the Company's annual financial statements as reported in its most recent annual report on Form 10-KSB.

In May 2004 the Company signed a definitive agreement, characterized as a share exchange transaction, which if approved by our shareholders would change the business model of the Company to the business of the acquired company and require the resignation of the present officers and directors. See Note 3.

Upon approval from its shareholders at the December 15, 2003 annual shareholders meeting, the Company effected a one-for -five reverse common stock split. The effective date of the stock split was January 7, 2004. All references to common shares, options, warrants or other issues convertible into common shares have been adjusted to reflect this stock split on a retroactive basis. The number of authorized common shares and the par value were not changed.

On August 1, 2003, the Company sold its affinity service automobile business (see Note 7). The accompanying financial statements reflect the results of this business as Discontinued Operations. Accordingly, certain prior period amounts have been reclassified.

This report may contain forward-looking statements that involve certain risks and uncertainties. Factors may arise, including those identified in the Company's Form 10-KSB for the year ended December 31, 2003, which could cause the Company's operating results to differ materially from those contained in any forward-looking statement.

2. BUSINESS OF THE COMPANY

THE FOLLOWING INFORMATION IS PROVIDED AS BACKGROUND RELATED TO THE HISTORIC AND CURRENT BUSINESS ACTIVITIES OF THE COMPANY. AS INDICATED IN NOTE 3, HOWEVER, THE COMPANY IS CURRENTLY ENGAGED IN A TRANSACTION WHICH, IF CONCLUDED, WOULD CHANGE ITS BUSINESS MODEL AND DIRECTORS AND OFFICERS.

repair and collision management from its inception in 1983, but has exited the automotive market and entered into a medical billing recovery business. It divested its original automotive business in February 2002, which provided collision repair and fleet management services primarily for numerous Fortune 500 companies.

The Company also offered collision repair management services during early 2003 for the insurance industry through a website on the Internet. Revenues for such services commenced in December 2001 and continued throughout 2002. However, under a strategic partnership agreement, effective January 2, 2003 (see Note 6), the Company transferred the operating responsibilities and management of this business to a third party and, currently, is no longer engaged in collision repair management. During the early part of 2003 it completed certain in-process repairs that had been initiated by its customers in late 2002. It remains liable for warranties of auto repairs provided, however warranty costs have historically not been significant.

In addition, the Company also sold its remaining automotive business, effective August 1, 2003, that provided automobile affinity services for individuals. A definitive agreement was completed for the sale of all of the outstanding shares of its wholly owned subsidiary to the president of the business (see Note 7). The Company believes that it operated its automotive-related businesses in one operating segment.

During the 2003 period presented, the Company provided collision and general repair programs and appraisal services, for the insurance industry and insurance carriers. The Company facilitated the repair process for insurance carriers by installing its internet-based software at customer sites, which permitted them to enter new claims and to monitor the Company's activities. Once a claim was initiated on the website, the Company commenced its efforts. This included the audit of repair estimates, negotiation of the repair price with one of its suppliers selected from its network of approximately 2,000 providers, management of time for completion of repair, selection or approval of part specifications, and obtaining third party appraisals if required. The Company assumed the risks and responsibilities of the vehicle repair process, from commencement to completion, for its insurance clients. It warranted all repairs completed through its network of repair facilities, for periods up to as long as the driver owned the vehicles and issued warranty certificates for claims processed through its supplier network. The Company recorded revenues gross in these circumstances, having acted as the principal in the transaction. As described in Note 6, this business is now managed by ClaimsNet, Inc. ("ClaimsNet").

During the third quarter of 2002, the Company began a new business, Sentaur Corp. ("Sentaur") engaged in medical billing recovery, a new business segment. The business provides benefits to the hospital segment of the healthcare industry by recouping inappropriate discounts taken from hospital billings by institutional or insurance payors. Sentaur began generating revenue during the second quarter of 2003. The Company records revenues net for this business, having acted as an agent of the hospitals.

Three of the Company's customers currently accounted for approximately 77% of its 2004 continuing revenues to date and three customers accounted for approximately 74% of its outstanding trade receivables at June 30, 2004. Of those three, the Company receives funds from one entity in the automotive segment as described in Note 6.

8

3. DEFINITIVE AGREEMENT SIGNED FOR POTENTIAL ACQUSITIONS, CHANGE OF

CONTROL AND CHANGE IN BUSINESS MODEL

On May 17, 2004 the Company signed a definitive agreement ("the Share Exchange Agreement") with Pacific Ethanol Inc., a California company, Kinergy Marketing, LLC, an Oregon limited liability company, and Re-Energy, LLC, a California company, (collectively hereinafter referred to as the "PEI Group") all of which are geographically located in California, to acquire those companies in exchange for a maximum of 18.8 million shares on a fully diluted basis (of which approximately 2.6 million shares would be reserved to replace existing PEI Group options and for its convertible debt, if converted) of Accessity common stock. The transaction is structured as a stock-for-stock share exchange; with Accessity continuing as the surviving parent company and the PEI Group entities becoming wholly owned subsidiaries of Accessity.

Privately held PEI Group, has developed a strategy aimed at becoming the first vertically integrated producer and marketer of ethanol in California, the nation's most populous state with the highest ethanol demand in the United States. Effective January 1, 2004 the State of California mandated the elimination of MTBE as a gasoline additive and required the use of ethanol as its replacement to improve air quality resulting from auto emissions. The California ethanol market is currently estimated to be annually approximately 900 million gallons, based on published market reports, and approximately \$1 billion dollars. At present, through Kinergy Marketing LLC, PEI Group is presently a re-seller of ethanol and its unaudited results indicated that it generated approximately \$37 million in revenue in the six months ended June 30, 2004. PEI Group owns land and a grain processing facility and has received the critical permits to begin construction of a 35 million gallon ethanol plant on its property in Madera, California. PEI Group believes this will be the largest ethanol plant in California. Funding for the production facility and working capital is expected to be provided in part through bank lending, a term sheet has been received from a large bank lending institution to provide debt of \$25 million subject to certain conditions precedent, including equity requirements, and the remaining part through other debt or equity arrangements which are currently being negotiated.

Under the terms of the Share Exchange Agreement, which is subject to satisfactory completion of due diligence and shareholder approval, the existing directors, officers and employees of the Company would terminate their positions with Accessity. The existing business operations of Sentaur, and the royalty stream from ClaimsNet (see Note 6), the building lease and related personal property, would be spun off and become the responsibility of CEO and founder Barry Siegel in lieu of cash payments required under his employment contract. Thereafter, the management of Accessity would be transferred to the management of the PEI Group, however the present Accessity Board of Directors will have the right to nominate one director with a term that expires until the 2005 Accessity shareholders meeting. The remaining assets, including all of the Company's cash and investment funds, certain prepaid and other assets and selected liabilities, would be retained under the control of the new management of the PEI Group. In addition, the Share Exchange Agreement requires Accessity to prosecute the lawsuit against Presidion's investment bankers, the Mercator Group LLC, Global Taurus LLC, et al, for in excess of \$100 million, as described in the Company's December 31, 2003 Form 10-KSB (also see Note 5). The proceeds, if any, from a successful outcome of this suit, after the payment of legal fees, will be distributed to current Accessity shareholders, on the date of closing of the transaction with the PEI Group, two-thirds of any recovery from this suit with

the remaining one-third being paid the Company. In the event Accessity terminates the Share Exchange Agreement for a Superior Proposal, as defined in the Share Exchange Agreement, Accessity will pay the expenses of the other parties up to a maximum amount of \$150,000.

9

The Company is currently preparing its proxy to be filed with the Securities and Exchange Commission and, upon its review, will forward the proxy to its shareholders and schedule a shareholder meeting to vote for approval of the transaction. The Company has also been in communication with the Nasdaq Stock Market which has deemed this proposed transaction with the PEI Group as a "Reverse Merger" and therefore, would require the Company to file an Initial Listing Application for continued listing of the Company's stock on the Nasdaq SmallCap Stock Market subsequent to the closing of this transaction.

On August 4, 2004, the Company signed Amendment No.1 to the Share Exchange Agreement revising certain non-material language in the agreement and extending the date upon which the Share Exchange Agreement shall terminate should the contemplated transaction not close, from July 31, 2004 to October 29, 2004.

4. PREFERRED STOCK REPURCHASE

In connection with the sale of the Company's former wholly-owned subsidiary, driversshield.com FS Corp. ("FS"), its collision repair and fleet services business, to PHH Vehicle Management Services, LLC, d/b/a PHH Arval ("PHH"), a subsidiary of the Cendant Corporation (NYSE, symbol CD) in February 2002 and, pursuant to the Preferred Stock Purchase Agreement, PHH acquired 1,000 shares of the Company's Series A Convertible Preferred Stock (the "Preferred Shares") for \$1.0 million. The Preferred Shares provided conversion, at the holder's discretion, into 100,000 shares of the Company's common stock (subject to adjustments for stock splits, re-capitalization and anti-dilution provisions).

Effective May 13, 2004, in exchange for certain mutual releases and the amendment of the Stock Purchase Agreement dated October 29, 2001 resulting in the extension of certain non-compete clauses in favor of PHH, the Company and PHH entered into a Stock Repurchase Agreement providing the Company, or its assigns, with the right to repurchase these Preferred Shares for \$350,000. Pursuant to the terms of the Stock Repurchase Agreement, the Company is required to repurchase the Preferred Shares only in the event that the arbitration matter between the Company and Presidion Solutions, Inc. (Note 5) is successfully concluded in favor of the Company, and the award has been fully collected. In June 2004 the arbitrator ruled in favor of the Company, and as of July 28, 2004 $\,$ the Company had collected \$189,000. The balance of \$91,000 is expected to be collected on August 28, 2004. Under the terms of the Stock Repurchase Agreement, PHH may exercise its right to convert all, but not less than all, of its Preferred Shares into common stock, so long as it precedes the Company's election to repurchase the stock, through the termination date of September 15, 2004.

The Company has reclassified its preferred stock out of the equity section, and into a liability account, at fair value, since the redemption of its preferred stock is outside the control of the Company.

10

AWARD GRANTED TO COMPANY IN ARBITRATION MATTER

5.

proceedings under the auspices of the American Arbitration Association had concluded that the Company was entitled to the \$250,000 break-up fee set forth in the Memorandum of Understanding ("MOU") between the Company and Presidion Solutions, Inc. ("Presidion"), as well as its share of the costs of the arbitration and interest from the date of the termination of that agreement by Presidion, aggregating approximately \$30,000. As described more fully in the Company's Form 10-KSB for the year ended December 31, 2003, the Company and Presidion had entered into a MOU with the contemplation of merging Presidion into the Company. Subsequently, Presidion breached the MOU and the Company filed for arbitration. According to the arbitration terms, Presidion was provided thirty days to make the payment. As an accommodation, the Company accepted an initial payment of \$98,332 on June 28, 2004, and an interest-bearing promissory note, in the aggregate amount of \$181,358, comprising two payments of approximately similar amounts to be made on July 28 and August 27, 2004. The Company received the first payment on the note on July 28, 2004. The note bears interest on the unpaid balance at 7% per annum. The arbitration award for the break-up fee and interest is included in Investment and Other Income in the accompanying Condensed Consolidated Statement of Operations and Comprehensive Loss. The Company also received a Confession of Judgment from Presidion to facilitate a court order in the event the note is not paid in full.

6. STRATEGIC PARTNERSHIP FOR INSURANCE BUSINESS

In December 2002, the Company entered into a Strategic Partnership Agreement (the "Partnership Agreement"), effective January 2, 2003, with ClaimsNet, a wholly-owned subsidiary of the CEI Group, Inc. ("CEI"), a Pennsylvania corporation, in which ClaimsNet assumed the responsibilities of servicing the operations and management of DriverShield CRM, the business that provided insurance carriers with collision repair management for their insureds. During 2003 the Company processed only those claims that were initiated prior to the effective date, and ClaimsNet has assumed responsibility for new repairs. The Company granted an exclusive license of its technology, including its website software, that enables insurance customers to access the vehicle claims management system via the Internet, and a non-transferable license of its network of repair facilities, as well as training of its processing methodologies, in order for ClaimsNet to fulfill its obligations under the Partnership Agreement. As consideration, ClaimsNet remits a share of the profits to the Company equivalent to 25% of vendor referral fees for repairs initiated and completed, beginning in March 2003, and 50% of administrative fees, as defined, on all existing customers, beginning in February 2003, as well as15% of all administrative and vendor referral fees for all new customers that use the licensed technology to have their vehicles repaired. The term of the partnership is for a five year period with two consecutive one year renewals. The contract also grants ClaimsNet an option to purchase this business, pursuant to a formula, beginning January 1, 2007.

For the six months ended June 30, 2004 and 2003 the Company recorded fees from ClaimsNet of \$95,000 and \$53,000 respectively. As part of the agreement, during the quarter ended March 31, 2003, the Company incurred \$12,000 for certain personnel costs associated with the transition. There were no comparable costs in the 2004 Quarter.

SADE TO REDATED TART

Upon approval of its board of directors, the Company negotiated a Stock Purchase Agreement ("the ADS Agreement"), effective August 1, 2003, for the sale of all of the outstanding shares of its wholly owned subsidiary, DriverShield ADS Corp. ("ADS") to an employee who is the president of this business. Under the terms of the ADS Agreement the Company received a one-time fee of \$10,000 on September 30, 2003, plus it received reimbursement for its legal fees of approximately \$10,000 incurred for this sale. As a component of the transaction, the individual purchaser also agreed to forego all future rights to receive compensation and other benefits associated with his employment contract, which was to expire in December 2004, but terminated on July 31, 2003. All of the employees and related costs of the ADS business were borne by the purchaser as of the effective date, and the Company has no continuing management of, or responsibility for, the operations. The net liabilities of the business at the closing date, of approximately \$31,000, consisting of primarily accounts receivable and payable, were retained by the Company.

The purchaser of the ADS business, Barry J. Spiegel, was one of the four members of the Board of Directors of the Company, and a significant shareholder, who retained his seat on the Board of Directors until he resigned in May, 2004. With the completion of this transaction, the Company had exited from all operating activities of its various automotive businesses.

The operating results of the affinity services business have been presented as discontinued operations in the accompanying financial statements. The Company recorded a net gain of \$10,000 on the transaction in the quarter ended September 30, 2003.

Operating results during the three and six months ended June 30, 2003, for the discontinued affinity services operations were as follows:

	Three Months Ended June 30, 2003	Six Months Ended June 30, 2003
Revenues	\$142,000	\$377 , 000
Cost of sales, selling, general and administrative expenses	(95,000)	(189,000)
Income from discontinued operations, pre-tax	× \$ 47,000	\$188,000

8. EARNINGS (LOSS) PER SHARE

Basic earnings (loss) per common share is computed by dividing earnings (loss) by the weighted average number of common shares outstanding during the period. Diluted earnings per share reflect the potential dilution that could occur if common stock equivalents, such as preferred stock, stock options and warrants, were exercised. For the six months ended June 30, 2004 and 2003, respectively, approximately 498,000 and 720,000 of potentially dilutive common stock equivalents were excluded from the earnings per share calculations, as their inclusion would have been anti-dilutive.

1

9. STOCK-BASED COMPENSATION PLANS

The Company issues stock options to its employees and outside directors pursuant to stockholder-approved stock option programs, and accounts for stock-based compensation plans under the intrinsic value method of accounting as defined by Accounting Principles Board Opinion No. 25, Accounting for Stock Issued to Employees, and related interpretations. No stock-based employee compensation cost is reflected in net income (loss) for the three and six months ended June 30, 2004 and 2003, as all options granted under these plans had an exercise price equal to the fair market value of the underlying common stock on the date of grant. See Note 10 for variable priced stock options. For pro forma disclosures, the estimated fair value of the option is amortized over the vesting period, which range from immediate vesting to three years. The following table illustrates the effect on net income (loss) and earnings (loss) per share if the Company had accounted for our stock option and stock purchase plans under the fair value method of accounting under Statement 123, as amended by Statement 148:

<TABLE><CAPTION>

	Three Months End			Ended June 30,		Six Months Ended June 30,			
		2004		2003		2004	200)3	
<\$>	<c></c>		<c></c>		<c></c>		<c></c>		
Net income (loss), as reported Deduct: Total stock-based employee compensation expense determined under fair value- based method for all awards,	(\$	145,313)	(\$	428,404)	(\$	569,302)	(\$	921,839)	
net of related tax effects	(\$	85,753)	(\$	135,447)	(\$	171,506)	(\$	275,681)	
Pro forma net income (loss)	(\$	231,066)	(\$	563,851)	(\$	740,808)	(\$ 1	L,197,520)	
Earnings (loss) per share:									
Basic, as reported	(\$.06)	(\$.22)	(\$.25)	(\$.51)	
Basic, pro forma	(\$.10)	(\$.26)	(\$.33)	(\$.55)	
Diluted, as reported	(\$.06)	(\$.22)	(\$.25)	(\$.51)	
Diluted, pro forma 									

 (\$ | .10) | (\$ | .26) | (\$ | .33) | (\$ | .55) |

10. NON-CASH COMPENSATION FOR VARIABLE PRICED OPTIONS

In October 1999 the Company repriced certain options previously granted to employees and third parties, representing the right to acquire 440,000 shares of common stock. The original grants gave holders the right to purchase common shares at prices ranging from \$5.00 to \$6.20; these were repriced to prices ranging from \$3.75 to \$4.15 per share. At the date of the repricing, the new exercise price was equal to the fair market value of the shares (110% of the fair market value in the case of an affiliate). In addition, in September 2002 the Company granted a five-year extension to the life of certain fully vested options that had expired. Pursuant to FASB Interpretation No. 44, the Company accounts for these as variable from the date of the modification until they are exercised, forfeited or expired, and records the intrinsic value of such grants. During the year ended

13

December 31, 2003 all of these options, except for 6,667, were either forfeited or expired, There was no charge or credit during 2004, or comparable period in 2003, as the price per share of the Company's common stock traded at levels below the exercise prices.

Investments at June 30, 2004 consist of available-for-sale securities that had a fair market value of \$3,544,000.

The Company evaluates its individual securities holdings to determine whether it believes that a decline in investment value may be permanent or other-than-temporary. In the quarter ended June 30, 2004 the Company recognized an impairment, characterized as other-than-temporary, of approximately \$73,000, which included the unrealized losses previously reported as the sole component of comprehensive losses, and most of the decrease in value in the current quarter. While the total return for each of its income securities covering the most recent twelve month period has been positive, the Company considered that the increasing interest rate environment, which has resulted in decreasing prices for each of the fixed income mutual funds that the Company held at June 30, 2004, would continue. Further, in order to support its current operating losses, the Company periodically sells some portion of its investment holdings which may preclude its ability to hold securities sufficiently to realize its initial investment. In July 2004 the Company made the decision to sell one of its investment positions, which represented \$33,000 of the impairment amount noted above, concluding that its total return no longer justified the market risk environment.

The investment balance, shown above, is valued at fair market value and accordingly already reflects the impairment.

12. PROFORMA INFORMATION

Proforma information, assuming that the disposal of ADS occurred at the beginning of the earliest quarterly period presented, has not been presented since the disposal has been accounted for as discontinued operations, and such amounts have been reclassified from continuing operations.

13. INCOME TAXES

At December 31, 2003, the Company had operating loss carry forwards of approximately \$3,800,000 and had established a valuation allowance for the full amount of its deferred tax asset as it is more likely that the Company will not be able to realize the tax benefits. To the extent the Company is profitable in the future periods such carry forwards may be available to offset future taxable earnings. To the extent the Company is not profitable it would not be able to realize this benefit. In the event the transaction with the PEI Group is consummated, there may be limitations on the amount of the net operating loss carryforward which may be utilized pursuant to the Internal Revenue Code.

14 14. FLORIDA OFFICE LEASE AND RELATED PARTY TRANSACTION

The 7,300 square foot building in Coral Springs, Florida which the Company leases for its headquarters is owned and operated by B & B Lakeview Realty Corp., whose three shareholders, Barry Siegel, Barry Spiegel and Ken Friedman are, or previously were, members of the Company's Board of Directors. In accordance with the terms of the lease, the Company paid required rentals to B & B Lakeview Realty of approximately \$33,000 in the current quarter and \$66,000 during the 2004 Period. Pursuant to the lease agreement, the Company is also required to pay various building maintenance, insurance and other specified charges, as incurred, to other unrelated vendors. It was also required to establish a restricted depository account, in the amount of \$300,000, as described in the Liquidity and Capital Resources section of Managements Discussion and Analysis or Plan of Operation.

15. SEGMENT INFORMATION

The Company currently reports two segments, medical and automotive. As described in Note 6, however, the Company participates in the automotive segment only through a profit-sharing arrangement; it no longer operates, or has liability for, the current activities of the automotive segment, which function under the managerial autonomy of ClaimsNet, pursuant to its contractual arrangement with the Company. The Company manages these segments separately since each serves different markets and users, as described in Note 2.

All of the Company's sales are made within the domestic United States. Segment information follows. <TABLE><CAPTION>

		Six Months Ended June 30,				Three Months Ended June 30,			
<s></s>		2004		2003		2004		2003	
		 <c></c>		 <c></c>		<c></c>	· · · · · · · · · · · · · · · · · · ·	<0	:>
Rev	enue:								
	Medical Automotive	Ş	144,000 55,000	Ş	63,000 49,000	Ş	330,000 95,000	\$	63,000 190,000
~	Consolidated total	\$	199,000	\$	112,000	\$	425,000	\$	253,000

Segment profit (loss):

Medical (1)	\$ (35,000)	\$ (58,000)	Ş	(25,000)	\$	(181,000)
Automotive (1)	30,000	(13,000)		46,000		(27,000)
Other/corporate (1)	 (140,000)	 (405,000)		(590,000)		(902,000)
Consolidated total						

 \$ (145,000) | \$ (476,000) | Ş | (569,000) | \$ (| (1,110,000) |Segment profit or (loss) reflects continuing operations before provision for income taxes (benefit).

Identifiable assets at June	30,	2004:
<table><caption></caption></table>		
<\$>	<c2< td=""><td>></td></c2<>	>
Medical	Ş	117,000
Automotive		94,000
Other, corporate		4,647,000
Consolidated total	\$	4,858,000
/		

</TABLE>

(1) The Company does not allocate taxes, other income, interest income or expense, or its corporate general and administrative expenses to its individual segments. The segment profit (loss) shown above reflects those costs that are directly and specifically identifiable with the operating activities of the segment.

15 ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OR PLAN OF OPERATION

Forward Looking Statements - Cautionary Factors

The following discussion and analysis should be read in conjunction with the Company's financial statements and the notes hereto appearing elsewhere in this report. This report contains forward-looking statements within the meaning of the Private Securities Litigation Act of 1995. The Company cautions that forward-looking statements are not guarantees of future performance and involve certain risks and uncertainties (including those identified in "Risk Factors" in the Company's Form 10-KSB for the year ended December 31, 2003 and Form 8-K dated May 17, 2004) and that actual results may differ materially from those in the forward-looking statements as a result of various factors. Except for the historical information and statements contained in this Report, the matters and items set forth in this Report are forward looking statements.

The 2004 Quarter reflected a net loss of \$145,000 compared to a net loss of \$428,000 in the 2003 Quarter. Loss from continuing operations was \$145,000 in the 2004 Quarter versus a loss of \$476,000 in the 2003 Quarter; a reduction in losses of 70%. The reduction was largely attributable to a one-time arbitration award of \$280,000 resulting from a breach of an agreement. Basic and diluted loss per share from continuing operations was \$.06 and \$.22 per share in the 2004 and 2003 Quarters respectively. Basic and diluted income per share from discontinued operations was zero in the 2004 Quarter and \$.02 in the 2003 Quarter.

REVENUES FROM CONTINUING OPERATIONS

Revenues were \$199,000 in the 2004 Quarter, versus \$112,000 in the 2003 Quarter, representing an increase of \$87,000 or 78%. Revenues increased by \$6,000 in the Company's automotive segment, from \$49,000 in the 2003 Quarter to \$55,000 in the 2004 Quarter. The revenues that the Company recorded from its automotive segment reflect its share of fees from claims processed by ClaimsNet. Sentaur revenues increased \$82,000 to \$145,000 in the 2004 Quarter from \$63,000 in the 2003 Quarter when its revenues commenced. However, Sentaur revenues declined from the first calendar 2004 quarter and it is currently not supporting its direct expenses and is losing money. A few of its existing contracts are now winding down, and Sentaur's new contracts were not sufficient to offset the declining revenues.

OPERATING INCOME AND EXPENSES FROM CONTINUING OPERATIONS

Losses from continuing operations decreased 70%, to \$145,000 in the 2004 Quarter compared to a loss of \$476,000 in the 2003 Quarter, a decrease in losses of \$331,000. The comparative amounts are described below.

Collision repair expense relating to its automotive repair business decreased to zero in the 2004 Quarter versus \$3,000 in the 2003 Quarter resulting from the transfer of the business to ClaimsNet, described above.

16

Selling expenses incurred by the Company's Sentaur business, which were relatively comparable on a quarter-to-quarter basis, increased by 7,000 (6%), to 117,000 in the 2004 Quarter, from 110,000 in the 2003 Quarter.

General and administrative expenses, which were also relatively comparable on a quarter-to-quarter basis, decreased by 17,000 (4%), from \$433,000 in the 2003 Quarter to \$416,000 in the 2004 Quarter

Depreciation declined \$25,000, from \$85,000 in the 2003 Quarter to \$60,000 in the 2004 Quarter, resulting from assets which became fully depreciated.

Investment and other income, net, increased \$192,000 from \$44,000 in the 2003 Quarter to \$236,000 in the 2004 Quarter. Other income in the 2004 Quarter included \$272,000 (excluding reimbursed arbitration costs) from Presidion resulting from an arbitration award from their breach of an agreement with the Company. The Company was awarded \$250,000 for the stipulated break-up fee, plus certain costs and interest. Offsetting the award was a non-cash impairment of \$73,000 recorded on marketable securities which recognized most of the unrealized losses incurred on fixed income mutual funds. Aside from arbitration award and the impairment, investment and other income declined \$7,000 resulting primarily from declining investment balances and lower interest rates.

The Company also recorded a tax credit to be repaid by the New York State Department of Taxation and Finance in the 2004 Quarter resulting from overpayments in the prior taxable year. There was no such amount in the 2003 Quarter.

DISCONTINUED OPERATIONS

Discontinued operations in the 2003 Quarter, reflects the net operating results of the affinity services subsidiary which was sold effective August 1, 2003. In the 2004 Quarter there were no discontinued operations.

00ME 30, 2003 (INE 2003 FERIOD).

The 2004 Period reflected a net loss of \$569,000 compared to a net loss of \$922,000 in the 2003 Period. Loss from continuing operations was \$569,000 in the 2004 Period versus a loss of \$1,110,000 in the 2003 Period; a reduction in losses of \$541,000, or 49%. The reduction was largely attributable to the receipt of a one-time arbitration award of \$280,000 resulting from a breach of an agreement. Basic and diluted loss per share from continuing operations was \$.25 and \$.51 per share in the 2004 and 2003 Period respectively. Basic and diluted income per share from discontinued operations was \$.09 in the 2003 Period and zero in the 2004 Period.

REVENUES FROM CONTINUING OPERATIONS

Revenues were \$425,000 in the 2004 Period, versus \$253,000 in the 2003 Period, representing an increase of \$172,000 or 68%. The Company's revenues decreased by \$95,000 in its automotive segment, from \$190,000 in the 2003 Period to \$95,000 in the 2004 Period as a result of transferring the operating responsibility of its CRM business to ClaimsNet, effective January 2003. However, as described below, the significant reduction in infrastructure costs eliminated the direct expenses and losses from this business

17

segment (excluding corporate overhead which the Company does not allocate to its operating units). The revenues the Company recorded in the 2004 Period reflect referral fees associated with claims processed by ClaimsNet. Offsetting the reduction in revenues from its automotive segment was an increase in revenues of \$267,000 from Sentaur, the Company's financial recovery business for hospitals. Its revenues increased from \$63,000 in the 2003 Period to \$330,000 in the 2004 Period. Sentaur commenced recording revenues in April 2003 and there were only three months billings in the 2003 Period. While revenues have increased, Sentaur is currently losing money and not supporting its direct expenses.

OPERATING INCOME AND EXPENSES FROM CONTINUING OPERATIONS

Losses from continuing operations decreased 49%, to 569,000 in the 2004 Period compared to a loss of 1,110,000 in the 2003 Period, a decrease in losses of 541,000. The comparative amounts are described below.

Collision repair expense relating to its automotive repair business, decreased to zero in the 2004 Period versus \$101,000 in the 2003 Period resulting from the transfer of the business to ClaimsNet, described above.

Selling expenses decreased by \$26,000 (10%), to \$223,000 in the 2004 Period, from \$249,000 in the 2003 Period. This was the result of lower selling expenses for all business activities including Sentaur and other corporate marketing activities, including some transition marketing expenses incurred for CRM in the 2003 Period for which there was no comparable amount in the 2004 Period.

General and administrative expenses increased by \$11,000 (1%), from \$930,000 in the 2003 Period to \$941,000 in the 2004 Period. Increased legal

expenses relating to our claim against Presidion Solutions, Inc. which was arbitrated (and an award granted to the Company) in the 2004 Period, as well as two other claims (as described in the Company's December 31, 2003 10-KSB), one in which it is a plaintiff and one as a defendant, are the primary reasons for the increase. In addition, the Company has incurred legal expenses associated with due diligence for the PEI Group transaction. During the 2004 Period the Company incurred legal expenses totaling \$172,000, versus \$133,000 in 2003 the Period, an increase of \$39,000. Excluding legal costs, all other general and administrative expenses in the aggregate declined by \$23,000.

Depreciation declined \$50,000, from \$169,000 in the 2003 Period to \$119,000 in the 2004 Period, resulting from assets which became fully depreciated.

Investment and other income, net, increased \$189,000 from \$87,000 in the 2003 Period to \$276,000 in the 2004 Period. Other income in the 2004 Period included \$272,000 (excluding reimbursed arbitration costs) from Presidion resulting from an arbitration award from their breach of an agreement with the Company. The Company was awarded \$250,000 for the stipulated break-up fee, plus certain costs and interest. Offsetting the award was a non-cash impairment of \$73,000 recorded on marketable securities which recognized most of the unrealized losses incurred on fixed income mutual funds. Aside from arbitration award and the impairment, other income declined \$10,000 resulting primarily from declining investment balances and lower interest rates.

18

The Company also recorded a tax credit to be repaid from the New York State Department of Taxation and Finance in the 2004 Period resulting from overpayments in the prior taxable year. There was no such amount in the 2003 Period.

DISCONTINUED OPERATIONS

Discontinued operations in the 2003 Period, reflects the net operating

results of the affinity services subsidiary which was sold effective August 1, 2003. In the 2004 Period there were no discontinued operations.

LIQUIDITY AND CAPITAL RESOURCES

_ _____

As of June 30, 2004 the Company had cash and cash equivalents of \$231,000. The Company also holds shares in a number of highly liquid mutual funds valued at \$3,544,000. Working capital of the Company as of June 30, 2004 was \$3,440,000 and its working capital ratio was 5:1.

In connection with the Company's rental of office space in Florida, in July 2002, the Company was required to establish a \$300,000 restricted cash fund, initially invested in a certificate of deposit, with a Florida bank for the five and a half year term of the lease, as a guarantee of its future rental commitments. Such amounts are excluded from liquidity and working capital, described above, and presented as restricted cash. The restricted fund amount declines as the remaining rental commitment declines, as follows; the balance of the certificate will be \$200,000 after the 36th month, \$100,000 after the 48th month, and zero after 60 months.

The Company has no major expenditures that it currently anticipates for capital equipment, however it is expending funds due to operating losses, including funding the growth of its Sentaur business unit. As Sentaur obtains new hospital customers and seeks to expand its sales, it may require additional funds for personnel expenses and software systems development, but this would occur in anticipation of future revenue growth. Should we not complete the transaction with the PEI Group described above in Note 3, we would expect to use our resources to support Sentaur's growth during the remainder of 2004 and thereafter. Also, the Company incurred an unusually high level of legal expense in the 2004 Period in connection with three claims; two of which the Company is the plaintiff (it has won one of those cases as discussed in Note 5 above) and one in which it is the defendant. We anticipate this level of legal costs will decline.

In addition, the Company has spent considerable management effort and time pursuing acquisition candidates, and has incurred varying levels of expenses in connection with each evaluation. These have ranged from minor amounts for such expenses as an initial business trip or, more extensively, multiple trips for due diligence, legal review and lien and judgment searches. We are currently expending funds for the transaction with the PEI Group described in Note 3, above. Should we not complete this transaction, and seek another acquisition, we may use a significant amount of our funds to either pay a portion of the purchase price and/or expand the business we acquire.

The Company believes that its present liquidity will enable it to continue to support its operations, as they are currently configured for our continuing business, for the next twelve months and for an extended period thereafter depending on the extent of use of its funds to build existing businesses or possible use of funds to develop or acquire new businesses. The Company announced on May 17, 2004 that it has signed a share exchange agreement to acquire Pacific Ethanol, Inc., Kinergy Marketing, LLC and Re-Energy, LLC in a stock-for-stock share exchange transaction. Upon consummation of the share exchange, each of the acquired companies will become wholly-owned subsidiaries of Accessity Corp. and Accessity Corp. will re-incorporate in the State of Delaware and change its name to Pacific Ethanol, Inc.

Accessity Corp. will issue approximately 18.8 million shares to acquire all the companies in this transaction. It is contemplated that the combined company will have approximately 22 million shares of common stock outstanding, on a fully-diluted basis, should all options and warrants be exercised following consummation of the share exchange transaction.

The proposed share exchange, expected to be completed as quickly as possible, is subject to satisfaction of due diligence investigations by all of the parties, approval by a majority of Accessity's shareholders and certain other additional conditions to closing including completion of audits of Pacific Ethanol, Kinergy Marketing and Re-Energy. As a further condition to the completion of the acquisitions, the current management of Accessity will resign and the current management of the acquired companies will assume management of the combined companies. The former Board of Directors of Accessity will designate one person to serve on the board of directors of Pacific Ethanol until the 2005 annual shareholders meeting. [See Note 3 to the Financial Statements herein]

ITEM 3. CONTROLS AND PROCEDURES

EVALUATION OF DISCLOSURE CONTROLS AND PROCEDURES.

Under the supervision and with the participation of our management, including our Chief Executive Officer and Chief Financial Officer, we evaluated the effectiveness of the design and operation of our disclosure controls and procedures (as defined in ss.240.13a-15(e) or 240.15d-15(e) under the Exchange Act) as of June 30, 2004. Based upon that evaluation required by section ss.240.13a-15 or 240.15d-15 under the Exchange Act, the Chief Executive Officer and Chief Financial Officer concluded that, our disclosure controls and procedures were effective in timely alerting them to the material information relating to us (or our consolidated subsidiaries) required to be included in our periodic SEC filings.

CHANGES IN INTERNAL CONTROLS.

There were no significant changes made in our internal controls during the period covered by this report, or to our knowledge, in other factors that could significantly affect these controls subsequent to the date of their evaluation.

20 PART II. OTHER INFORMATION

ITEM 6. EXHIBITS AND REPORTS ON FORM 8-K

(a) EXHIBITS

- 10.1 Share Exchange Agreement dated May 12, 2004 by and among Accessity Corp., a New York corporation Pacific Ethanol, Inc., a California corporation ("PEI"); Kinergy Marketing, LLC, an Oregon limited liability company ("Kinergy"); Reenergy, LLC, a California limited liability company ("Reenergy,"; each of the shareholders of PEI identified on the signature pages thereto (collectively, the "PEI Shareholders"); each of the holders of options or warrants to acquire shares of common stock of PEI identified on the signature pages thereto (collectively, the "PEI Warrantholders"); each of the limited liability company members of Kinergy identified on the signature pages thereto (collectively, the "Kinergy Members"); each of the limited liability company members of Reenergy identified on the signature pages thereto (collectively, the "Reenergy Members")
- 10.2 Amendment No. 1 to Share Exchange Agreement made and entered into on July 29, 2004 by and among Accessity Corp., a New York corporation Pacific Ethanol, Inc., a California corporation ("PEI"); Kinergy Marketing, LLC, an Oregon limited liability company ("Kinergy"); Reenergy, LLC, a California limited liability company ("Reenergy,"; each of the shareholders of PEI identified on the signature pages thereto (collectively, the "PEI Shareholders"); each of the holders of options or warrants to acquire shares of common stock of PEI identified on the signature pages thereto (collectively, the "PEI Warrantholders"); each of the limited liability company members of Kinergy identified on the signature pages thereto (collectively, the "Kinergy Members"); each of the limited liability company members of Reenergy identified on the

signature pages thereto (collectively, the "Reenergy Members")

- 31.1 Certification of Chief Executive Officer
- 31.2 Certification of Chief Financial Officer
- 32.1 Certification of Chief Executive Officer
- 32.2 Certification of Chief Financial Officer

(b) REPORTS ON FORM 8-K

The Company filed a Current Report on Form 8-K dated March 31, 2004 disclosing a press release that announced the Company's financial results for the period ended March 31, 2004.

The Company filed a Current Report on Form 8-K dated May 17, 2004 disclosing a press release that announced the execution of the Share Exchange Agreement, by and among the Company and Pacific Ethanol Inc., Kinergy Marketing, LLC, and Re-Energy LLC. for the Company to acquire each of these entities as wholly owned subsidiaries in a share exchange transaction.

21

SIGNATURES

 $\label{eq:pursuant to the requirements of the Exchange Act, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.$

Accessity Corp.

Date: August 16, 2004

By: Barry Siegel ------Chairman of the Board, President and Chief Executive Officer

Date: August 16, 2004

22 INDEX OF EXHIBITS

10.1 Share Exchange Agreement dated May 12, 2004 by and among Accessity Corp., a New York corporation Pacific Ethanol, Inc., a California corporation ("PEI"); Kinergy Marketing, LLC, an Oregon limited liability company ("Kinergy"); Reenergy, LLC, a California limited liability company ("Reenergy,"; each of the shareholders of PEI identified on the signature pages thereto (collectively, the "PEI Shareholders"); each of the holders of options or warrants to acquire shares of common stock of PEI identified on the signature pages thereto (collectively, the "PEI Warrantholders"); each of the limited liability company members of Kinergy identified on the signature pages thereto (collectively, the "Kinergy Members"); each of the limited liability company members of Reenergy identified on the signature pages thereto (collectively, the "Reenergy Members")

10.2 Amendment No. 1 to Share Exchange Agreement made and entered into on July 29, 2004 by and among Accessity Corp., a New York corporation Pacific Ethanol, Inc., a California corporation ("PEI"); Kinergy Marketing, LLC, an Oregon limited liability company ("Kinergy"); Reenergy, LLC, a California limited liability company ("Reenergy,"; each of the shareholders of PEI identified on the signature pages thereto (collectively, the "PEI Shareholders"); each of the holders of options or warrants to acquire shares of common stock of PEI identified on the signature pages thereto (collectively, the "PEI Warrantholders"); each of the limited liability company members of Kinergy identified on the signature pages thereto (collectively, the "Kinergy Members"); each of the limited liability company members of Reenergy identified on the signature pages thereto (collectively, the "Reenergy Members")

- 31.1 Certification of Chief Executive Officer
- 31.2 Certification of Chief Financial Officer
- 32.1 Certification of Chief Executive Officer
- 32.2 Certification of Chief Financial Officer

SHARE EXCHANGE AGREEMENT

THIS SHARE EXCHANGE AGREEMENT (this "Agreement") is made and entered into as of May 12, 2004, by and among Accessity Corp., a New York corporation ("Accessity"); Pacific Ethanol, Inc., a California corporation ("PEI"); Kinergy Marketing, LLC, an Oregon limited liability company ("Kinergy"); Reenergy, LLC, a California limited liability company ("Reenergy," and together with PEI and Kinergy, the "Acquired Companies"); each of the shareholders of PEI identified on the signature pages hereof (collectively, the "PEI Shareholders"); each of the holders of options or warrants to acquire shares of common stock of PEI identified on the signature pages hereof (collectively, the "PEI Warrantholders"); each of the limited liability company members of Kinergy identified on the signature pages hereof (collectively, the "Kinergy Members"); each of the limited liability company members of Reenergy identified on the signature pages hereof (collectively, the "Reenergy Members").

R E C I T A L S

A. PEI is in the business of developing a large-scale ethanol plant (the "PEI Business"). Kinergy is in the business of marketing ethanol throughout the Western United States (the "Kinergy Business"). Reenergy is in the business of developing a large-scale ethanol plant (the "Reenergy Business"). The PEI Business, Kinergy Business and Reenergy Business are sometimes collectively referred to herein as the "Businesses."

B. The PEI Shareholders are the holders of all of the issued and outstanding capital stock of PEI (collectively, the "PEI Stock"); the PEI Warrantholders are the holders of all of the issued and outstanding options and warrants to acquire shares of PEI Stock (collectively, the "PEI Warrants"); Kinergy Members are the holders of all of the outstanding limited liability company membership interests of Kinergy (collectively, the "Kinergy Interests"); and the Reenergy Members are the holders of all of the outstanding limited liability company membership interests of Reenergy (collectively, the "Reenergy Interests").

C. Accessity desires to acquire the PEI Stock from the PEI Shareholders, and the PEI Shareholders desire to transfer the PEI Stock to Accessity, in exchange for shares of Common Stock of Accessity ("Accessity Exchange Shares"), subject to and in accordance with the terms and conditions set forth herein; Accessity desires to have the PEI Warrants cancelled in exchange for the issuance to the PEI Warrantholders of warrants to acquire shares of common stock of Accessity, subject to and in accordance with the terms and conditions set forth herein; Accessity desires to acquire the Kinergy Interests from the Kinergy Members, and the Kinergy Members desire to transfer the Kinergy Interests to Accessity, in exchange for Accessity Exchange Shares, subject to and in accordance with the terms and conditions set forth herein; and Accessity desires to acquire the Reenergy Interests from the Reenergy Members, and the Reenergy Members desire to transfer the Reenergy Interests to Accessity, in exchange for Accessity Exchange Shares, subject to and in accordance with the terms and conditions set forth herein.

D. This Agreement is being entered into by the parties as part of a unified plan for the exchange of stock that is intended by the parties to qualify for non-recognition treatment under Section 351 of the Internal Revenue Code of 1986, as amended (the "Code").

AGREEMENT

NOW, THEREFORE, in consideration of the foregoing premises and the respective promises of the parties set forth herein, the parties hereto agree as follows: ARTICLE I

DEFINITIONS

For purposes of this Agreement, in addition to the other capitalized terms defined elsewhere in this Agreement, the following terms shall have the

meanings specified or referred to in this Article 1:

"Action" or "Actions" shall mean any litigation, suits, actions, causes of actions, and proceedings or investigations, collectively.

"Affiliate" shall mean, with respect to any individual, partnership, corporation, limited liability company, association, business trust, joint venture, governmental entity or other entity of any nature ("Person"), any Person that controls, is controlled by, or is under common control with, such Person.

"Code" shall mean the Internal Revenue Code of 1986, as amended.

"Environmental, Health, and Safety Liabilities" shall mean any cost, damages, expense, liability, obligation, or other responsibility arising from or under Environmental Law or Occupational Safety and Health Law and consisting of or relating to:

(a) any environmental, health, or safety matters or conditions
 (including on-site or off-site contamination, occupational safety and health, and regulation of chemical substances or products);

(b) fines, penalties, judgments, awards, settlements, legal or administrative proceedings, damages, losses, claims, demands and response, investigative, remedial, or inspection costs and expenses arising under Environmental Law or Occupational Safety and Health Law;

(c) financial responsibility under Environmental Law or Occupational Safety and Health Law for cleanup costs or corrective action, including any investigation, cleanup, removal, containment, or other remediation or response actions ("Cleanup") required by applicable Environmental Law or Occupational Safety and Health Law (whether or not such Cleanup has been required or requested by any Governmental Body or any other Person) and for any natural resource damages; or

(d) any other compliance, corrective, investigative, or remedial measures required under Environmental Law or Occupational Safety and Health Law.

The terms "removal," "remedial," and "response action" include but are not limited to the types of activities covered by the United States Comprehensive Environmental Response, Compensation, and Liability Act, 42 U.S.C. Section 9601 et seq., as amended ("CERCLA").

2

"Environmental Laws" shall mean any federal, state and local environmental laws, rules, regulations, standards and requirements, including, without limitation, those respecting hazardous materials and substances (including, without limitation, the Comprehensive Environmental Response, Compensation and Liability Act, as amended, 42 U.S.C. sec. 9601, et. seq.; the Resource Conservation and Recovery Act, as amended, 42 U.S.C. sec. 6901. et. seq.; the Federal Water Pollution Control Act, as amended, 33 U.S.C sec. 1251, et. seq.; the Toxic Substances Control Act, as amended, 15 U.S.C. sec. 9601, et. seq.; the Emergency Planning and Community Right to Know Act, 42 U.S.C. sec. 11001, et. seq.; the Safe Drinking Water Act, 42 U.S.C. sec. 300f, et. seq.; the Solid Waste Disposal Act, as amended; and all comparable state and local laws; and any common law (including without limitation common law that may impose strict liability) that may impose liability or obligations for injuries or damages to, or threatened as a result of, the present of or exposure to any hazardous materials or substances).

"ERISA" shall mean the Employee Retirement Income Security Act of 1974 or any successor law, and regulations and rules issued pursuant to that Act or any successor law.

"Exchange Act" shall mean the Securities Exchange Act of 1934, as amended, or any successor law, and regulations and rules issued pursuant to that Act or any successor law

"GAAP" shall mean generally accepted United States accounting principles, applied on a basis consistent with the basis.

(a) nation, state, county, city, town, village, district, or other jurisdiction of any nature;

(b) federal, state, local, municipal, foreign, or other government;

(c) governmental or quasi-governmental authority of any nature (including any governmental agency, branch, department, official, or entity and any court or other tribunal);

(d) multi-national organization or body; or

(e) body exercising, or entitled to exercise, any administrative, executive, judicial, legislative, police, regulatory, or taxing authority or power of any nature.

"Legal Requirement" shall mean any federal, state, local, municipal, foreign, international, multinational, or other administrative order, constitution, law, ordinance, principle of common law, regulation, statute, or treaty.

"Liability" or "Liabilities" shall mean debts, liabilities, commitments or obligations of any nature, absolute, accrued, contingent or otherwise.

"Lien" or "Liens" shall mean any mortgage, pledge, security interest, conditional sale or other title retention agreement, encumbrance, lien, easement, claim, right, covenant, restriction, right of way, warrant, option or charge of any kind.

"Occupational Safety and Health Law" shall mean any Legal Requirement designed to provide safe and healthful working conditions and to reduce occupational safety and health

3

hazards, and any program, whether governmental or private (including those promulgated or sponsored by industry associations and insurance companies), designed to provide safe and healthful working conditions.

"Owners" shall mean the PEI Shareholders, the Kinergy Members and the Reenergy Members, collectively.

"Permit" or "Permits" shall mean any licenses, permits, authorizations, approvals, consents, franchises and orders required for the conduct and operation of business as presently conducted.

"Permitted Liens" shall mean any (i) Liens for taxes not yet due and payable or for taxes that are being contested in good faith through appropriate proceedings, (ii) Liens for purchase money security interests and Liens securing rental payments under capital lease arrangements, (iii) other Liens arising in the ordinary course of business and not incurred in connection with the borrowing of money, and (iv) Liens described on Schedules 4.13, 6.13 and 8.13.

"SEC" shall mean the Securities and Exchange Commission.

"Securities Act" shall mean the Securities Act of 1933, as amended, or any successor law, and regulations and rules issued pursuant to that Act or any successor law.

"Share Exchange" shall mean the PEI Exchange, the Kinergy Exchange and the Reenergy Exchange, collectively.

ARTICLE II EXCHANGE OF OWNERSHIP INTERESTS

2.1 Exchange of PEI Stock. Subject to the terms and conditions of this Agreement, each PEI Shareholder hereby agrees to assign, transfer and deliver to Accessity the shares of PEI Stock owned by the PEI Shareholder free and clear of all liens, claims, encumbrances, pledges, options, security interests and any other adverse interests of any kind or nature whatsoever, and Accessity hereby agrees to accept delivery of the PEI Stock from each of the PEI Shareholders. In consideration for the assignment and transfer of the PEI Stock to Accessity by the PEI Shareholders, Accessity shall issue to each of the PEI Shareholders one (1) Accessity Exchange Share for each one (1) share of PEI Stock ("PEI Exchange Ratio"), as set forth on Exhibit A (the "PEI Exchange"). No fractional shares shall be issued and in the event that the conversion results in a fraction, the number of Accessity Exchange Shares to be issued shall be rounded up to the nearest whole number.

2.2 Cancellation and Replacement of PEI Warrants. Subject to the terms and conditions of this Agreement, each PEI Warrantholder hereby agrees that the PEI Warrants which such PEI Warrantholder has to acquire shares of common stock of PEI shall be cancelled on and as of the Closing Date in consideration for the issuance by Accessity to such PEI Warrantholder of warrants to acquire the same number of shares of Accessity Common Stock at the same exercise price and on the same terms and conditions as provided for in the PEI Warrants of such PEI Warrantholder (the "Accessity Replacement Warrants").

4

2.3 Exchange of Kinergy Interests. Subject to the terms and conditions of this Agreement, each Kinergy Member hereby agrees to assign, transfer and deliver to Accessity the Kinergy Interests owned by each of the Kinergy Members free and clear of all Liens and any other adverse interests of any kind or nature whatsoever, and Accessity hereby agrees to accept delivery of the Kinergy Interests from each of the Kinergy Members. In consideration for the assignment and transfer of the Kinergy Interests to Accessity by the Kinergy Members, Accessity shall issue 1,875,000 Accessity Exchange Shares to Neil Koehler, the sole Kinergy Member, for 100% of the Kinergy Interests ("Kinergy Exchange Ratio"), as set forth on Exhibit A (the "Kinergy Exchange"). No fractional shares shall be issued and in the event that the conversion results in a fraction, the number of Accessity Exchange Shares to be issued shall be rounded up to the nearest whole number.

2.4 Exchange of Reenergy Interests. Subject to the terms and conditions of this Agreement, each Reenergy Member hereby agrees to assign, transfer and deliver to Accessity the Reenergy Interests owned by each of the Reenergy Members free and clear of all liens, claims, encumbrances, pledges, options, security interests and any other adverse interests of any kind or nature whatsoever, and Accessity hereby agrees to accept delivery of the Reenergy Interests from each of the Reenergy Members. In consideration for the assignment and transfer of the Reenergy Interests to Accessity by the Reenergy Members, Accessity shall issue to each of the Reenergy Members 21,250 Accessity Exchange Shares for each one percent (1%) of Reenergy Interests ("Reenergy Exchange Ratio"), as set forth on Exhibit A (the "Reenergy Exchange"). No fractional shares shall be issued and in the event that the conversion results in a fraction, the number of Accessity Exchange Shares to be issued shall be rounded up to the nearest whole number.

2.5 Share Exchange. The parties intend to adopt this Agreement and consummate the Share Exchange as part of a unified plan for the exchange of stock that is qualified for non-recognition treatment under Section 351 of the Code. The Accessity Exchange Shares issued in the PEI Exchange will be issued solely in exchange for shares of PEI Stock, the Accessity Exchange Shares issued in the Kinergy Exchange will be issued solely in exchange for the Kinergy Interests, and the Accessity Exchange Shares issued in the Reenergy Exchange will be issued solely in exchange for the Reenergy Interests, and no other transaction other than the Share Exchange represents, provides for or is intended to be an adjustment to the consideration given for the PEI Stock, the Kinergy Interests and the Reenergy Interests. No consideration that could constitute "other property or money" within the meaning of Section 351(b) of the Code is being transferred by Accessity for the PEI Stock, the Kinergy Interests or the Reenergy Interests. The parties shall not take a position on any tax return inconsistent with this Section 2.5. In addition, the parties represent that, as of the Closing Date and after giving effect to the transactions contemplated by this Agreement, PEI, Kinergy and Reenergy shall be "in control" of Accessity within the meaning of Section 351(a) of the Code.

ARTICLE III CLOSING

3.1 Closing. The closing (the "Closing") of the transactions contemplated by this Agreement will be held at 10:00 a.m. at the offices of Rutan & Tucker, LLP, 611 Anton Boulevard, Suite 1400, Costa Mesa, California

92626, on such date as the parties hereto shall mutually agree upon, or at such other time, date or location as the parties hereto may mutually agree upon (the "Closing Date").

5

3.2 Deliveries By Accessity. At the Closing, Accessity shall deliver:

(a) to each PEI Shareholder, a stock certificate evidencing his or her ownership of the number of Accessity Exchange Shares set forth opposite his or her name as set forth on Exhibit B; to each PEI Warrantholder, an Accessity Replacement Warrant evidencing such PEI Warrantholder's right to acquire the number of shares of Accessity Common Stock set forth opposite his or her name as set forth on Exhibit B and otherwise providing for the same terms and conditions as provided for in the PEI Warrants of such PEI Warrantholder; to each Kinergy Member, a stock certificate evidencing his or her ownership of the number of Accessity Exchange Shares set forth opposite his or her name as set forth on Exhibit C; and to each Reenergy Member, a stock certificate evidencing his or her ownership of the number of Accessity Exchange Shares set forth opposite his or her name as set forth on Exhibit D;

(b) a copy of the Articles of Incorporation and Bylaws of Accessity, each as amended to date, and the resolutions adopted by the Board of Directors of Accessity approving, authorizing and directing the execution of this Agreement and the transactions contemplated thereby, each certified by the Secretary of Accessity as being in full force and effect on and as of the Closing Date;

(c) a certificate of the Secretary of State of Delaware to the effect that Accessity is a validly existing corporation in good standing under the laws of the State of Delaware and a certificate from the Secretary of State of each other state in which the character of its properties owned or leased or the nature of its activities requires qualification as a foreign corporation doing business in such state to the effect that Accessity (as a New York corporation) is a foreign corporation in good standing under the laws of such state;

(d) the written resignations of each of Barry Siegel, Barry J. Spiegel, Kenneth J. Friedman and Bruce S. Udell as directors of Accessity dated as of the Closing Date, in form and substance reasonably acceptable to each of PEI, Kinergy and Reenergy; provided, that one current director may temporarily remain for the sole purpose of confirming said resignations and appointing one individual designated by Accessity as a Class II director (thereby holding such board seat until the annual meeting of Accessity shareholders to be held in the fourth calendar quarter of 2005) pursuant to a unanimous written consent of such remaining sole director, in form and substance reasonably acceptable to PEI, Kinergy and Reenergy, to be delivered by Accessity at the Closing; and the written resignation of such remaining sole director dated as of the Closing Date effective immediately after the effectiveness of such appointment, in form and substance reasonably acceptable to PEI, Kinergy and Reenergy;

(e) a certificate of the president or chief executive officer of Accessity certifying that the representations and warranties by Accessity set forth in this Agreement and in any certificate or document delivered pursuant to the provisions of this Agreement are true and accurate, on and as of the Closing Date, and that Accessity has performed and complied in all material respects with all agreements, obligations and conditions required by this Agreement to be performed or complied with by it on or prior to the Closing Date.

(f) the written resignation of each executive of Accessity who has entered into an employment agreement with Accessity (including, without limitation, a confirmation of the voluntary termination by such individual of his or her existing employment agreement with

6

Accessity), including but not limited to Barry Siegel and Philip Kart, dated as of the Closing Date, in form and substance reasonably acceptable to each of PEI, Kinergy and Reenergy.

(g) an opinion of legal counsel to Accessity to the effect that: (i) Accessity is a corporation duly incorporated, validly existing and in good standing under the laws of New York and is duly qualified as a foreign corporation in each state in which the character of its properties owned or

leased or the nature of its activities requires qualification as a foreign corporation doing business in such state, except where the failure to be so qualified would not have a material adverse effect on Accessity; (ii) this Agreement and each related agreement to which Accessity is a party has been duly authorized, executed and delivered by Accessity and each of this Agreement and each such related agreement constitutes the valid and binding obligation of Accessity enforceable against Accessity in accordance with its terms, except (x) as the same may be limited by bankruptcy, insolvency, reorganization, moratorium or similar laws now or hereafter in effect relating to creditors' rights generally and (y) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to equitable defenses and to the discretion of the court before which any proceeding therefor may be brought; (iii) Accessity, through its Board of Directors and shareholders, has taken all corporate action necessary for the approval of the execution, delivery and performance of this Agreement by Accessity; (iv) the Accessity Exchange Shares when issued to the Owners in exchange for the PEI Stock, Kinergy Interests and Reenergy Interests, will be duly and validly issued, fully paid and nonassessable; (v) except as otherwise disclosed in any Accessity SEC Documents (as defined below), to the knowledge of such legal counsel, there are no pending or threatened claims or litigation against Accessity; and (vi) neither the execution of this Agreement, nor the consummation of the Share Exchange and the other transactions contemplated hereby or any announcement of the execution of this Agreement or the consummation of the Share Exchange and the other transactions contemplated hereby constitutes or shall constitute a "Triggering Event" or a "Business Combination" as such terms are defined and used in that certain Rights Agreement dated as of December 28, 1998, between Accessity (formerly First Priority Group, Inc.) and North American Transfer Co., as Rights Agent.

(h) an original stock certificate evidencing the ownership by GV Capital Corp. of 150,000 shares of common stock of Accessity, together with a letter from Larry Kaplan confirming that issuance of such shares to GV Capital Corp. shall constitute full payment of a finder's fee for introducing Accessity to PEI and the other Acquired Companies;

(i) evidence, in form and substance reasonably acceptable to each of PEI, Kinergy and Reenergy, that each of the conditions precedent set forth in Article XIII below have been satisfied; and

(j) any and all consents, approvals, notices, filings or recordations of third parties required with respect to the execution and delivery of this Agreement or the transactions contemplated hereby or by any of the agreements, documents or instruments referred to herein.

3.3 Deliveries by Acquired Companies and Owners.

(a) PEI. At the Closing, PEI and the PEI Shareholders and PEI Warrantholders shall deliver to Accessity:

(i) the original PEI Warrants and the original stock certificates representing the PEI Stock, accompanied by stock powers separate from such stock certificates duly executed in blank by the PEI Shareholders evidencing the transfer of PEI Stock to Accessity and, for each married PEI Shareholder that is a resident of California or a resident of any other community property state, a Consent of Spouse in the form attached hereto duly executed by the spouse of such PEI Shareholder;

7

(ii) any and all consents, approvals, notices, filings or recordations of third parties required with respect to the execution and delivery of this Agreement or the transactions contemplated hereby or by any of the agreements, documents or instruments referred to herein;

(iii) evidence, in form and substance reasonably satisfactory to Accessity, that any and all shareholder agreements or similar agreements to which PEI and the PEI Shareholders, or any of them, are a party or to which they or any of them may be subject have been duly terminated;

(iv) a certificate of the president or chief executive officer of PEI certifying that the representations and warranties by PEI set forth in this Agreement and in any certificate or document delivered pursuant to the provisions of this Agreement are true and accurate, on and as of the Closing

Date, and that PEI has performed and complied in all material respects with all agreements, obligations and conditions required by this Agreement to be performed or complied with by it on or prior to the Closing Date;

(v) a copy of the Articles of Incorporation and Bylaws of PEI, each as amended to date, and the resolutions adopted by the Board of Directors of PEI approving, authorizing and directing the execution of this Agreement and the transactions contemplated thereby, each certified by the Secretary of PEI as being in full force and effect on and as of the Closing Date;

(vi) a certificate of the Secretary of State of California to the effect that PEI is a validly existing corporation in good standing under the laws of the State of California and a certificate from the Secretary of State of each other state in which the character of its properties owned or leased or the nature of its activities requires qualification as a foreign corporation doing business in such state to the effect that PEI is a foreign corporation in good standing under the laws of such state; and

(vii) an opinion of legal counsel to PEI to the effect that: (i) PEI is a corporation duly incorporated, validly existing and in good standing under the laws of California and is duly qualified as a foreign corporation in each state in which the character of its properties owned or leased or the nature of its activities requires qualification as a foreign corporation doing business in such state, except where the failure to be so qualified would not have a material adverse effect on PEI; (ii) this Agreement and each related agreement to which PEI is a party has been duly authorized, executed and delivered by PEI and each of this Agreement and each such related agreement constitutes the valid and binding obligation of PEI enforceable against PEI in accordance with its terms, except (x) as the same may be limited by bankruptcy, insolvency, reorganization, moratorium or similar laws now or hereafter in effect relating to creditors' rights generally and (y) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to equitable defenses and to the discretion of the

8

court before which any proceeding therefor may be brought; and (iii) PEI, through its Board of Directors and shareholders, has taken all corporate action necessary for the approval of the execution, delivery and performance of this Agreement by PEI.

(b) Kinergy. At the Closing, Kinergy and the Kinergy Members shall deliver to Accessity:

(i) original certificates, if any have been issued, representing the Kinergy Interests, accompanied by assignments of interest duly executed in blank by the Kinergy Members evidencing the transfer of the Kinergy Interests to Accessity and, for each married Kinergy Member that is a resident of California or a resident of any other community property state, a Consent of Spouse in the form attached hereto as Exhibit E, duly executed by the spouse of such Kinergy Member;

(ii) any and all consents, approvals, notices, filings or recordations of third parties required with respect to the execution and delivery of this Agreement or the transactions contemplated hereby or by any of the agreements, documents or instruments referred to herein;

(iii) evidence, in form and substance reasonably satisfactory to Accessity, that any and all member agreements or similar agreements to which Kinergy and the Kinergy Members, or any of them, are a party or to which they or any of them may be subject have been duly terminated;

(iv) a certificate of the Managers or Managing Members of Kinergy certifying that the representations and warranties by Kinergy set forth in this Agreement and in any certificate or document delivered pursuant to the provisions of this Agreement are true and accurate, on and as of the Closing Date, and that Kinergy has performed and complied in all material respects with all agreements, obligations and conditions required by this Agreement to be performed or complied with by it on or prior to the Closing Date;

(v) a copy of the Articles of Organization and Operating Agreement of Kinergy, each as amended to date, and the resolutions adopted by the Managers or Managing Members of Kinergy and the Kinergy Members approving, authorizing and

directing the execution of this Agreement and the transactions contemplated thereby, each certified by the Managers or Managing Members of Kinergy as being in full force and effect on and as of the Closing Date;

(vi) a certificate of the Secretary of State of Oregon to the effect that Kinergy is a validly existing limited liability company in good standing under the laws of the State of Oregon and a certificate from the Secretary of State of each other state in which the character of its properties owned or leased or the nature of its activities requires qualification as a foreign limited liability company doing business in such state to the effect that Kinergy is a foreign limited liability company in good standing under the laws of such state; and

(vii) an opinion of legal counsel to Kinergy to the effect that: (i) Kinergy is a limited liability company duly organized, validly existing and in good standing under the laws of California and is duly qualified as a foreign limited liability company in each state in which the character of its properties owned or leased or the nature of its activities

9

requires qualification as a foreign limited liability company doing business in such state, except where the failure to be so qualified would not have a material adverse effect on Kinergy; (ii) this Agreement and each related agreement to which Kinergy is a party has been duly authorized, executed and delivered by Kinergy and each of this Agreement and each such related agreement constitutes the valid and binding obligation of Kinergy enforceable against Kinergy in accordance with its terms, except (x) as the same may be limited by bankruptcy, insolvency, reorganization, moratorium or similar laws now or hereafter in effect relating to creditors' rights generally and (y) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to equitable defenses and to the discretion of the court before which any proceeding therefor may be brought; and (iii) Kinergy, through its Managers or Managing Members and the Kinergy Members, has taken all limited liability company action necessary for the approval of the execution, delivery and performance of this Agreement by Kinergy.

(c) Reenergy. At the Closing, Reenergy and the Reenergy Members shall deliver to Accessity:

(i) original certificates, if any have been issued, representing the Reenergy Interests, accompanied by assignments of interest duly executed in blank by the Reenergy Members evidencing the transfer of the Reenergy Interests to Accessity and, for each married Reenergy Member that is a resident of California or a resident of any other community property state, a Consent of Spouse in the form attached hereto as Exhibit E duly executed by the spouse of such Reenergy Member;

(ii) any and all consents, approvals, notices, filings or recordations of third parties required with respect to the execution and delivery of this Agreement or the transactions contemplated hereby or by any of the agreements, documents or instruments referred to herein;

(iii) evidence, in form and substance reasonably satisfactory to Accessity, that any and all member agreements or similar agreements to which Reenergy and the Reenergy Members, or any of them, are a party or to which they or any of them may be subject have been duly terminated;

(iv) a certificate of the Managers or Managing Members of Reenergy certifying that the representations and warranties by Reenergy set forth in this Agreement and in any certificate or document delivered pursuant to the provisions of this Agreement are true and accurate, on and as of the Closing Date, and that Reenergy has performed and complied in all material respects with all agreements, obligations and conditions required by this Agreement to be performed or complied with by it on or prior to the Closing Date;

(v) a copy of the Articles of Organization and Operating Agreement of Reenergy, each as amended to date, and the resolutions adopted by the Managers or Managing Members of Reenergy and the Reenergy Members approving, authorizing and directing the execution of this Agreement and the transactions contemplated thereby, each certified by the Managers or Managing Members of Reenergy as being in full force and effect on and as of the Closing Date; (vi) a certificate of the Secretary of State of California to the effect that Reenergy is a validly existing limited liability company in good standing under the laws of the State of California and a certificate from the Secretary of State of each other state in which the character of its properties owned or leased or the nature of its activities requires qualification as a foreign limited liability company doing business in such state to the effect that Reenergy is a foreign limited liability company in good standing under the laws of such state; and

(vii) an opinion of legal counsel to Reenergy to the effect that: (i) Reenergy is a limited liability company duly organized, validly existing and in good standing under the laws of Oregon and is duly qualified as a foreign limited liability company in each state in which the character of its properties owned or leased or the nature of its activities requires qualification as a foreign limited liability company doing business in such state, except where the failure to be so qualified would not have a material adverse effect on Reenergy; (ii) this Agreement and each related agreement to which Reenergy is a party has been duly authorized, executed and delivered by Reenergy and each of this Agreement and each such related agreement constitutes the valid and binding obligation of Reenergy enforceable against Reenergy in accordance with its terms, except x) as the same may be limited by bankruptcy, insolvency, reorganization, moratorium or similar laws now or hereafter in effect relating to creditors' rights generally and (y) that the remedy of specific performance and injunctive and other forms of equitable relief may be subject to equitable defenses and to the discretion of the court before which any proceeding therefor may be brought; and (iii) Reenergy, through its Managers or Managing Members and the Reenergy Members, has taken all limited liability company action necessary for the approval of the execution, delivery and performance of this Agreement by Reenergy.

3.4 Further Assurances. From time to time after the Closing, and without further consideration, each of the Acquired Companies and Owners shall execute and deliver such other instruments of conveyance, assignment, transfer and delivery, and take such other actions as Accessity may reasonably request in order to more effectively transfer to Accessity, and to place Accessity in possession or control of, the Acquired Companies and to reasonably assist in the collection of any and all such rights, properties and assets, and to enable Accessity to exercise and enjoy all of the rights and benefits with respect thereto.

> ARTICLE IV REPRESENTATIONS AND WARRANTIES OF PEI

PEI hereby represents and warrants to Accessity that the following representations and warranties are true and correct on and as of the date hereof:

4.1 Organization and Good Standing. PEI is a corporation duly organized, validly existing and in good standing under the laws of the State of California, has the power and authority to own, operate and lease its properties and to carry on its business as now conducted, and is duly qualified to do business and is in good standing in each jurisdiction where the character of its properties owned or leased or the nature of its activities make such qualification necessary (each such jurisdiction being listed on Schedule 4.1), except where the failure to be so qualified would not have a material adverse effect on PEI.

4.2 Capitalization. The authorized capital stock of PEI consists solely of 20,000,000 shares of common stock, without

11

par value, and 30,000,000 shares of Preferred Stock, without par value. There are no shares of Preferred Stock issued and outstanding as of the date of this Agreement. A total of 12,252,200 shares of PEI Stock are issued and outstanding as of the date of this Agreement, all of which are held of record and owned by the PEI Shareholders as set forth in Exhibit B. No equity securities of PEI are issued and outstanding as of the date of this Agreement other than the shares of PEI Stock set forth on Exhibit B. Exhibit B sets forth the number of shares of PEI Stock that are held by each PEI Shareholder as of the date of this

Agreement. All issued and outstanding shares of PEI Stock have been duly authorized and validly issued, are fully paid and nonassessable, are not subject to any right of rescission and have been offered, issued, sold and delivered by PEI in compliance with all requirements of applicable laws.

4.3 Power and Authority. PEI has full power and authority to enter into this Agreement, to perform its obligations hereunder and to carry out the transactions contemplated hereby. The execution and delivery of this Agreement, the performance by PEI of its obligations hereunder and the consummation of the transactions contemplated hereby have been duly authorized by all corporate, shareholder and other actions on the part of PEI required by applicable law, PEI's Articles of Incorporation or its Bylaws. This Agreement constitutes the legal, valid and binding obligation of PEI, enforceable against it in accordance with its terms.

4.4 Options/Rights. Except as set forth in Exhibit B or in Schedule 4.4, there are no (i) stock appreciation rights, options, warrants, calls, rights, commitments, conversion privileges or preemptive or other rights or agreements outstanding to purchase or otherwise acquire any shares of PEI, specifically including the PEI Stock (collectively, "PEI Capital Stock"), (ii) securities or debt convertible into or exchangeable for PEI Capital Stock or obligating PEI to grant, extend or enter into any such option, warrant, call, commitment, conversion privileges or preemptive or other right or agreement, or (iii) voting agreements, registration rights, rights of first refusal, preemptive rights, co-sale rights, or other restrictions applicable to any outstanding securities of PEI.

4.5 Subsidiaries. Except as set forth in Schedule 4.5, PEI does not have any subsidiaries or any equity interest, direct or indirect, in, or loans to, any corporation, partnership, joint venture, limited liability company or other business entity.

4.6 No Violation. Neither the execution and delivery of this Agreement nor the performance by PEI of its obligations hereunder nor the consummation of the transactions contemplated hereby will (a) contravene any provision of the Articles of Incorporation or Bylaws of PEI; (b) violate, be in conflict with, constitute a default under, permit the termination of, cause the acceleration of the maturity of any debt or obligation of PEI under, require (except as disclosed on Schedule 4.6 hereto) the consent of any other party to, constitute a breach of, create a loss of a material benefit under, or result in the creation or imposition of any Lien, upon any property or assets of PEI under, any mortgage, indenture, lease, contract, agreement, instrument or commitment to which PEI is a party or by which it or he or any of its or his respective assets or properties may be bound; (c) to the knowledge of PEI, violate any statute or law or any judgment, decree, order, regulation or rule of any court or Governmental Body to which PEI or the PEI Business is subject or by which PEI, any of its assets or properties are bound; or (d) result in the loss of any license, privilege or certificate benefiting PEI or the PEI Business.

12

4.7 Consents and Approvals. Except as set forth on Schedule 4.7 hereto, no consent, approval or authorization of, or declaration, filing or registration with, any governmental or regulatory authority or any other third party is required to be made or obtained by PEI in connection with the execution, delivery or performance of this Agreement.

4.8 Litigation. There are no Actions to which PEI is a party, including, without limitation, Actions for personal injury, products liability, wrongful death or other tortious conduct, or breach of warranty arising from or relating to materials, commodities, products or goods used, transferred, processed, manufactured, sold, distributed or shipped by PEI (a) involving or relating to PEI or any of its assets, properties or rights, or (b) pending, or, to PEI's knowledge, threatened, against PEI, or any of their respective assets, properties or rights, before any court, arbitrator or administrative or Governmental Body which, if adversely resolved, would have a material adverse effect on the PEI Business.

4.9 PEI Financial Statements. PEI has previously delivered to Accessity the unaudited balance sheet of PEI as of December 31, 2003 and the related statements of income and changes in financial position or cash flows, as appropriate, for the period then ended (the "Unaudited 2003 PEI Financial Statements"), together with an unaudited balance sheet of PEI as of March 31,

2004 and the related statements of income and changes in financial position or cash flows, as appropriate, for the period then ended and, subsequent to the date hereof, PEI will deliver to Accessity the audited balance sheet of PEI as of December 31, 2003 and the related statements of income and changes in financial position or cash flows, as appropriate, for the period then ended (the "Audited 2003 PEI Financial Statements") (all such financial statements, excluding, upon delivery to Accessity of the Audited 2003 PEI Financial Statements, the Unaudited 2003 PEI Financial Statements, are hereinafter collectively referred to as the "PEI Financial Statements"). The PEI Financial Statements, together with the notes thereto, (i) were compiled from the books and records of PEI regularly maintained by management and used to prepare the financial statements of PEI, (ii) were prepared in accordance with GAAP consistently applied throughout the period then ended and all periods prior to that period; and (iii) present fairly and accurately the financial condition of PEI for the period or as of the dates thereof, subject, where appropriate, to normal year-end audit adjustments, in each case in accordance with GAAP consistently applied during the period covered.

4.10 No Undisclosed Liabilities. PEI has, and on the Closing Date will have, no Liabilities other than those which (a) are fully reflected reserved against in the PEI Financial Statements, (b) have been incurred since March 31, 2004 in the ordinary course of business in amounts and for terms consistent, individually and in the aggregate, with the past practice of PEI or (c) have been specifically disclosed in the Schedules hereto by reference to the specific section of this Agreement to which such disclosure relates.

4.11 Taxes and Tax Returns. All of the tax returns and reports of PEI required by applicable law to be filed prior to the date hereof have been duly filed and all taxes shown as due thereon have been paid. There are in effect no waivers of the applicable statutes of limitations for any federal, state, local or foreign taxes for any period. No liability for any federal, state, local or foreign income, sales, use, withholding, payroll, franchise, real property or personal property taxes is pending, and there is no proposed liability for any such taxes to be imposed upon the properties or assets of PEI. PEI does not have any liability for any federal, state, local or foreign income, sales, use, withholding, payroll, franchise, real property or personal property taxes, assessments, amounts, interest or penalties of any nature whatsoever other than as shown on the March 31, 2004 PEI Financial Statements and there is no basis for any additional claim or assessment other than with respect to liabilities for taxes which may have accrued since the date

13

of the March 31, 2004 PEI Financial Statements in the ordinary course of business and reserved against on the books and records of PEI compiled in accordance with generally accepted accounting principles which have been consistently applied to the Closing Date. The provisions for taxes reflected in the March 31, 2004 PEI Financial Statements are adequate for federal, state, county and local taxes for the period ended on December 31, 2003 and for all prior periods, whether disputed or undisputed. There are no present disputes about taxes of any nature payable by PEI. PEI has never filed, and will not file on or before the Closing Date, any consent under section 341(f) of the Code. PEI's tax returns have never been audited by any taxation authority.

4.12 Absence of Certain Changes. Except as set forth on Schedule 4.12, since December 31, 2003, PEI has conducted the PEI Business only in the ordinary course and consistent with prior practices and has not:

(a) suffered any material adverse change in its condition (financial or otherwise), results of operations, assets, liabilities, reserves, the PEI Business, or operations;

(b) suffered any damage, destruction or loss, whether covered by insurance or not, materially adversely affecting the PEI Business, operations, assets, or condition (financial or otherwise);

(c) paid, discharged or satisfied any Liability or other expenses, other than the payment, discharge or satisfaction of the Liabilities described in Section 4.10 at the time the same were due and payable and in the ordinary course of business;

(d) paid or otherwise made any contribution to any profit-sharing or pension plan or other Employee Benefit Plan (as defined in Section 4.17 below);

(e) mortgaged or pledged, or permitted the imposition of any Lien upon, any of its properties or assets (real, personal or mixed, tangible or intangible), other than those incurred in the ordinary course of business or otherwise listed on Schedule 4.12 hereto;

(f) cancelled or compromised any debts, or waived or permitted to lapse any material claims or rights, or sold, assigned, transferred or otherwise disposed of, other than in the ordinary course, any of its properties or assets (real, personal or mixed, tangible or intangible);

(g) disposed of or permitted to lapse any rights to the use of any patent, registered trademark, service mark, trade name or copyright, or disposed of or disclosed to any person any trade secret, formula, process or know-how material to the PEI Business not theretofore a matter of public knowledge;

(h) except as disclosed on Schedule 4.12, granted any increase in the compensation of any officer, employee or consultant of PEI (including any such increase pursuant to any bonus, pension, profit-sharing or other plan or commitment) or any increase in the compensation payable or to become payable to any officer, employee or consultant;

(i) other than commitments, transactions and expenditures in connection with engineering work and other preliminary site work at the site of the ethanol plant currently being developed by PEI, the anticipated expenditures for which are set forth on Schedule 4.12, and commitments, transactions and expenditures in connection with the repair of the grain facility of

14

PEI located in Madera, California due to the fire that occurred at such facility in the first quarter of 2004, the expenditures for which are anticipated to be reimbursed under applicable insurance coverage, entered into any commitment or transaction not in the ordinary course of business or made any capital expenditure or commitment for any additions to property, plant or equipment, except commitments, transactions or capital expenditures which do not in any single case exceed \$25,000 or in the aggregate exceed \$50,000;

(j) made any change in any method of accounting or accounting practice (including, without limitation, any change in depreciation or amortization policies or rates);

(k) paid, loaned or advanced any amount to, or sold, transferred or leased any properties or assets (real, personal or mixed, tangible or intangible) to, or entered into any agreement or arrangement with, any of its officers, directors, employees, shareholders, or any family member or Affiliate of any of its officers, directors, employees or shareholders, or any officer, director, employee or shareholder of any such Affiliate;

(1) declared, set aside, paid or made any dividend or other distribution or payment in respect of the capital stock of PEI, or any direct or indirect redemption, purchase or other acquisition of any of its shares of capital stock;

(m) knowingly waived or released any right or claim of PEI;

(n) received a commencement notice or, to the knowledge of PEI, received any threat of commencement, of any civil or criminal litigation, investigation or proceeding against PEI;

(o) experienced any labor trouble or, to the knowledge of PEI, received any claim of wrongful discharge or worker's compensation claim;

(p) agreed, whether in writing or otherwise, to take any action referred to in and prohibited by this Section 4.12; or

(q) become aware of any other event or condition that has had or would reasonably be expected to have a material adverse effect on the condition (financial or otherwise), results of operations, assets, liabilities, reserves, the PEI Business, the PEI Intellectual Property (as defined below) or the operations of PEI.

4.13 Title to Properties; Encumbrances.

(a) PEI has good and marketable title to all of its properties and assets (real, personal or mixed, tangible or intangible), including without limitation the PEI Intellectual Property. None of PEI's properties or assets is subject to any Lien, except Permitted Liens (including the Liens set forth on Schedule 4.13), none of which adversely affects the PEI Business or the continued operations of PEI.

(b) All material property and assets (real, personal or mixed, tangible or intangible) used or required by PEI in the conduct of the PEI Business are fully owned by PEI (except to the extent of any Permitted Liens). All such property and assets, or the leases or

15

licenses thereof, constitute all property, assets and contractual rights necessary for the conduct of the PEI Business as presently conducted.

4.14 Leases.

(a) Schedule 4.14 contains a true and complete list of:

(i) all leases pursuant to which PEI leases or subleases any real property interests, whether as lessor, lessee, sublessor or sublessee;

(ii) all leases pursuant to which PEI leases any type of personal property;

(iii) all leases pursuant to which PEI leases any vehicles or related equipment; and

(iv) all leases pursuant to which PEI leases to others any type of property.

(b) Each such lease described on Schedule 4.14 is the legal, valid and binding obligation of PEI and, to the knowledge of PEI, the other parties thereto, enforceable in accordance with their respective terms, and is in full force and effect. PEI is not in default under any such lease, and PEI has not received any notice from any person or entity asserting that PEI is in default under any such lease, and no events or circumstances exist which, with notice or the passage of time or both, would constitute a default under any such lease.

4.15 Intellectual Property. PEI owns all right, title and interest in, or has the right to use, sell or license all patent applications, patents, trademark applications, trademarks, service marks, trade names, copyright applications, copyrights, trade secrets, know-how, technology, customer lists, proprietary processes and formulae, all source and object code, algorithms, inventions, development tools and all documentation and media constituting, describing or relating to the above, including, without limitation, manuals, memoranda and records and other intellectual property and proprietary rights used in or reasonably necessary or required for the conduct of its respective business as presently conducted (collectively, the "PEI Intellectual Property").

4.16 Compliance with Laws. PEI has not been charged with, and, to PEI's knowledge, PEI is not threatened with or under any investigation with respect to, any charge concerning any violation of any provision of any federal, state, local or foreign law, regulation, ordinance, order or administrative ruling affecting the PEI Business or PEI, and PEI is not in default with respect to any order, writ, injunction or decree of any court, agency or instrumentality affecting the PEI Business or PEI. To PEI's knowledge, PEI is not in violation of any federal, state, local or foreign law, ordinance or regulation or any other requirement of any Governmental Body or regulatory body, court or arbitrator applicable to the PEI Business. Without limiting the generality of the foregoing, PEI is in compliance in all material respects with all Occupational Safety and Health Laws, including those rules and regulations promulgated by OSHA, except where such non-compliance would not have a material adverse effect on PEI or the PEI Business.

4.17 Employee Benefit Plans.

(a) Schedule 4.17 contains a true and complete list and description of each pension, retirement, severance, welfare, profit-sharing, stock purchase, stock option, vacation, deferred compensation, bonus or other incentive plan, or other employee benefit program, arrangement, agreement or understanding, or medical, vision, dental or other health plan, or life insurance or disability plan, retiree medical or life insurance plan or any other employee benefit plans, including, without limitation, any "employee benefit plan" (as defined in Section 3(3) of ERISA), to which PEI contributes or is a party or by which it is bound or under which it may have liability and under which employees or former employees of PEI (or their beneficiaries) are eligible to participate or derive a benefit. Each employee benefit plan which is a "group health plan" (as such term is defined in Section 5000(b)(i) of the Code) satisfies the applicable requirements of Section 4980B of the Code. Except as described on Schedule 4.17, PEI has no formal plan or commitment, whether legally binding or not, to create any additional plan, practice or agreement or modify or change any existing plan, practice or agreement that would affect any of its employees or terminated employees. Benefits under all employee benefit plans are as represented and have not been and will not be increased subsequent to the date copies of such plans have been provided.

(b) PEI does not contribute to or have any obligation to contribute to, has not at any time contributed to or had an obligation to contribute to, sponsor or maintain, and has not at any time sponsored or maintained, a "multi-employer plan" (within the meaning of Section 3(37) of ERISA) for the benefit of employees or former employees of PEI.

(c) PEI has, in all material respects, performed all obligations, whether arising by operation of law, contract, or past custom, required to be performed under or in connection with the employee benefit plans disclosed on Schedule 4.17 (individually, a "PEI Employee Benefit Plan" and, collectively, the "PEI Employees Benefit Plans"), and PEI has no knowledge of any default or violation by any other party with respect thereto.

(d) There are no Actions, suits or claims (other than routine claims for benefits) pending, or, to PEI's knowledge, threatened, against any PEI Employee Benefit Plan or against the assets funding any PEI Employee Benefit Plan.

(e) PEI neither maintains nor contributes to any "employee welfare benefit" (as such term is defined in Section 3(i) of ERISA) plan which provides any benefits to retirees or former employees of PEI.

4.18 Employment Law Matters.

(a) PEI (i) is in material compliance with all applicable laws respecting employment, employment practices, terms and conditions of employment and wages and hours; (ii) is in material compliance with all applicable laws and regulations relating to the employment of aliens or similar immigration matters; and (iii) is not engaged in any unfair labor practice, including, but not limited to, discrimination or wrongful discharge.

(b) PEI has not at anytime had, nor to PEI's knowledge, is there now threatened, a strike, picket, work stoppage, work slowdown or other labor trouble, against or directly affecting PEI that had or would reasonably be expected to have a material adverse effect on the PEI Business or PEI.

17

(c) None of the employees of PEI is represented by a labor union, and no petition has been filed or proceedings instituted by any employee or group of employees with any labor relations board seeking recognition of a bargaining representative. PEI is not a party to any multi-employer collective bargaining agreement covering any of its employees.

(d) There are no controversies or disputes (including any union grievances or arbitration proceeding) pending, or, to PEI's knowledge, threatened, between PEI and any employees of PEI (or any union or other representative of such employees). No unfair labor practice complaints have been filed against PEI with the National Labor Relations Board or any other Governmental Body or administrative body, and PEI has not received any written notice or communication reflecting an intention or a threat to file any such complaint. 4.19 Contracts and Commitments.

(a) Together with the leases set forth on Schedule 4.14, the insurance policies set forth on Schedule 4.23, and the PEI Employee Benefit Plans and commitments set forth on Schedule 4.17, Schedule 4.19 contains a true and complete list and description (stated without duplication), of:

(i) all contracts (including, without limitation, letters of credit, and obligations for borrowed money) and commitments of PEI which are material to the operations, business, prospects or condition (financial or otherwise) of PEI;

(ii) all consulting agreements (whether written or oral), regardless of amounts or duration;

(iii) all material contracts or commitments (whether written or oral) with distributors, brokers, manufacturer's representatives, sales representatives, service or warranty representatives, customers and other persons, firms, corporations or other entities engaged in the sale, distribution, service or repair of PEI's products;

(iv) all contracts relating to construction-in-progress of capital assets; and

(v) all joint venture, licensing, profit sharing, royalty or similar agreements or arrangements to which PEI is a party in any way associated with the manufacture, marketing, sale or distribution of any products or provision of any services of PEI.

(b) PEI has delivered to Accessity true and complete copies of all of the documents identified on Schedule 4.19 (collectively, the "PEI Material Contracts") and shall deliver true and complete copies of all such other agreements, instruments and documents as Accessity may reasonably request relating to the operation, ownership or conduct of the PEI Business.

(c) PEI is not a party to any written agreement that would restrict it from carrying on the PEI Business anywhere in the world.

18

(d) PEI is not a party to any "take-or-pay" contracts.

(e) Except as identified on Schedule 4.19, PEI is not a party to any employment agreements, arrangements and commitments, including severance or termination arrangements and commitments (whether written or oral), between PEI and any employees of PEI.

(f) PEI is not, and to the knowledge of PEI and the PEI Majority Shareholders, no other party is, in default under or in breach or violation of, nor has PEI received notice of any asserted claim of default by PEI or by any other party under, or a breach or violation of, any of the PEI Material Contracts.

4.20 No Brokers. Except as disclosed on Schedule 4.20, PEI is not obligated for the payment of fees or expenses of any investment banker, broker or finder in connection with the origin, negotiation or execution of this Agreement in connection with any exchange of stock transaction provided for herein.

4.21 Environmental Matters. PEI is in compliance in all material respects with all Environmental Laws. There is no Action pending before any court, Governmental Body or board or other forum or threatened by any person or entity (i) for noncompliance by PEI with any Environmental Law (ii) relating to the release into the environment by PEI of any pollutant, toxic or hazardous material or waste generated by PEI, whether or not occurring at or on a site owned, leased or operated by PEI. There has not been by PEI, nor to the knowledge of PEI has there been at all, any past, storage, disposal, generation, manufacture, refinement, transportation, production or treatment of any hazardous materials or substances at, upon or from the facilities occupied or used by PEI and any other real property presently or formerly owned by, used by or leased to or by PEI, any predecessor of PEI (collectively, the "PEI Property"). To the knowledge of PEI, neither PEI nor any properties owned or operated by PEI has been or is in violation or is otherwise liable under, any Environmental Law. To the knowledge of PEI, there are no asbestos-containing materials, underground storage tanks or polychlorinated biphenyls (PCBs) located on the PEI Property. To the knowledge of PEI, there has been no spill, discharge, leak, emission, injection, disposal, escape, dumping or release of any kind on, beneath or above the PEI Property or into the environment surrounding such PEI Property of any hazardous materials or substances in violation of any Environmental Law or requiring any remedial action. To the knowledge of PEI, PEI has all permits, registrations, approvals and licenses required by any Governmental Body under any Environmental Law to be obtained by PEI in connection with the conduct of the business of PEI as presently conducted.

4.22 Bank Accounts. Schedule 4.22 sets forth the names and locations of all banks, trust companies, savings and loan associations, and other financial institutions at which PEI maintains accounts of any nature and the names of all persons authorized to draw thereon or make withdrawals therefrom.

4.23 Insurance. Schedule 4.23 sets forth a true and complete list and description of (a) all of PEI's self-insurance practices and items covered by such self-insurance and (b) all policies of fire, liability, worker's compensation and other forms of insurance owned or held by PEI. No installment premiums are due under the policies set forth on Schedule 4.23 or, if installment premiums shall be due and owing under such policies prior to the Closing Date, such premiums shall have been paid up-to-date prior to the Closing Date. All such policies are in full

19

force and effect, insure against risks and liabilities to the extent and in the manner deemed appropriate and sufficient by PEI in its reasonable business judgment, and, other than the receipt of a notice of nonrenewal of a general liability insurance policy expiring in June 2004 applicable to the grain facility of PEI located in Madera, California due to a fire that occurred at such facility in the first quarter of 2004, PEI has not received any notice of cancellation with respect thereto. To PEI's knowledge, PEI is not in default with respect to any provision contained in any such policy and has not failed to give any notice or present any claim under any such policy in a timely fashion.

4.24 Suppliers and Customers. PEI does not have any knowledge that any supplier or customer or group of related suppliers or customers of PEI has canceled or otherwise terminated or threatened to cancel or otherwise terminate, its relationship with PEI, which termination would have a material adverse effect on the PEI Business or PEI, or that any such supplier or customer or group of related suppliers or customers expects to reduce its business with PEI by reason of the transactions contemplated by this Agreement or for any other reason whatsoever.

4.25 Licenses, Permits and Authorizations. PEI has all necessary Permits for the use and ownership or leasing of its properties and assets as currently operated, used, owned or leased (including, without limitation, the operation of a plant to produce up to 40 million gallons of ethanol per year), except for such Permits as to which the lack thereof does not and would not have a material adverse effect on PEI or the PEI Business and except for certain non-discretionary grading, foundation and building Permits to be obtained by PEI in connection with the construction of an ethanol plant, as more particularly described on Schedule 4.25. All of the Permits are valid, in full force and effect and in good standing. Schedule 4.25 contains a true and complete list and description of all the Permits. There is no claim or Action pending, or, to PEI's knowledge, threatened, which disputes the validity of any such Permit or threatens to revoke, cancel, suspend or limit any such Permit.

4.26 Accounts Receivable. All accounts receivable of PEI shown on the PEI Financial Statements and all accounts receivable created after March 31, 2004, subject to reserves created in the ordinary course of business on a basis consistent with the past practices and policies of PEI and otherwise in accordance with generally accepted accounting principles, (a) have been collected or (b) to PEI's knowledge, are valid and enforceable, arose from bona-fide sales to third parties in the ordinary course of business, and are collectible at the aggregate recorded amounts thereof on the books of PEI.

4.27 Condition of Tangible Assets. Except as disclosed on Schedule 4.12, PEI's facilities and tangible assets, including, without limitation,

machinery, equipment, vehicles, furniture, plants and buildings, are in good operating condition and repair (ordinary wear and tear excepted) and are adequate for the uses to which they have been put by PEI in the ordinary course of business, except for parts or repairs of an immaterial nature in the aggregate, and PEI has not received any notice that any of such facilities or assets is in need of substantial maintenance or repair.

4.28 Disclosure. No representation or warranty of PEI in this Agreement (including, without limitation, the Schedules of PEI hereto) contains or will contain any untrue statement of a material fact or omits or will omit to state any material fact necessary to make the statements herein or therein not misleading.

20

ARTICLE V REPRESENTATIONS AND WARRANTIES OF THE PEI SHAREHOLDERS

Each of the PEI Shareholders, severally and not jointly, hereby represents and warrants to Accessity that the following representations and warranties are true and correct on and as of the date hereof:

5.1 Power and Authority. Such PEI Shareholder has full power and authority to enter into this Agreement, perform its respective obligations hereunder, and carry out the transactions contemplated hereby. This Agreement constitutes the legal, valid and binding obligation of such PEI Shareholder, enforceable against such PEI Shareholder in accordance with its terms.

5.2 No Violation. Neither the execution and delivery of this Agreement nor the performance by such PEI Shareholder of its respective obligations hereunder nor the consummation of the transactions contemplated hereby will violate, or be in conflict with, or constitute a default under, any mortgage, indenture, lease, or any agreement, instrument or commitment to which such PEI Shareholder is a party.

5.3 Title; Consents and Approvals; No Claims. Such PEI Shareholder is the owner, beneficially and of record, of its shares of PEI Stock, free and clear of all Liens. To such PEI Shareholder's knowledge, no consent, approval or authorization of, or declaration, filing or registration with, any governmental or regulatory authority or any other third party is required to be made or obtained by such PEI Shareholder in connection with the execution, delivery or performance of this Agreement. To such PEI Shareholder's knowledge, there is no claim, action, litigation, suit, cause of action or other proceeding pending or threatened before any federal, state or local court, governmental agency or regulatory body against such PEI Shareholder which seeks or may seek, directly or indirectly, (a) to invalidate or set aside, in whole or in part, this Agreement, (b) to restrain, prohibit, invalidate or set aside, in whole or in part, the consummation of the transactions contemplated hereby or (c) to obtain substantial damages in connection therewith.

5.4 Securities Law Compliance.

(a) Such PEI Shareholder has a net worth sufficient to bear the economic risk (including the entire loss) of its investment made in the Accessity Exchange Shares;

(b) Such PEI Shareholder is an "accredited investor," as such term is defined in Rule 501 of Regulation D promulgated under the rules and regulations of the Securities Act;

(c) Such PEI Shareholder has adequate means of providing for its current cash needs and personal contingencies and has no need for liquidity in this investment in the Accessity Exchange Shares and has no reason to anticipate any change in its personal circumstances, financial or otherwise, which may cause or require any sale or distribution by such PEI Shareholder or all or any part of the Accessity Exchange Shares acquired by it herein;

(d) by reason of such PEI Shareholder's business or financial experience or the business or financial experience of such PEI Shareholder's professional advisor(s) who are unaffiliated with and who are not compensated by Accessity or any affiliate or selling agent of Accessity, directly or indirectly, such PEI Shareholder has the capacity to protect its own interests in connection with an investment in the Accessity Exchange Shares;

(e) Such PEI Shareholder understands that the he, she or it is acquiring Accessity Exchange Shares without being furnished any prospectus or offering circular, other than a copy of this Agreement, a copy of the Proxy Statement (as defined in Section 11.6 below) and a copy of the Owner Disclosure Document (as defined in Section 11.7 below);

(f) No representations or warranties have been made to such PEI Shareholder by Accessity or any employee or agent of Accessity and in entering into this Agreement, such PEI Shareholder is not relying on any information, other than as a result of the independent investigation of Accessity by such PEI Shareholder, and no guarantee of any profit or return on its investment made in the Accessity Exchange Shares has been made to such PEI Shareholder;

(g) In evaluating the merits and risk of this investment, such PEI Shareholder has relied on the advice of its personal tax advisor, investment advisor and/or legal counsel;

(h) Such PEI Shareholder is aware that the Accessity Exchange Shares have not been registered or qualified, nor is registration or qualification contemplated (except where such PEI Shareholder is a party to a Registration Rights Agreement with PEI, to the extent provided for therein), with the SEC under the Securities Act or any state securities law. Accordingly, the Accessity Exchange Shares may be sold or otherwise transferred or hypothecated only if they are subsequently registered or qualified under the Securities Act or applicable laws or if, in the opinion of counsel, an exemption from registration or qualification thereunder is available and the transaction will not jeopardize the availability of the exemptions under applicable federal and state securities laws relied upon by Accessity in connection with the offering in which such PEI Shareholder acquired its Accessity Exchange Shares;

(i) Such PEI Shareholder acknowledges that the Accessity Exchange Shares were not offered by means of any general solicitation or advertising;

(j) Such PEI Shareholder is acquiring its Accessity Exchange Shares solely for its own account, for investment purposes only, and not with an intent to sell, or for resale in connection with any distribution of all or any portion of the Accessity Exchange Shares within the meaning of the Securities Act; and

(k) The address of such PEI Shareholder set forth on the signature pages hereto is the principal residence of such PEI Shareholder, if such PEI Shareholder is an individual, or the principal business address of such PEI Shareholder, if such PEI Shareholder is a business or other entity, and that all offers to such PEI Shareholder have been made only in the state specified in such address.

> ARTICLE VI REPRESENTATIONS AND WARRANTIES OF KINERGY

Kinergy hereby represents and warrants to Accessity that the following representations and warranties are true and correct on and as of the date hereof:

22

6.1 Organization and Good Standing. Kinergy is a limited liability company duly organized, validly existing and in good standing under the laws of the State of Oregon, has the power and authority to own, operate and lease its properties and to carry on its business as now conducted, and is duly qualified to do business and is in good standing in each jurisdiction where the character of its properties owned or leased or the nature of its activities make such qualification necessary (each such jurisdiction being listed on Schedule 6.1), except where the failure to be so qualified would not have a material adverse effect on Kinergy.

6.2 Capitalization. The limited liability company membership interests of Kinergy are owned by the Kinergy Members as set forth in Exhibit C. No

limited liability company membership interests of Kinergy are outstanding as of the date of this Agreement other than the Kinergy Interests set forth on Exhibit C. The Kinergy Interests have been duly authorized and validly issued, are fully paid and nonassessable, are not subject to any right of rescission and have been offered, issued, sold and delivered by Kinergy in compliance with all requirements of applicable laws.

6.3 Power and Authority. Kinergy has full power and authority to enter into this Agreement, to perform its obligations hereunder and to carry out the transactions contemplated hereby. The execution and delivery of this Agreement, the performance by Kinergy of its obligations hereunder and the consummation of the transactions contemplated hereby have been duly authorized by the Managers or the Managing Members and the Kinergy Members as required by applicable law, Kinergy's Articles of Organization or its Operating Agreement. This Agreement constitutes the legal, valid and binding obligation of Kinergy, enforceable against it in accordance with its terms.

6.4 Options/Rights. Except as set forth in Schedule 6.4, there are no (i) options, warrants, calls, rights, commitments, conversion privileges or preemptive or other rights or agreements outstanding to purchase or otherwise acquire any limited liability company membership interests of Kinergy, (ii) securities or debt convertible into or exchangeable for Kinergy limited liability company membership interests or obligating Kinergy to grant, extend or enter into any such option, warrant, call, commitment, conversion privileges or preemptive or other right or agreement, or (iii) voting agreements, rights of first refusal, preemptive rights, co-sale rights, or other restrictions applicable to any outstanding limited liability company membership interests of Kinergy.

6.5 Subsidiaries. Kinergy does not have any subsidiaries or any equity interest, direct or indirect, in, or loans to, any corporation, partnership, joint venture, limited liability company or other business entity.

6.6 No Violation. Neither the execution and delivery of this Agreement nor the performance by Kinergy of its obligations hereunder nor the consummation of the transactions contemplated hereby will (a) contravene any provision of the Articles of Organization or Operating Agreement of Kinergy; (b) violate, be in conflict with, constitute a default under, permit the termination of, cause the acceleration of the maturity of any debt or obligation of Kinergy under, require (except as disclosed on Schedule 6.6 hereto) the consent of any other party to, constitute a breach of, create a loss of a material benefit under, or result in the creation or imposition of any Lien, upon any property or assets of Kinergy under, any mortgage, indenture, lease, contract, agreement, instrument or commitment to which Kinergy is a party or by which it or any of its assets or properties may be bound; (c) to the knowledge of Kinergy,

23

violate any statute or law or any judgment, decree, order, regulation or rule of any court or Governmental Body to which Kinergy or the Kinergy Business is subject or by which Kinergy, any of its assets or properties are bound; or (d) result in the loss of any license, privilege or certificate benefiting Kinergy or the Kinergy Business.

6.7 Consents and Approvals. Except as set forth on Schedule 6.7 hereto, no consent, approval or authorization of, or declaration, filing or registration with, any governmental or regulatory authority or any other third party is required to be made or obtained by Kinergy in connection with the execution, delivery or performance of this Agreement.

6.8 Litigation. There are Actions to which Kinergy is a party, including, without limitation, Actions for personal injury, products liability, wrongful death or other tortious conduct, or breach of warranty arising from or relating to materials, commodities, products or goods used, transferred, processed, manufactured, sold, distributed or shipped by Kinergy (a) involving or relating to Kinergy, or any its assets, properties or rights, or (b) pending, or, to Kinergy's knowledge, threatened, against Kinergy, or any of its assets, properties or rights, before any court, arbitrator or administrative or Governmental Body which, if adversely resolved, would have a material adverse effect on the Kinergy Business.

6.9 Kinergy Financial Statements. Kinergy has previously delivered to Accessity the unaudited balance sheet of Kinergy as of December 31, 2003 and the

related statements of income and changes in financial position or cash flows, as appropriate, for the period then ended (the "Unaudited 2003 Kinergy Financial Statements"), together with an unaudited balance sheet of Kinergy as of March 31, 2004 and the related statements of income and changes in financial position or cash flows, as appropriate, for the period then ended and, subsequent to the date hereof, Kinergy will deliver to Accessity the audited balance sheet of Kinergy as of December 31, 2003 and the related statements of income and changes in financial position or cash flows, as appropriate, for the period then ended (the "Audited 2003 Kinergy Financial Statements") (all such financial statements, excluding, upon delivery to Accessity of the Audited 2003 Kinergy Financial Statements, the Unaudited 2003 Kinergy Financial Statements, are hereinafter collectively referred to as the "Kinergy Financial Statements"). The Kinergy Financial Statements, together with the notes thereto, (i) were compiled from the books and records of Kinergy regularly maintained by management and used to prepare the financial statements of Kinergy, (ii) were prepared in accordance with GAAP consistently applied throughout the period then ended and all periods prior to that period; and (iii) present fairly and accurately the financial condition of Kinergy for the period or as of the dates thereof, subject, where appropriate, to normal year-end audit adjustments, in each case in accordance with GAAP consistently applied during the period covered.

6.10 No Undisclosed Liabilities. Kinergy has, and on the Closing Date will have, no Liabilities other than those which (a) are fully reflected reserved against in the Kinergy Financial Statements, (b) have been incurred since March 31, 2004 in the ordinary course of business in amounts and for terms consistent, individually and in the aggregate, with the past practice of Kinergy or (c) have been specifically disclosed in the Schedules hereto by reference to the specific section of this Agreement to which such disclosure relates.

6.11 Taxes and Tax Returns. All of the tax returns and reports of Kinergy required by applicable law to be filed prior to the date hereof have been duly filed and all taxes shown as due thereon have been paid. There are in effect no waivers of the applicable statutes of

24

limitations for any federal, state, local or foreign taxes for any period. No liability for any federal, state, local or foreign income, sales, use, withholding, payroll, franchise, real property or personal property taxes is pending, and there is no proposed liability for any such taxes to be imposed upon the properties or assets of Kinergy. Kinergy does not have any liability for any federal, state, local or foreign income, sales, use, withholding, payroll, franchise, real property or personal property taxes, assessments, amounts, interest or penalties of any nature whatsoever other than as shown on the March 31, 2004 Kinergy Financial Statements and there is no basis for any additional claim or assessment other than with respect to liabilities for taxes which may have accrued since the date of the March 31, 2004 Kinergy Financial Statements in the ordinary course of business and reserved against on the books and records of Kinergy compiled in accordance with generally accepted accounting principles which have been consistently applied to the Closing Date. The provisions for taxes reflected in the March 31, 2004 Kinergy Financial Statements are adequate for federal, state, county and local taxes for the period ended on December 31, 2003 and for all prior periods, whether disputed or undisputed. There are no present disputes about taxes of any nature payable by Kinergy. Kinergy has never filed, and will not file on or before the Closing Date, any consent under section 341(f) of the Code. Kinergy's tax returns have never been audited by any taxation authority.

6.12 Absence of Certain Changes. Except as set forth on Schedule 6.12, since December 31, 2003, Kinergy has conducted the Kinergy Business only in the ordinary course and consistent with prior practices and has not:

(a) suffered any material adverse change in its condition (financial or otherwise), results of operations, assets, liabilities, reserves, the Kinergy Business, or operations;

(b) suffered any damage, destruction or loss, whether covered by insurance or not, materially adversely affecting the Kinergy Business, operations, assets, or condition (financial or otherwise);

(c) paid, discharged or satisfied any Liability or other expenses, other than the payment, discharge or satisfaction of the Liabilities described in Section 6.10 at the time the same were due and payable and in the ordinary

course of business;

(d) paid or otherwise made any contribution to any profit-sharing or pension plan or other Employee Benefit Plan (as defined in Section 6.17 below);

(e) mortgaged or pledged, or permitted the imposition of any Lien upon, any of its properties or assets (real, personal or mixed, tangible or intangible), other than those incurred in the ordinary course of business or otherwise listed on Schedule 6.12 hereto;

(f) cancelled or compromised any debts, or waived or permitted to lapse any material claims or rights, or sold, assigned, transferred or otherwise disposed of, other than in the ordinary course, any of its properties or assets (real, personal or mixed, tangible or intangible);

(g) disposed of or permitted to lapse any rights to the use of any patent, registered trademark, service mark, trade name or copyright, or disposed of or disclosed to any person any trade secret, formula, process or know-how material to the Kinergy Business not theretofore a matter of public knowledge;

25

(h) except as disclosed on Schedule 6.12, granted any increase in the compensation of any officer, employee or consultant of Kinergy (including any such increase pursuant to any bonus, pension, profit-sharing or other plan or commitment) or any increase in the compensation payable or to become payable to any officer, employee or consultant;

(i) entered into any commitment or transaction not in the ordinary course of business or made any capital expenditure or commitment for any additions to property, plant or equipment, except commitments, transactions or capital expenditures which do not in any single case exceed \$25,000 or in the aggregate exceed \$50,000;

(j) made any change in any method of accounting or accounting practice (including, without limitation, any change in depreciation or amortization policies or rates);

(k) paid, loaned or advanced any amount to, or sold, transferred or leased any properties or assets (real, personal or mixed, tangible or intangible) to, or entered into any agreement or arrangement with, any of its officers, employees, members, or any family member or Affiliate of any of its officers, employees or members, or any officer, employee or member of any such Affiliate;

(1) declared, set aside, paid or made any distribution or payment in respect of the limited liability company membership interests of Kinergy, or any direct or indirect redemption, purchase or other acquisition of any of its limited liability company membership interests;

(m) knowingly waived or released any right or claim of Kinergy;

(n) received a commencement notice or, to the knowledge of Kinergy, received any threat of commencement, of any civil or criminal litigation, investigation or proceeding against Kinergy;

(o) experienced any labor trouble or, to the knowledge of Kinergy, received any claim of wrongful discharge or worker's compensation claim;

(p) agreed, whether in writing or otherwise, to take any action referred to in and prohibited by this Section 6.12; or

(q) become aware of any other event or condition that has had or would reasonably be expected to have a material adverse effect on the condition (financial or otherwise), results of operations, assets, liabilities, reserves, the Kinergy Business, the Kinergy Intellectual Property (as defined below) or the operations of Kinergy.

6.13 Title to Properties; Encumbrances.

(a) Kinergy has good and marketable title to all of its properties and assets (real, personal or mixed, tangible or intangible), including without limitation the Kinergy Intellectual Property. None of Kinergy's properties or

assets is subject to any Lien, except for Permitted Liens (including the Liens set forth on Schedule 6.13), none of which adversely affects the Kinergy Business or the continued operations of Kinergy.

26

(b) All material property and assets (real, personal or mixed, tangible or intangible) used or required by Kinergy in the conduct of the Kinergy Business are fully owned by Kinergy (except to the extent of any Permitted Liens). All such property and assets, or the leases or licenses thereof, constitute all property, assets and contractual rights necessary for the conduct of the Kinergy Business as presently conducted.

6.14 Leases.

(a) Schedule 6.14 contains a true and complete list of:

(i) all leases pursuant to which Kinergy leases or subleases any real property interests, whether as lessor, lessee, sublessor or sublessee;

(ii) all leases pursuant to which Kinergy leases any type of personal property;

(iii) all leases pursuant to which Kinergy leases any vehicles or related equipment; and

(iv) all leases pursuant to which Kinergy leases to others any type of property.

(b) Each such lease described on Schedule 6.14 is the legal, valid and binding obligation of Kinergy and, to the knowledge of Kinergy, the other parties thereto, enforceable in accordance with their respective terms, and is in full force and effect. Kinergy is not in default under any such lease, and Kinergy has not received any notice from any person or entity asserting that Kinergy is in default under any such lease, and no events or circumstances exist which, with notice or the passage of time or both, would constitute a default under any such lease.

6.15 Intellectual Property. Kinergy owns all right, title and interest in, or has the right to use, sell or license all patent applications, patents, trademark applications, trademarks, service marks, trade names, copyright applications, copyrights, trade secrets, know-how, technology, customer lists, proprietary processes and formulae, all source and object code, algorithms, inventions, development tools and all documentation and media constituting, describing or relating to the above, including, without limitation, manuals, memoranda and records and other intellectual property and proprietary rights used in or reasonably necessary or required for the conduct of its respective business as presently conducted (collectively, the "Kinergy Intellectual Property").

6.16 Compliance with Laws. Kinergy has not been charged with, and, to Kinergy's knowledge, Kinergy is not threatened with or under any investigation with respect to, any charge concerning any violation of any provision of any federal, state, local or foreign law, regulation, ordinance, order or administrative ruling affecting the Kinergy Business or Kinergy, and Kinergy is not in default with respect to any order, writ, injunction or decree of any court, agency or instrumentality affecting the Kinergy Business or Kinergy. To Kinergy's knowledge, Kinergy is not in violation of any federal, state, local or foreign law, ordinance or regulation or any other requirement of any Governmental Body or regulatory body, court or arbitrator applicable to the Kinergy Business or Kinergy which would have a material adverse effect on Kinergy or the

27

Kinergy Business. Without limiting the generality of the foregoing, Kinergy is in compliance in all material respects with all Occupational Safety and Health Laws, including those rules and regulations promulgated by OSHA, except where such non-compliance would not have a material adverse effect on Kinergy or the Kinergy Business.

6.17 Employee Benefit Plans.

(a) Schedule 6.17 contains a true and complete list and description of each pension, retirement, severance, welfare, profit-sharing, stock purchase, stock option, vacation, deferred compensation, bonus or other incentive plan, or other employee benefit program, arrangement, agreement or understanding, or medical, vision, dental or other health plan, or life insurance or disability plan, retiree medical or life insurance plan or any other employee benefit plans, including, without limitation, any "employee benefit plan" (as defined in Section 3(3) of ERISA), to which Kinergy contributes or is a party or by which it is bound or under which it may have liability and under which employees or former employees of Kinergy (or their beneficiaries) are eligible to participate or derive a benefit. Each employee benefit plan which is a "group health plan" (as such term is defined in Section 5000(b)(i) of the Code) satisfies the applicable requirements of Section 4980B of the Code. Except as described on Schedule 6.17, Kinergy has no formal plan or commitment, whether legally binding or not, to create any additional plan, practice or agreement or modify or change any existing plan, practice or agreement that would affect any of its employees or terminated employees. Benefits under all employee benefit plans are as represented and have not been and will not be increased subsequent to the date copies of such plans have been provided.

(b) Kinergy does not contribute to or have any obligation to contribute to, has not at any time contributed to or had an obligation to contribute to, sponsor or maintain, and has not at any time sponsored or maintained, a "multi-employer plan" (within the meaning of Section 3(37) of ERISA) for the benefit of employees or former employees of Kinergy.

(c) Kinergy has, in all material respects, performed all obligations, whether arising by operation of law, contract, or past custom, required to be performed under or in connection with the employee benefit plans disclosed on Schedule 6.17 (individually, a "Kinergy Employee Benefit Plan" and, collectively, the "Kinergy Employees Benefit Plans"), and Kinergy has no knowledge of any default or violation by any other party with respect thereto.

(d) There are no Actions, suits or claims (other than routine claims for benefits) pending, or, to Kinergy's knowledge, threatened, against any Kinergy Employee Benefit Plan or against the assets funding any Kinergy Employee Benefit Plan.

(e) Kinergy neither maintains nor contributes to any "employee welfare benefit" (as such term is defined in Section 3(i) of ERISA) plan which provides any benefits to retirees or former employees of Kinergy.

6.18 Employment Law Matters.

(a) Kinergy (i) is in material compliance with all applicable laws respecting employment, employment practices, terms and conditions of employment and wages and hours; (ii) is in material compliance with all applicable laws and regulations relating to the employment

28

of aliens or similar immigration matters; and (iii) is not engaged in any unfair labor practice, including, but not limited to, discrimination or wrongful discharge.

(b) Kinergy has not at anytime during the last three (3) years had, nor to Kinergy's knowledge, is there now threatened, a strike, picket, work stoppage, work slowdown or other labor trouble, against or directly affecting Kinergy that had or would reasonably be expected to have a material adverse effect on the Kinergy Business or Kinergy.

(c) None of the employees of Kinergy is represented by a labor union, and no petition has been filed or proceedings instituted by any employee or group of employees with any labor relations board seeking recognition of a bargaining representative. Kinergy is not a party to any multi-employer collective bargaining agreement covering any of its employees.

(d) There are no controversies or disputes (including any union grie vances or arbitration proceeding) pending, or, to Kinergy's knowledge, threatened, between Kinergy and any employees of Kinergy (or any union or other representative of such employees). No unfair labor practice complaints have been filed against Kinergy with the National Labor Relations Board or any other Governmental Body or administrative body, and Kinergy has not received any written notice or communication reflecting an intention or a threat to file any such complaint.

6.19 Contracts and Commitments.

(a) Together with the leases set forth on Schedule 6.14, the insurance policies set forth on Schedule 6.23, and the Kinergy Employee Benefit Plans and commitments set forth on Schedule 6.17, Schedule 6.19 contains a true and complete list and description (stated without duplication), of:

(i) all contracts (including, without limitation, letters of credit, and obligations for borrowed money) and commitments of Kinergy which are material to the operations, business, prospects or condition (financial or otherwise) of Kinergy;

(ii) all consulting agreements (whether written or oral), regardless of amounts or duration;

(iii) all material contracts or commitments (whether written or oral) with distributors, brokers, manufacturer's representatives, sales representatives, service or warranty representatives, customers and other persons, firms, corporations or other entities engaged in the sale, distribution, service or repair of Kinergy's products;

(iv) all contracts relating to construction-in-progress of capital assets; and

(v) all joint venture, licensing, profit sharing, royalty or similar agreements or arrangements to which Kinergy is a party in any way associated with the manufacture, marketing, sale or distribution of any products or provision of any services of Kinergy.

29

(b) Kinergy has delivered to Accessity true and complete copies of all of the documents identified on Schedule 6.19 (collectively, the "Kinergy Material Contracts") and shall deliver true and complete copies of all such other agreements, instruments and documents as Accessity may reasonably request relating to the operation, ownership or conduct of the Kinergy Business.

(c) Kinergy is not a party to any written agreement that would restrict it from carrying on the Kinergy Business anywhere in the world.

(d) Kinergy is not a party to any "take-or-pay" contracts.

(e) Except as identified on Schedule 6.19, Kinergy is not a party to any employment agreements, arrangements and commitments, including severance or termination arrangements and commitments (whether written or oral), between Kinergy and any employees of Kinergy.

(f) Kinergy is not, and to the knowledge of Kinergy, no other party is, in default under or in breach or violation of, nor has Kinergy received notice of any asserted claim of default by Kinergy or by any other party under, or a breach or violation of, any of the Kinergy Material Contracts.

6.20 No Brokers. Except as disclosed on Schedule 6.20, Kinergy is not obligated for the payment of fees or expenses of any investment banker, broker or finder in connection with the origin, negotiation or execution of this Agreement in connection with any exchange of stock transaction provided for herein.

6.21 Environmental Matters. Kinergy is in compliance in all material respects with all Environmental Laws. There is no Action pending before any court, Governmental Body or board or other forum or threatened by any person or entity (i) for noncompliance by Kinergy with any Environmental Law (ii) relating to the release into the environment by Kinergy of any pollutant, toxic or hazardous material or waste generated by Kinergy. Whether or not occurring at or on a site owned, leased or operated by Kinergy. There has not been by Kinergy, nor to the knowledge of Kinergy has there been at all, any past, storage, disposal, generation, manufacture, refinement, transportation, production or treatment of any hazardous materials or substances at, upon or from the facilities occupied or used by Kinergy and any other real property presently or formerly owned by, used by or leased to or by Kinergy, any predecessor of

Kinergy (collectively, the "Kinergy Property"). To the knowledge of Kinergy, neither Kinergy nor any properties owned or operated by Kinergy has been or is in violation or is otherwise liable under, any Environmental Law. To the knowledge of Kinergy, there are no asbestos-containing materials, underground storage tanks or polychlorinated biphenyls (PCBs) located on the Kinergy Property. To the knowledge of Kinergy, there has been no spill, discharge, leak, emission, injection, disposal, escape, dumping or release of any kind on, beneath or above the Kinergy Property or into the environment surrounding such Kinergy Property of any hazardous materials or substances in violation of any Environmental Law or requiring any remedial action. To the knowledge of Kinergy, Kinergy has all permits, registrations, approvals and licenses required by any Governmental Body under any Environmental Law to be obtained by Kinergy in connection with the conduct of the business of Kinergy as presently conducted.

30

6.22 Bank Accounts. Schedule 6.22 sets forth the names and locations of all banks, trust companies, savings and loan associations, and other financial institutions at which Kinergy maintains accounts of any nature and the names of all persons authorized to draw thereon or make withdrawals therefrom.

6.23 Insurance. Schedule 6.23 sets forth a true and complete list and description of (a) all of Kinergy's self-insurance practices and items covered by such self-insurance and (b) all policies of fire, liability, worker's compensation and other forms of insurance owned or held by Kinergy. No installment premiums are due under the policies set forth on Schedule 6.23 or, if installment premiums shall be due and owing under such policies prior to the Closing Date, such premiums shall have been paid up-to-date prior to the Closing Date. All such policies are in full force and effect, insure against risks and liabilities to the extent and in the manner deemed appropriate and sufficient by Kinergy in its reasonable business judgment, and Kinergy has not received any notice of cancellation with respect thereto. To Kinergy's knowledge, Kinergy is not in default with respect to any provision contained in any such policy and has not failed to give any notice or present any claim under any such policy in a timely fashion.

6.24 Suppliers and Customers. Kinergy does not have any knowledge that any supplier or customer or group of related suppliers or customers of Kinergy has canceled or otherwise terminated or threatened to cancel or otherwise terminate, its relationship with Kinergy, which termination would have a material adverse effect on the Kinergy Business or Kinergy, or that any such supplier or customer or group of related suppliers or customers expects to reduce its business with Kinergy by reason of the transactions contemplated by this Agreement or for any other reason whatsoever.

6.25 Licenses, Permits and Authorizations. Kinergy has all necessary Permits for the use and ownership or leasing of its properties and assets as currently operated, used, owned or leased, except for such Permits as to which the lack thereof does not and would not have a material adverse effect on Kinergy or the Kinergy Business. All of the Permits are valid, in full force and effect and in good standing. Schedule 6.25 contains a true and complete list and description of all the Permits and Kinergy has previously delivered to Accessity true and complete copies of all such Permits identified in Schedule 6.25 and in effect as of the date hereof. There is no claim or Action pending, or, to Kinergy's knowledge, threatened, which disputes the validity of any such Permit or threatens to revoke, cancel, suspend or limit any such Permit.

6.26 Accounts Receivable. All accounts receivable of Kinergy shown on the Kinergy Financial Statements and all accounts receivable created after March 31, 2004, subject to reserves created in the ordinary course of business on a basis consistent with the past practices and policies of Kinergy and otherwise in accordance with generally accepted accounting principles, (a) have been collected or (b) to Kinergy's knowledge, are valid and enforceable, arose from bona-fide sales to third parties in the ordinary course of business, and are collectible at the aggregate recorded amounts thereof on the books of Kinergy.

6.27 Condition of Tangible Assets. Kinergy's facilities and tangible assets, including, without limitation, machinery, equipment, vehicles, furniture, plants and buildings, are in good operating condition and repair (ordinary wear and tear excepted) and are adequate for the uses to which they have been put by Kinergy in the ordinary course of business, except for parts or repairs of an immaterial nature in the aggregate, and Kinergy has not received any notice that any of such facilities or assets is in need of substantial maintenance or repair.

6.28 Disclosure. No representation or warranty of Kinergy in this Agreement (including, without limitation, the Schedules of Kinergy hereto) contains or will contain any untrue statement of a material fact or omits or will omit to state any material fact necessary to make the statements herein or therein not misleading.

ARTICLE VII REPRESENTATIONS AND WARRANTIES OF THE KINERGY MEMBERS

Each of the Kinergy Members, severally and not jointly, hereby represents and warrants to Accessity that the following representations and warranties are true and correct on and as of the date hereof:

7.1 Power and Authority. Such Kinergy Member has full power and authority to enter into this Agreement, perform its respective obligations hereunder, and carry out the transactions contemplated hereby. This Agreement constitutes the legal, valid and binding obligation of such Kinergy Member, enforceable against such Kinergy Member in accordance with its terms.

7.2 No Violation. Neither the execution and delivery of this Agreement nor the performance by such Kinergy Member of its respective obligations hereunder nor the consummation of the transactions contemplated hereby will violate, or be in conflict with, or constitute a default under, any mortgage, indenture, lease, or any agreement, instrument or commitment to which such Kinergy Member is a party.

7.3 Title; Consents and Approvals; No Claims. Such Kinergy Member is the owner, beneficially and of record, of its Kinergy Interests, free and clear of all liens, encumbrances, security agreements, options, claims, charges and restrictions. To such Kinergy Member's knowledge, no consent, approval or authorization of, or declaration, filing or registration with, any governmental or regulatory authority or any other third party is required to be made or obtained by such Kinergy Member in connection with the execution, delivery or performance of this Agreement. To such Kinergy Member's knowledge, there is no claim, action, litigation, suit, cause of action or other proceeding pending or threatened before any federal, state or local court, governmental agency or regulatory body against such Kinergy Member which seeks or may seek, directly or indirectly, (a) to invalidate or set aside, in whole or in part, this Agreement, (b) to restrain, prohibit, invalidate or set aside, in whole or in part, the consummation of the transactions contemplated hereby or (c) to obtain substantial damages in connection therewith.

7.4 Securities Law Compliance.

(a) Such Kinergy Member has a net worth sufficient to bear the economic risk (including the entire loss) of its investment made in the Accessity Exchange Shares;

(b) Such Kinergy Member is an "accredited investor," as such term is defined in Rule 501 of Regulation D promulgated under the rules and regulations of the Securities Act;

32

(c) Such Kinergy Member has adequate means of providing for its current cash needs and personal contingencies and has no need for liquidity in this investment in the Accessity Exchange Shares and has no reason to anticipate any change in its personal circumstances, financial or otherwise, which may cause or require any sale or distribution by such Kinergy Member or all or any part of the Accessity Exchange Shares acquired by it herein;

(d) by reason of such Kinergy Member's business or financial experience or the business or financial experience of such Kinergy Member's professional advisor(s) who are unaffiliated with and who are not compensated by Accessity or any affiliate or selling agent of Accessity, directly or indirectly, such Kinergy Member has the capacity to protect its own interests in connection with an investment in the Accessity Exchange Shares; (e) Such Kinergy Member understands that he, she or it is acquiring Accessity Exchange Shares without being furnished any prospectus or offering circular, other than a copy of this Agreement, a copy of the Proxy Statement (as defined in Section 11.6 below) and a copy of the Owner Disclosure Document (as defined in Section 11.7 below);

(f) No representations or warranties have been made to such Kinergy Member by Accessity or any employee or agent of Accessity and in entering into this Agreement, such Kinergy Member is not relying on any information, other than as a result of the independent investigation of Accessity by such Kinergy Member, and no guarantee of any profit or return on its investment made in the Accessity Exchange Shares has been made to such Kinergy Member;

(g) In evaluating the merits and risk of this investment, such Kinergy Member has relied on the advice of its personal tax advisor, investment advisor and/or legal counsel;

(h) Such Kinergy Member is aware that the Accessity Exchange Shares have not been registered or qualified, nor is registration or qualification contemplated, with the SEC under the Securities Act or any state securities law. Accordingly, the Accessity Exchange Shares may be sold or otherwise transferred or hypothecated only if they are subsequently registered or qualified under the Securities Act or applicable laws or if, in the opinion of counsel, an exemption from registration or qualification thereunder is available and the transaction will not jeopardize the availability of the exemptions under applicable federal and state securities laws relied upon by Accessity in connection with the offering in which such Kinergy Member acquired its Accessity Exchange Shares;

(i) Such Kinergy Member acknowledges that the Accessity Exchange Shares were not offered by means of any general solicitation or advertising;

(j) Such Kinergy Member is acquiring its Accessity Exchange Shares solely for its own account, for investment purposes only, and not with an intent to sell, or for resale in connection with any distribution of all or any portion of the Accessity Exchange Shares within the meaning of the Securities Act; and

(k) The address of such Kinergy Member furnished by such Kinergy Member at the end of this Agreement is the principal residence of such Kinergy Member, if such Kinergy Member is an individual, or the principal business address of such Kinergy Member, if such Kinergy Member is a business or other entity, and that all offers to such Kinergy Member have been made only in the state specified in such address.

33

ARTICLE VIII REPRESENTATIONS AND WARRANTIES OF REENERGY

Reenergy hereby represents and warrants to Accessity that the following representations and warranties are true and correct on and as of the date hereof:

8.1 Organization and Good Standing. Reenergy is a limited liability company duly organized, validly existing and in good standing under the laws of the State of California, has the power and authority to own, operate and lease its properties and to carry on its business as now conducted, and is duly qualified to do business and is in good standing in each jurisdiction where the character of its properties owned or leased or the nature of its activities make such qualification necessary (each such jurisdiction being listed on Schedule 8.1), except where the failure to be so qualified would not have a material adverse effect on Reenergy.

8.2 Capitalization. The limited liability company membership interests of Reenergy are owned by the Reenergy Members as set forth in Exhibit D. No limited liability company membership interests of Reenergy are outstanding as of the date of this Agreement other than the Reenergy Interests set forth on Exhibit D. The Reenergy Interests have been duly authorized and validly issued, are fully paid and nonassessable, are not subject to any right of rescission and have been offered, issued, sold and delivered by Reenergy in compliance with all requirements of applicable laws. 8.3 Power and Authority. Reenergy has full power and authority to enter into this Agreement, to perform its obligations hereunder and to carry out the transactions contemplated hereby. The execution and delivery of this Agreement, the performance by Reenergy of its obligations hereunder and the consummation of the transactions contemplated hereby have been duly authorized by Reenergy and the Reenergy Majority Members as required by applicable law, Reenergy's Articles of Organization or Operating Agreement. This Agreement constitutes the legal, valid and binding obligation of Reenergy, enforceable against it in accordance with its terms.

8.4 Options/Rights. Except as set forth in Schedule 8.4, there are no (i) options, warrants, calls, rights, commitments, conversion privileges or preemptive or other rights or agreements outstanding to purchase or otherwise acquire any limited liability company membership interests of Reenergy, (ii) securities or debt convertible into or exchangeable for Reenergy limited liability company membership interests or obligating Reenergy to grant, extend or enter into any such option, warrant, call, commitment, conversion privileges or preemptive or other right or agreement, or (iii) voting agreements, rights of first refusal, preemptive rights, co-sale rights, or other restrictions applicable to any outstanding limited liability company membership interests of Reenergy.

8.5 Subsidiaries. Reenergy does not have any subsidiaries or any equity interest, direct or indirect, in, or loans to, any corporation, partnership, joint venture, limited liability company or other business entity.

8.6 No Violation. Neither the execution and delivery of this Agreement nor the performance by Reenergy of its obligations hereunder nor the consummation of the transactions contemplated hereby will (a) contravene any provision of the Articles of Organization or Operating Agreement of Reenergy; (b) violate, be in conflict with, constitute a default under,

34

permit the termination of, cause the acceleration of the maturity of any debt or obligation of Reenergy under, require (except as disclosed on Schedule 8.6 hereto) the consent of any other party to, constitute a breach of, create a loss of a material benefit under, or result in the creation or imposition of any Lien, upon any property or assets of Reenergy under, any mortgage, indenture, lease, contract, agreement, instrument or commitment to which Reenergy is a party or by which it or he or any of its or his respective assets or properties may be bound; (c) to the knowledge of Reenergy, violate any statute or law or any judgment, decree, order, regulation or rule of any court or Governmental Body to which Reenergy or the Reenergy Business is subject or by which Reenergy, any of its assets or properties are bound; or (d) result in the loss of any license, privilege or certificate benefiting Reenergy or the Reenergy Business.

8.7 Consents and Approvals. Except as set forth on Schedule 8.7 hereto, no consent, approval or authorization of, or declaration, filing or registration with, any governmental or regulatory authority or any other third party is required to be made or obtained by Reenergy in connection with the execution, delivery or performance of this Agreement by Reenergy.

8.8 Litigation. There are Actions to which Reenergy is a party, including, without limitation, Actions for personal injury, products liability, wrongful death or other tortious conduct, or breach of warranty arising from or relating to materials, commodities, products or goods used, transferred, processed, manufactured, sold, distributed or shipped by Reenergy (a) involving or relating to Reenergy, or any its assets, properties or rights, or (b) pending, or, to Reenergy's knowledge, threatened, against Reenergy or any of its assets, properties or rights, before any court, arbitrator or administrative or Governmental Body which, if adversely resolved, would have a material adverse effect on the Reenergy Business.

8.9 Reenergy Financial Statements. Reenergy has previously delivered to Accessity the unaudited balance sheet of Reenergy as of December 31, 2003 and the related statements of income and changes in financial position or cash flows, as appropriate, for the period then ended (the "Unaudited 2003 Reenergy Financial Statements"), together with an unaudited balance sheet of Reenergy as of March 31, 2004 and the related statements of income and changes in financial position or cash flows, as appropriate, for the period then ended and, subsequent to the date hereof, Reenergy will deliver to Accessity the audited balance sheet of Reenergy as of December 31, 2003 and the related statements of income and changes in financial position or cash flows, as appropriate, for the period then ended (the "Audited 2003 Reenergy Financial Statements") (all such financial statements, excluding, upon delivery to Accessity of the Audited 2003 Reenergy Financial Statements, the Unaudited 2003 Reenergy Financial Statements, are hereinafter collectively referred to as the "Reenergy Financial Statements"). The Reenergy Financial Statements, together with the notes thereto, (i) were compiled from the books and records of Reenergy regularly maintained by management and used to prepare the financial statements of Reenergy, (ii) were prepared in accordance with GAAP consistently applied throughout the period then ended and all periods prior to that period; and (iii) present fairly and accurately the financial condition of Reenergy for the period or as of the dates thereof, subject, where appropriate, to normal year-end audit adjustments, in each case in accordance with GAAP consistently applied during the period covered.

8.10 No Undisclosed Liabilities. Reenergy has, and on the Closing Date will have, no Liabilities other than those which (a) are fully reflected reserved against in the Reenergy Financial Statements, (b) have been incurred since March 31, 2004 in the ordinary course of

35

business in amounts and for terms consistent, individually and in the aggregate, with the past practice of Reenergy or (c) have been specifically disclosed in the Schedules hereto by reference to the specific section of this Agreement to which such disclosure relates.

8.11 Taxes and Tax Returns. All of the tax returns and reports of Reenergy required by applicable law to be filed prior to the date hereof have been duly filed and all taxes shown as due thereon have been paid. There are in effect no waivers of the applicable statutes of limitations for any federal, state, local or foreign taxes for any period. No liability for any federal, state, local or foreign income, sales, use, withholding, payroll, franchise, real property or personal property taxes is pending, and there is no proposed liability for any such taxes to be imposed upon the properties or assets of Reenergy. Reenergy does not have any liability for any federal, state, local or foreign income, sales, use, withholding, payroll, franchise, real property or personal property taxes, assessments, amounts, interest or penalties of any nature whatsoever other than as shown on the March 31, 2004 Reenergy Financial Statements and there is no basis for any additional claim or assessment other than with respect to liabilities for taxes which may have accrued since the date of the March 31, 2004 Reenergy Financial Statements in the ordinary course of business and reserved against on the books and records of Reenergy compiled in accordance with generally accepted accounting principles which have been consistently applied to the Closing Date. The provisions for taxes reflected in the March 31, 2004 Reenergy Financial Statements are adequate for federal, state, county and local taxes for the period ended on December 31, 2003 and for all prior periods, whether disputed or undisputed. There are no present disputes about taxes of any nature payable by Reenergy. Reenergy has never filed, and will not file on or before the Closing Date, any consent under section 341(f) of the Code. Reenergy's tax returns have never been audited by any taxation authority.

8.12 Absence of Certain Changes. Except as set forth on Schedule 8.12, since December 31, 2003, Reenergy has conducted the Reenergy Business only in the ordinary course and consistent with prior practices and has not:

(a) suffered any material adverse change in its condition (financial or otherwise), results of operations, assets, liabilities, reserves, the Reenergy Business, or operations;

(b) suffered any damage, destruction or loss, whether covered by insurance or not, materially adversely affecting the Reenergy Business, operations, assets, or condition (financial or otherwise);

(c) paid, discharged or satisfied any Liability or other expenses, other than the payment, discharge or satisfaction of the Liabilities described in Section 8.10 at the time the same were due and payable and in the ordinary course of business;

(d) paid or otherwise made any contribution to any profit-sharing or pension plan or other Employee Benefit Plan (as defined in Section 8.17 below);

(e) mortgaged or pledged, or permitted the imposition of any Lien upon, any of its properties or assets (real, personal or mixed, tangible or intangible), other than those incurred in the ordinary course of business or otherwise listed on Schedule 8.12 hereto;

36

(f) cancelled or compromised any debts, or waived or permitted to lapse any material claims or rights, or sold, assigned, transferred or otherwise disposed of, other than in the ordinary course, any of its properties or assets (real, personal or mixed, tangible or intangible);

(g) disposed of or permitted to lapse any rights to the use of any patent, registered trademark, service mark, trade name or copyright, or disposed of or disclosed to any person any trade secret, formula, process or know-how material to the Reenergy Business not theretofore a matter of public knowledge;

(h) except as disclosed on Schedule 8.12, granted any increase in the compensation of any officer, employee or consultant of Reenergy (including any such increase pursuant to any bonus, pension, profit-sharing or other plan or commitment) or any increase in the compensation payable or to become payable to any officer, employee or consultant;

(i) entered into any commitment or transaction not in the ordinary course of business or made any capital expenditure or commitment for any additions to property, plant or equipment, except commitments, transactions or capital expenditures which do not in any single case exceed \$25,000 or in the aggregate exceed \$50,000;

(j) made any change in any method of accounting or accounting practice (including, without limitation, any change in depreciation or amortization policies or rates);

(k) paid, loaned or advanced any amount to, or sold, transferred or leased any properties or assets (real, personal or mixed, tangible or intangible) to, or entered into any agreement or arrangement with, any of its officers, employees, members, or any family member or Affiliate of any of its officers, employees or members, or any officer, employee or member of any such Affiliate;

(1) declared, set aside, paid or made any distribution or payment in respect of the limited liability company membership interests of Reenergy, or any direct or indirect redemption, purchase or other acquisition of any of its limited liability company membership interests;

(m) knowingly waived or released any right or claim of Reenergy;

(n) received a commencement notice or, to the knowledge of Reenergy, received any threat of commencement, of any civil or criminal litigation, investigation or proceeding against Reenergy;

(o) experienced any labor trouble or, to the knowledge of Reenergy, received any claim of wrongful discharge or worker's compensation claim;

(p) agreed, whether in writing or otherwise, to take any action referred to in and prohibited by this Section 8.12; or

(q) become aware of any other event or condition that has had or would reasonably be expected to have a material adverse effect on the condition (financial or otherwise), results of operations, assets, liabilities, reserves, the Reenergy Business, the Reenergy Intellectual Property (as defined below) or the operations of Reenergy.

37

8.13 Title to Properties; Encumbrances.

(a) Reenergy has good and marketable title to all of its properties and assets (real, personal or mixed, tangible or intangible), including without limitation the Reenergy Intellectual Property. None of Reenergy's properties or assets is subject to any Lien, except Permitted Liens (including the Liens set forth on Schedule 8.13), none of which adversely affects the Reenergy Business or the continued operations of Reenergy.

(b) All material property and assets (real, personal or mixed, tangible or intangible) used or required by Reenergy in the conduct of the Reenergy Business are fully owned by Reenergy (except to the extent of any Permitted Liens). All such property and assets, or the leases or licenses thereof, constitute all property, assets and contractual rights necessary for the conduct of the Reenergy Business.

8.14 Leases.

(a) Schedule 8.14 contains a true and complete list of:

(i) all leases pursuant to which Reenergy leases or subleases any real property interests, whether as lessor, lessee, sublessor or sublessee;

(ii) all leases pursuant to which Reenergy leases any type of personal property;

(iii) all leases pursuant to which Reenergy leases any vehicles or related equipment; and

(iv) all leases pursuant to which Reenergy leases to others any type of property.

(b) Each such lease described on Schedule 8.14 is the legal, valid and binding obligation of Reenergy and, to the knowledge of Reenergy, the other parties thereto, enforceable in accordance with their respective terms, and is in full force and effect. Reenergy is not in default under any such lease, and Reenergy has not received any notice from any person or entity asserting that Reenergy is in default under any such lease, and no events or circumstances exist which, with notice or the passage of time or both, would constitute a default under any such lease.

8.15 Intellectual Property. Reenergy owns all right, title and interest in, or has the right to use, sell or license all patent applications, patents, trademark applications, trademarks, service marks, trade names, copyright applications, copyrights, trade secrets, know-how, technology, customer lists, proprietary processes and formulae, all source and object code, algorithms, inventions, development tools and all documentation and media constituting, describing or relating to the above, including, without limitation, manuals, memoranda and records and other intellectual property and proprietary rights used in or reasonably necessary or required for the conduct of its respective business as presently conducted (collectively, the "Reenergy Intellectual Property").

38

8.16 Compliance with Laws. Reenergy has not been charged with, and, to Reenergy's knowledge, Reenergy is not threatened with or under any investigation with respect to, any charge concerning any violation of any provision of any federal, state, local or foreign law, regulation, ordinance, order or administrative ruling affecting the Reenergy Business or Reenergy, and Reenergy is not in default with respect to any order, writ, injunction or decree of any court, agency or instrumentality affecting the Reenergy Business or Reenergy. To Reenergy's knowledge, Reenergy is not in violation of any federal, state, local or foreign law, ordinance or regulation or any other requirement of any Governmental Body or regulatory body, court or arbitrator applicable to the Reenergy Business or Reenergy which would have a material adverse effect on Reenergy or the Reenergy Business. Without limiting the generality of the foregoing, Reenergy is in compliance in all material respects with all Occupational Safety and Health Laws, including those rules and regulations promulgated by OSHA, except where such non-compliance would not have a material adverse effect on Reenergy or the Reenergy Business.

8.17 Employee Benefit Plans.

(a) Schedule 8.17 contains a true and complete list and description of each pension, retirement, severance, welfare, profit-sharing, stock purchase, stock option, vacation, deferred compensation, bonus or other incentive plan, or other employee benefit program, arrangement, agreement or understanding, or medical, vision, dental or other health plan, or life insurance or disability plan, retiree medical or life insurance plan or any other employee benefit plans, including, without limitation, any "employee benefit plan" (as defined in Section 3(3) of ERISA), to which Reenergy contributes or is a party or by which it is bound or under which it may have liability and under which employees or former employees of Reenergy (or their beneficiaries) are eligible to participate or derive a benefit. Each employee benefit plan which is a "group health plan" (as such term is defined in Section 5000(b)(i) of the Code) satisfies the applicable requirements of Section 4980B of the Code. Except as described on Schedule 8.17, Reenergy has no formal plan or commitment, whether legally binding or not, to create any additional plan, practice or agreement or modify or change any existing plan, practice or agreement that would affect any of its employees or terminated employees. Benefits under all employee benefit plans are as represented and have not been and will not be increased subsequent to the date copies of such plans have been provided.

(b) Reenergy does not contribute to or have any obligation to contribute to, has not at any time contributed to or had an obligation to contribute to, sponsor or maintain, and has not at any time sponsored or maintained, a "multi-employer plan" (within the meaning of Section 3(37) of ERISA) for the benefit of employees or former employees of Reenergy.

(c) Reenergy has, in all material respects, performed all obligations, whether arising by operation of law, contract, or past custom, required to be performed under or in connection with the employee benefit plans disclosed on Schedule 8.17 (individually, a "Reenergy Employee Benefit Plan" and, collectively, the "Reenergy Employees Benefit Plans"), and Reenergy has no knowledge of any default or violation by any other party with respect thereto.

(d) There are no Actions, suits or claims (other than routine claims for benefits) pending, or, to Reenergy's knowledge, threatened, against any Reenergy Employee Benefit Plan or against the assets funding any Reenergy Employee Benefit Plan.

39

(e) Reenergy neither maintains nor contributes to any "employee welfare benefit" (as such term is defined in Section 3(i) of ERISA) plan which provides any benefits to retirees or former employees of Reenergy.

8.18 Employment Law Matters.

(a) Reenergy (i) is in material compliance with all applicable laws respecting employment, employment practices, terms and conditions of employment and wages and hours; (ii) is in material compliance with all applicable laws and regulations relating to the employment of aliens or similar immigration matters; and (iii) is not engaged in any unfair labor practice, including, but not limited to, discrimination or wrongful discharge.

(b) Reenergy has not at anytime during the last three (3) years had, nor to Reenergy's knowledge, is there now threatened, a strike, picket, work stoppage, work slowdown or other labor trouble, against or directly affecting Reenergy that had or would reasonably be expected to have a material adverse effect on the Reenergy Business or Reenergy.

(c) None of the employees of Reenergy is represented by a labor union, and no petition has been filed or proceedings instituted by any employee or group of employees with any labor relations board seeking recognition of a bargaining representative. Reenergy is not a party to any multi-employer collective bargaining agreement covering any of its employees.

(d) There are no controversies or disputes (including any union grievances or arbitration proceeding) pending, or, to Reenergy's knowledge, threatened, between Reenergy and any employees of Reenergy (or any union or other representative of such employees). No unfair labor practice complaints have been filed against Reenergy with the National Labor Relations Board or any other Governmental Body or administrative body, and Reenergy has not received any written notice or communication reflecting an intention or a threat to file any such complaint.

8.19 Contracts and Commitments.

(a) Together with the leases set forth on Schedule 8.14, the insurance policies set forth on Schedule 8.23, and the Reenergy Employee Benefit Plans and commitments set forth on Schedule 8.17, Schedule 8.19 contains a true and complete list and description (stated without duplication), of:

(i) all contracts (including, without limitation, letters of credit, and obligations for borrowed money) and commitments of Reenergy which are material to the operations, business, prospects or condition (financial or otherwise) of Reenergy;

(ii) all consulting agreements (whether written or oral), regardless of amounts or duration;

(iii) all material contracts or commitments (whether written or oral) with distributors, brokers, manufacturer's representatives, sales representatives, service or warranty representatives, customers and other persons, firms, corporations or other entities engaged in the sale, distribution, service or repair of Reenergy's products;

40

(iv) all contracts relating to construction-in-progress of capital assets; and

(v) all joint venture, licensing, profit sharing, royalty or similar agreements or arrangements to which Reenergy is a party in any way associated with the manufacture, marketing, sale or distribution of any products or provision of any services of Reenergy.

(b) Reenergy has delivered to Accessity true and complete copies of all of the documents identified on Schedule 8.19 (collectively, the "Reenergy Material Contracts") and shall deliver true and complete copies of all such other agreements, instruments and documents as Accessity may reasonably request relating to the operation, ownership or conduct of the Reenergy Business.

(c) Reenergy is not a party to any written agreement that would restrict it from carrying on the Reenergy Business anywhere in the world.

(d) Reenergy is not a party to any "take-or-pay" contracts.

(e) Except as identified on Schedule 8.19, Reenergy is not a party to any employment agreements, arrangements and commitments, including severance or termination arrangements and commitments (whether written or oral), between Reenergy and any employees of Reenergy.

(f) Reenergy is not, and to the knowledge of Reenergy, no other party is, in default under or in breach or violation of, nor has Reenergy received notice of any asserted claim of default by Reenergy or by any other party under, or a breach or violation of, any of the Reenergy Material Contracts.

8.20 No Brokers. Except as disclosed on Schedule 8.20, Reenergy is not obligated for the payment of fees or expenses of any investment banker, broker or finder in connection with the origin, negotiation or execution of this Agreement in connection with any exchange of stock transaction provided for herein.

8.21 Environmental Matters. Reenergy is in compliance in all material respects with all Environmental Laws. There is no Action pending before any court, Governmental Body or board or other forum or threatened by any person or entity (i) for noncompliance by Reenergy with any Environmental Law (ii) relating to the release into the environment by Reenergy of any pollutant, toxic or hazardous material or waste generated by Reenergy, whether or not occurring at or on a site owned, leased or operated by Reenergy. There has not been by Reenergy, nor to the knowledge of Reenergy has there been at all, any past, storage, disposal, generation, manufacture, refinement, transportation, production or treatment of any hazardous materials or substances at, upon or from the facilities occupied or used by Reenergy and any other real property presently or formerly owned by, used by or leased to or by Reenergy, any predecessor of Reenergy (collectively, the "Reenergy Property"). To the knowledge of Reenergy, neither Reenergy nor any properties owned or operated by Reenergy has been or is in violation or is otherwise liable under, any Environmental Law. To the knowledge of Reenergy, there are no asbestos-containing materials, underground storage tanks or polychlorinated biphenyls (PCBs) located on the Reenergy Property. To the knowledge of Reenergy, there has been no spill,

discharge, leak, emission, injection, disposal, escape, dumping or release of any kind on, beneath or above the Reenergy Property or into the environment surrounding such Reenergy Property of any hazardous materials or substances in violation of any Environmental Law or requiring any remedial action. To the knowledge of Reenergy, Reenergy has all permits, registrations, approvals and licenses required by any Governmental Body under any Environmental Law to be obtained by Reenergy in connection with the conduct of the business of Reenergy as presently conducted. 8.22 Bank Accounts. Schedule

8.22 sets forth the names and locations of all banks, trust companies, savings and loan associations, and other financial institutions at which Reenergy maintains accounts of any nature and the names of all persons authorized to draw thereon or make withdrawals therefrom.

8.23 Insurance. Schedule 8.23 sets forth

a true and complete list and description of (a) all of Reenergy's self-insurance practices and items covered by such self-insurance and (b) all policies of fire,

liability, worker's compensation and other forms of insurance owned or held by Reenergy. No installment premiums are due under the policies set forth on Schedule 8.23 or, if installment premiums shall be due and owing under such policies prior to the Closing Date, such premiums shall have been paid up-to-date prior to the Closing Date. All such policies are in full force and effect, insure against risks and liabilities to the extent and in the manner deemed appropriate and sufficient by Reenergy in its reasonable business judgment, and Reenergy has not received any notice of cancellation with respect thereto. To Reenergy's knowledge, Reenergy is not in default with respect to any provision contained in any such policy and has not failed to give any notice or present any claim under any such policy in a timely fashion.

8.24 Suppliers and Customers. Reenergy does not have any knowledge that any supplier or customer or group of related suppliers or customers of Reenergy has canceled or otherwise terminated or threatened to cancel or otherwise terminate, its relationship with Reenergy, which termination would have a material adverse effect on the Reenergy Business or Reenergy, or that any such supplier or customer or group of related suppliers or customers expects to reduce its business with Reenergy by reason of the transactions contemplated by this Agreement or for any other reason whatsoever.

8.25 Licenses, Permits and Authorizations. Reenergy has all necessary Permits for the use and ownership or leasing of its properties and assets as currently operated, used, owned or leased, except for such Permits as to which the lack thereof does not and would not have a material adverse effect on Reenergy or the Reenergy Business. All of the Permits are valid, in full force and effect and in good standing. Schedule 8.25 contains a true and complete list and description of all the Permits. There is no claim or Action pending, or, to Reenergy's knowledge, threatened, which disputes the validity of any such Permit or threatens to revoke, cancel, suspend or limit any such Permit.

8.26 Accounts Receivable. All accounts receivable of Reenergy shown on the Reenergy Financial Statements and all accounts receivable created after March 31, 2004, subject to reserves created in the ordinary course of business on a basis consistent with the past practices and policies of Reenergy and otherwise in accordance with generally accepted accounting principles, (a) have been collected or (b) to Reenergy's knowledge, are valid and enforceable, arose from bona-fide sales to third parties in the ordinary course of business, and are collectible at the aggregate recorded amounts thereof on the books of Reenergy.

42

8.27 Condition of Tangible Assets. Reenergy's facilities and tangible assets, including, without limitation, machinery, equipment, vehicles, furniture, plants and buildings, are in good operating condition and repair (ordinary wear and tear excepted) and are adequate for the uses to which they have been put by Reenergy in the ordinary course of business, except for parts or repairs of an immaterial nature in the aggregate, and Reenergy has not received any notice that any of such facilities or assets is in need of substantial maintenance or repair.

8.28 Disclosure. No representation or warranty of Reenergy in this Agreement (including, without limitation, the Schedules of Reenergy hereto)

contains or will contain any untrue statement of a material fact or omits or will omit to state any material fact necessary to make the statements herein or therein not misleading.

ARTICLE IX REPRESENTATIONS AND WARRANTIES OF THE REENERGY MEMBERS

Each of the Reenergy Members, severally and not jointly, hereby represents and warrants to Accessity that the following representations and warranties are true and correct on and as of the date hereof:

9.1 Power and Authority. Such Reenergy Member has full power and authority to enter into this Agreement, perform its respective obligations hereunder, and carry out the transactions contemplated hereby. This Agreement constitutes the legal, valid and binding obligation of such Reenergy Member, enforceable against such Reenergy Member in accordance with its terms.

9.2 No Violation. Neither the execution and delivery of this Agreement nor the performance by such Reenergy Member of its respective obligations hereunder nor the consummation of the transactions contemplated hereby will violate, or be in conflict with, or constitute a default under, any mortgage, indenture, lease, or any agreement, instrument or commitment to which such Reenergy Member is a party.

9.3 Title; Consents and Approvals; No Claims. Such Reenergy Member is the owner, beneficially and of record, of its Reenergy Interests, free and clear of all liens, encumbrances, security agreements, options, claims, charges and restrictions. To such Reenergy Member's knowledge, no consent, approval or authorization of, or declaration, filing or registration with, any governmental or regulatory authority or any other third party is required to be made or obtained by such Reenergy Member in connection with the execution, delivery or performance of this Agreement. To such Reenergy Member's knowledge, there is no claim, action, litigation, suit, cause of action or other proceeding pending or threatened before any federal, state or local court, governmental agency or regulatory body against such Reenergy Member which seeks or may seek, directly or indirectly, (a) to invalidate or set aside, in whole or in part, this Agreement, (b) to restrain, prohibit, invalidate or set aside, in whole or in part, the consummation of the transactions contemplated hereby or (c) to obtain substantial damages in connection therewith.

9.4 Securities Law Compliance.

(a) Such Reenergy Member has a net worth sufficient to bear the economic risk (including the entire loss) of its investment made in the Accessity Exchange Shares;

(b) Such Reenergy Member is an "accredited investor," as such term is defined in Rule 501 of Regulation D promulgated under the rules and regulations of the Securities Act (except in regard to Celilo Group, a member of Kinergy Resources, LLC, and Tom Koehler, a Reenergy Member, each of whom shall be provided with an Owner Disclosure Document (as defined in Section 11.7 below) that is in compliance with paragraph (b)(2) of Rule 502 of Regulation D promulgated under the Securities Act);

(c) Such Reenergy Member has adequate means of providing for its current cash needs and personal contingencies and has no need for liquidity in this investment in the Accessity Exchange Shares and has no reason to anticipate any change in its personal circumstances, financial or otherwise, which may cause or require any sale or distribution by such Reenergy Member or all or any part of the Accessity Exchange Shares acquired by it herein;

(d) by reason of such Reenergy Member's business or financial experience or the business or financial experience of such Reenergy Member's professional advisor(s) who are unaffiliated with and who are not compensated by Accessity or any affiliate or selling agent of Accessity, directly or indirectly, such Reenergy Member has the capacity to protect its own interests in connection with an investment in the Accessity Exchange Shares;

(e) Such Reenergy Member understands that he, she or it is acquiring Accessity Exchange Shares without being furnished any prospectus or offering circular, other than a copy of this Agreement, a copy of the Proxy Statement (as defined in Section 11.6 below) and a copy of the Owner Disclosure Document (as

defined in Section 11.7 below);

(f) No representations or warranties have been made to such Reenergy Member by Accessity or any employee or agent of Accessity and in entering into this Agreement, such Reenergy Member is not relying on any information, other than as a result of the independent investigation of Accessity by such Reenergy Member, and no guarantee of any profit or return on its investment made in the Shares has been made to such Reenergy Member;

(g) In evaluating the merits and risk of this investment, such Reenergy Member has relied on the advice of its personal tax advisor, investment advisor and/or legal counsel;

(h) Such Reenergy Member is aware that the Accessity Exchange Shares have not been registered or qualified, nor is registration or qualification contemplated, with the SEC under the Securities Act or any state securities law. Accordingly, the Accessity Exchange Shares may be sold or otherwise transferred or hypothecated only if they are subsequently registered or qualified under the Securities Act or applicable laws or if, in the opinion of counsel, an exemption from registration or qualification thereunder is available and the transaction will not jeopardize the availability of the exemptions under applicable federal and state securities laws relied upon by Accessity in connection with the offering in which such Reenergy Member acquired its Accessity Exchange Shares;

(i) Such Reenergy Member acknowledges that the Accessity Exchange Shares were not offered by means of any general solicitation or advertising;

44

(j) Such Reenergy Member is acquiring its Accessity Exchange Shares solely for its own account, for investment purposes only, and not with an intent to sell, or for resale in connection with any distribution of all or any portion of the Accessity Exchange Shares within the meaning of the Securities Act; and

(k) The address of such Reenergy Member furnished by such Reenergy Member at the end of this Agreement is the principal residence of such Reenergy Member, if such Reenergy Member is an individual, or the principal business address of such Reenergy Member, if such Reenergy Member is a business or other entity, and that all offers to such Reenergy Member have been made only in the state specified in such address.

> ARTICLE X REPRESENTATIONS AND WARRANTIES OF ACCESSITY

Accessity hereby represents and warrants that, except as disclosed herein or set forth on its Schedules attached hereto:

10.1 Due Organization, Authorization and Validity. Accessity is a corporation duly organized, validly existing and in good standing under the laws of the state of its incorporation and has all necessary power, legal capacity and authority to (i) conduct its business in the manner in which its business is currently being conducted and to own and use its assets in the manner in which its assets are currently owned and used and (ii) enter into and perform its obligations under this Agreement and under all contracts, agreements and instruments to which it is a party or by which it or its assets or properties are subject or bound. This Agreement and the consummation of the transactions contemplated hereby has been duly and validly approved by the Board of Directors of Accessity. This Agreement constitutes the legal, valid and binding obligation of Accessity, enforceable against it in accordance with its terms.

10.2 Absence of Certain Changes. Since December 31, 2003, there has not been any change in the financial condition, properties, assets, liabilities, business or results of operations of Accessity, which change by itself or in conjunction with all other such changes, whether or not arising in the ordinary course of business, has had or can reasonably be expected to have a material adverse effect on Accessity.

10.3 No Brokers. Accessity is not obligated for the payment of fees or expenses of any investment banker, broker or finder in connection with the origin, negotiation or execution of this Agreement in connection with the exchange of stock transaction provided for herein.

10.4 SEC Filings; Financial Statements.

(a) Accessity has delivered to each of the Acquired Companies accurate and complete copies (excluding copies of exhibits) of each report, registration statements (on a form other than Form S-8) and definitive proxy statement required to be filed with the SEC by Accessity with the SEC between January 1, 2002 and the date of this Agreement (collectively, the "Accessity SEC Documents"). As of the time it was filed with the SEC (or, if amended or superseded by a filing prior to the date of this Agreement, then on the date of such filing): (i) each of the Accessity SEC Documents complied in all material respects with the applicable requirements of the Securities Act or the Exchange Act (as the case may be); and (ii) none of the Accessity SEC Documents contained any untrue statement of a material fact or omitted to state a

45

material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading. All of the Accessity SEC Documents were timely filed, unless a filing under Rule 12b-25 of the Exchange Act was timely filed, in which case the applicable filing was made within the time period prescribed in Rule 12b-25.

(b) The consolidated financial statements contained in the Accessity SEC Documents: (i) complied as to form in all material respect with the then applicable accounting requirements and with the published rules and regulations of the SEC applicable thereto; (ii) were prepared in accordance with GAAP throughout the periods covered, except as may be indicated in the notes to such financial statements and (in the case of unaudited statements) as permitted by Form 10-Q or Form 10-QSB, as applicable, and except that unaudited financial statements may not contain footnotes and are subject to year-end audit adjustments; and (iii) fairly present the consolidated financial position of Accessity and its subsidiaries as of the respective dates thereof and the consolidated results of operations of Accessity and its subsidiaries for the period covered thereby.

10.5 Valid Issuance. The Accessity Exchange Shares to be issued in the Share Exchange will, when issued in accordance with the provisions of this Agreement, be validly issued, fully paid and nonassessable.

10.6 Disclosure. None of the information in the Proxy Statement (as defined in Section 11.6 below) will contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The Proxy Statement will comply as to form in all material respects with the provisions of the Exchange Act and the rules and regulations promulgated by the SEC thereunder, except that no representation or warranty is made by Accessity with respect to statements made therein based on information supplied by any of the Acquired Companies for inclusion in the Proxy Statement.

10.7 Compliance With Applicable Laws. The business of Accessity and the businesses of the Accessity Subsidiaries (as defined below) are not being conducted in violation of any law, ordinance, regulation, rule or order of any Governmental Body where such violation would have a material adverse effect. Accessity has not been notified by any Governmental Body that any investigation or review with respect to Accessity or any of the Accessity Subsidiaries is pending or threatened, nor has any Governmental Body notified Accessity of its intention to conduct the same. Accessity and the Accessity Subsidiaries Exchange Shares to be issued in the Share Exchange will, when issued in accordance with the provisions of this Agreement, be validly issued, fully paid and nonassessable.

10.8 No Conflict. Neither the execution, delivery and performance of this Agreement nor the consummation of the transactions contemplated hereby or thereby nor compliance with the provisions hereof or thereof will: (i) conflict with, or result in any violations of, or cause a default (with or without notice or lapse of time, or both) under, or give rise to a right of termination, amendment, cancellation or acceleration of any obligation contained in, or the loss of any material benefit under, or result in the creation of any lien, security interest, charge or encumbrance upon any of the material properties or assets of Accessity or any of the Accessity Subsidiaries under, any term, condition or provision of (x) the articles of incorporation or bylaws of 46

mortgage, indenture, lease or other material agreement, judgment, order, decree, statute, law, ordinance, rule or regulation applicable to Accessity or any of the Accessity Subsidiaries or their respective properties or assets, other than any such conflicts, violations, defaults, losses, liens, security interests, charges, or encumbrances which, individually or in the aggregate, would not have a material adverse effect on Accessity or the business of Accessity; or (ii) require the affirmative vote of the holders of greater than a majority of the issued and outstanding shares of the common stock of Accessity.

10.9 Governmental Consents. No consent, approval, order or authorization of, or registration, declaration or filing with, any Governmental Body, is required to be obtained by Accessity in connection with the execution and delivery of this Agreement or the consummation of the transactions contemplated hereby, except for: (i) the filing with the SEC of (x) the Proxy Statement relating to the Accessity Shareholders' Meeting (as defined in Section 11.6 below), and (y) such reports and information under the Exchange Act and the rules and regulations promulgated by the SEC thereunder, as may be required in connection with this Agreement and the transactions contemplated hereby; (ii) such filings, authorizations, orders and approvals as may be required under state "control share acquisition," "anti-takeover" or other similar statutes and regulations (collectively, "State Takeover Laws"); (iii) such filings, authorizations, order and approvals as may be required under foreign laws, state securities laws and the Bylaws of the National Association of Securities Dealers, Inc. ("NASD"); (iv) such filings and notifications (if any) as may be necessary under the Hart-Scott-Rodino Antitrust Improvement Act of 1976, as amended ("HSR Act"); and (v) where the failure to obtain such consents, approvals, etc., would not prevent or delay the consummation of the Share Exchange or otherwise prevent Accessity from performing its obligations under this Agreement and would not reasonably be expected to have a material adverse effect on Accessity or the business of Accessity.

> ARTICLE XI CERTAIN OBLIGATIONS OF THE PARTIES PRIOR TO CLOSING

Accessity and the Acquired Companies hereby covenant as follows:

11.1 Access to Accessity Prior to Closing. Accessity shall afford the Acquired Companies and their respective counsel, accountants, investment bankers, investors and other authorized agents and representatives (their "Advisors") reasonable access during normal business hours to Accessity's properties, books, records and personnel in order that the Acquired Companies and their respective Advisors may have the opportunity to make such reasonable investigations as they shall desire to make of the affairs of Accessity. Accessity shall furnish, or shall cause its accountants to furnish, such additional financial and operating data and other information as any of the Acquired Companies or any of their Advisors shall from time to time reasonably request, including, without limitation, all financial and operating data as shall be necessary for verification of the accuracy of the financial statements of Accessity. Accessity shall, upon the reasonable request of any of the Acquired Companies, assist the Acquired Companies and their respective Advisors in contacting and communicating with suppliers, customers, employees and Advisors of Accessity.

11.2 Access to Acquired Companies Prior to Closing. Each Acquired Company shall afford Accessity and the other Acquired Companies and their respective Advisors reasonable access during normal business hours to such Acquired Company's properties, books, records and personnel in order that Accessity and the other Acquired Companies and their

47

respective Advisors may have the opportunity to make such reasonable investigations as they shall desire to make of the affairs of such Acquired Company. Each Acquired Company shall furnish, or shall cause its accountants to furnish, such additional financial and operating data and other information as Accessity or any of the other Acquired Companies or any of their respective Advisors shall from time to time reasonably request, including, without limitation, all financial and operating data as shall be necessary for verification of the accuracy of the financial statements of such Acquired Company. Such Acquired Company shall, upon the reasonable request of Accessity or any of the other Acquired Companies, assist Accessity and the Acquired Companies and their respective Advisors in contacting and communicating with suppliers, customers, employees and Advisors of such Acquired Company.

11.3 Confidentiality Prior to Closing. Except as required by law or any securities exchange, and subject to Section 17.1 below, each party hereto shall, and shall cause its officers and Advisors to, hold in strict confidence, and not disclose to others (except its Advisors) for any reason whatsoever, without the prior written consent of the other party, any nonpublic information received by it from the other party in connection with the transactions contemplated hereby and will not use such information for any purpose in the event that no Closing occurs under this Agreement. No party hereto shall communicate, directly or indirectly, to any third party other than their respective employees, agents and Advisors any of the terms, conditions and other aspects of this Agreement and the negotiation and preparation hereof until the Closing has occurred or the negotiations between the parties has terminated.

11.4 Conduct Prior to Closing Date. Except as otherwise contemplated by this Agreement, prior to the Closing Date:

(a) PEI shall:

(i) conduct the PEI Business and operations of PEI only in the ordinary course, including, without limitation, maintaining inventories at levels not in excess of those consistent with past practices;

(ii) maintain the properties and assets of PEI in good condition and repair and not dispose of any of its assets except in the ordinary course of business consistent with past practices, perform its obligations under all agreements to which it is a party or by which it or any of its assets or properties are bound and maintain all of its Permits in good standing;

(iii) continue in effect the policies of insurance (or similar coverage) referred to in Section 4.23;

(iv) not borrow any money except for amounts that are not in the aggregate material to the financial condition of PEI;

(v) use its commercially reasonable efforts to keep available the services of the present employees of PEI;

(vi) not declare, set aside or pay any cash or stock dividend or other distribution in respect of capital stock, or redeem or otherwise acquire any of its capital stock;

48

(vii) maintain and preserve the goodwill of the suppliers, customers and others having business relations with PEI;

(viii) not lend any amount to any person or entity, other than advances for travel and expenses which are incurred in the ordinary course of business consistent with past practices, not material in amount and documented by receipts for the claim amounts;

(ix) not guarantee or act as a surety for any obligation except for obligations in amounts that are not material; (x) not waive or release any right or claim except for the waiver or release of non-material claims in the ordinary course of business consistent with past practices;

(xi) not issue or sell any shares of its capital stock of any class (except upon the exercise of a bona fide option, warrant or other right to acquire such capital stock currently outstanding or conversion of any currently outstanding securities which are by their terms convertible in shares of its capital stock), or any other of its securities, or issue or create any warrants, obligations, subscriptions, options, convertible securities or other commitments to issue shares of capital stock, or accelerate the vesting of any outstanding option or other security;

(xii) not split or combine the outstanding shares of its capital stock

of any class or enter into any recapitalization or agreement affecting the number or rights of outstanding shares of its capital stock of any class or affecting any other of its securities;

(xiii) not form, merge, consolidate or reorganize with, or acquire, any entity; or

(xiv) not amend its articles of incorporation or bylaws.

(b) Kinergy shall:

(i) conduct the Kinergy Business and operations of Kinergy only in the ordinary course, including, without limitation, maintaining inventories at levels not in excess of those consistent with past practices;

(ii) maintain the properties and assets of Kinergy in good condition and repair and not dispose of any of its assets except in the ordinary course of business consistent with past practices, perform its obligations under all agreements to which it is a party or by which it or any of its assets or properties are bound and maintain all of its Permits in good standing;

(iii) continue in effect the policies of insurance (or similar coverage) referred to in Section 6.23;

(iv) not borrow any money except for amounts that are not in the aggregate material to the financial condition of Kinergy;

(v) use its commercially reasonable efforts to keep available the services of the present employees of Kinergy;

49

(vi) not declare, set aside or pay any cash or dividend or other distribution in respect of its limited liability company membership interests, or redeem or otherwise acquire any of its limited liability company membership interests; provided, however, that effective the close of business on the day preceding the Closing Date, the Managers or Managing Members of Kinergy shall distribute to the Members of Kinergy in the form of cash, a promissory note or a combination of cash and a promissory note, the dollar amount of Kinergy's net worth as set forth on Kinergy's balance sheet dated as of such date, which balance sheet shall have been prepared in accordance with GAAP;

(vii) maintain and preserve the goodwill of the suppliers, customers and others having business relations with Kinergy;

(viii) not lend any amount to any person or entity, other than advances for travel and expenses which are incurred in the ordinary course of business consistent with past practices, not material in amount and documented by receipts for the claim amounts;

(ix) not guarantee or act as a surety for any obligation except for obligations in amounts that are not material;

 $({\rm x})$ not waive or release any right or claim except for the waiver or release of non-material claims in the ordinary course of business consistent with past practices;

(xi) not issue or sell any limited liability company membership interests, or any other of its securities, or issue or create any warrants, obligations, subscriptions, options, convertible securities or other commitments to issue limited liability company membership interests, or accelerate the vesting of any outstanding option or other security;

(xii) not enter into any recapitalization or agreement affecting the number or rights of outstanding limited liability company membership interests or affecting any other of its securities;

(xiii) not form, merge, consolidate or reorganize with, or acquire, any entity; or

(xiv) not amend its articles of organization or operating agreement.

(c) Reenergy shall:

(i) conduct the Reenergy Business and operations of Reenergy only in the ordinary course, including, without limitation, maintaining inventories at levels not in excess of those consistent with past practices;

(ii) maintain the properties and assets of Reenergy in good condition and repair and not dispose of any of its assets except in the ordinary course of business consistent with past practices, perform its obligations under all agreements to which it is a party or by which it or any of its assets or properties are bound and maintain all of its Permits in good standing;

50

(iii) continue in effect the policies of insurance (or similar coverage) referred to in Section 8.23;

(iv) not borrow any money except for amounts that are not in the aggregate material to the financial condition of Reenergy;

(v) use its commercially reasonable efforts to keep available the services of the present employees of Reenergy;

(vi) not declare, set aside or pay any cash or dividend or other distribution in respect of its limited liability company membership interests, or redeem or otherwise acquire any of its limited liability company membership interests; provided, however, that effective the close of business on the day preceding the Closing Date, the Managers or Managing Members of Reenergy shall distribute to the Members of Reenergy in the form of cash, a promissory note or a combination of cash and a promissory note, the dollar amount of Reenergy's net worth as set forth on Reenergy's balance sheet dated as of such date, which balance sheet shall have been prepared in accordance with GAAP;

(vii) maintain and preserve the goodwill of the suppliers, customers and others having business relations with Reenergy;

(viii) not lend any amount to any person or entity, other than advances for travel and expenses which are incurred in the ordinary course of business consistent with past practices, not material in amount and documented by receipts for the claim amounts;

(ix) not guarantee or act as a surety for any obligation except for obligations in amounts that are not material; (x) not waive or release any right or claim except for the waiver or release of non-material claims in the ordinary course of business consistent with past practices;

(xi) not issue or sell any limited liability company membership interests, or any other of its securities, or issue or create any warrants, obligations, subscriptions, options, convertible securities or other commitments to issue limited liability company membership interests, or accelerate the vesting of any outstanding option or other security; (xii) not enter into any recapitalization or agreement affecting the number or rights of outstanding limited liability company membership interests or affecting any other of its securities;

 $% \left(x \right)$ (xiii) not form, merge, consolidate or reorganize with, or acquire, any entity; or

(xiv) not amend its articles of organization or operating agreement.

(d) Accessity shall:

(i) conduct the business and operations of Accessity only in the ordinary course, including, without limitation, maintaining inventories at levels not in excess of those consistent with past practices;

51

(ii) maintain the properties and assets of Accessity in good condition and repair and not dispose of any of its assets except in the ordinary course of business consistent with past practices, perform its obligations under all agreements to which it is a party or by which it or any of its assets or properties are bound and maintain all of its Permits in good standing; (iii) not change any insurance coverage;

(iv) not borrow any money except for amounts that are not in the aggregate material to the financial condition of Accessity;

 (\mathbf{v}) use its commercially reasonable efforts to keep available the services of the present employees of Accessity, except as otherwise provided for herein;

(vi) not declare, set aside or pay any cash or stock dividend or other distribution in respect of capital stock, or redeem or otherwise acquire any of its capital stock;

(vii) maintain and preserve the goodwill of the suppliers, customers and others having business relations with Accessity;

(viii) not lend any amount to any person or entity, other than advances for travel and expenses which are incurred in the ordinary course of business consistent with past practices, not material in amount and documented by receipts for the claim amounts;

(ix) not guarantee or act as a surety for any obligation except for obligations in amounts that are not material;

 (x) not waive or release any right or claim except for the waiver or release of non-material claims in the ordinary course of business consistent with past practices;

(xi) not issue or sell any shares of its capital stock of any class (except upon the exercise of a bona fide option, warrant or other right to acquire such capital stock currently outstanding or conversion of any currently outstanding securities which are by their terms convertible in shares of its capital stock), or any other of its securities, or issue or create any warrants, obligations, subscriptions, options, convertible securities or other commitments to issue shares of capital stock, or accelerate the vesting of any outstanding option or other security;

(xii) not split or combine the outstanding shares of its capital stock of any class or enter into any recapitalization or agreement affecting the number or rights of outstanding shares of its capital stock of any class or affecting any other of its securities;

(xiii) not form, merge, consolidate or reorganize with, or acquire, any entity (other than in connection with the reincorporation of Accessity in Delaware); or

(xiv) not amend its articles of incorporation or bylaws (except as may be necessary in connection with the reincorporation of Accessity in Delaware).

11.5 Accessity Special Shareholders' Meeting. Accessity shall, in accordance with its articles of incorporation and bylaws and the applicable requirements of New York law, call and hold a special meeting of its shareholders as promptly as practicable for the purpose of

52

permitting them to consider and to vote upon and approve the Share Exchange and the transactions contemplated by this Agreement, the reincorporation of Accessity in the State of Delaware referred to in Section 13.6 below, the sale or other disposition of the Accessity Subsidiaries (as defined below) to Barry Siegel referred to in Section 13.11 below, the adoption of an Amended and Restated 1995 Incentive Stock Plan of Accessity, in form and substance reasonably acceptable to the Acquired Companies, and the adoption of a new stock option plan referred to in Section 13.16 below, in form and substance reasonably acceptable to the Acquired Companies) (the "Accessity Special Shareholders' Meeting"). As soon as permissible under all applicable Legal Requirements, Accessity shall cause a copy of the Proxy Statement (as defined in Section 11.6 below) to be delivered to each shareholder of Accessity who is entitled to vote on such matter under its articles of incorporation and bylaws and the applicable requirements of New York law.

11.6 Proxy Statement. Following delivery to Accessity of the audited

balance sheets as of December 31, 2003 and the related statements of income and changes in financial position or cash flows, as appropriate, for the period then ended for each of the Acquired Companies and subject to the reasonable satisfaction of Accessity with same and with the results of its due diligence investigation of the Acquired Companies as of such time, promptly thereafter, Accessity shall prepare and cause to be filed with the SEC a Proxy Statement with respect to the Accessity Special Shareholders' Meeting (the "Proxy Statement") and any other documents required by the Securities Act, the Exchange Act or any other federal, foreign or state Blue Sky or related laws in connection with the Share Exchange and the transactions contemplated by this Agreement (collectively, "Other Filings"). Accessity will notify each of the Acquired Companies of any comments from the SEC or its staff or any other Governmental Body and of any request by the SEC or its staff or any other Governmental Body for amendments to the Proxy Statement or any Other Filings or for additional information and will supply each of the Acquired Companies with copies of all correspondence between Accessity and any of its Advisors or representatives, on the one hand, and the SEC or its staff or any other Governmental Body, on the other hand, with respect to the Proxy Statement, any Other Filings or the Share Exchange. Accessity shall use all commercially reasonable efforts to cause the Proxy Statement and any Other Filings to comply with the rules and regulations promulgated by the SEC and to respond promptly to any comments of the SEC or its staff or any other Governmental Body. Accessity will use all reasonable efforts to cause the Proxy Statement to be mailed to Accessity's shareholders, as promptly as practicable after the Proxy Statement is permitted to be mailed under the rules and regulations promulgated by the SEC. Each of the Acquired Companies shall promptly furnish to Accessity all information concerning such Acquired Company and such Acquired Company's shareholders that may be required or reasonably requested in connection with any action contemplated by this Section 11.6.

11.7 Nonpublic Offering; Preparation of Disclosure Document. Prior to the Closing Date, Accessity shall use all commercially reasonable efforts to obtain or comply with all regulatory approvals and provisions of any and all applicable Governmental Bodies to ensure that the Accessity Exchange Shares to be issued in the Share Exchange will be exempted from registration or qualification under the securities law of every jurisdiction of the United States in which any Owner has an address of record as set forth on the signature pages hereto; provided, that Accessity shall not be required (i) to qualify to do business as a foreign corporation in any jurisdiction in which it is not now qualified or (ii) file a general consent to service of process in any jurisdiction in which it has not already filed a general consent to service of process. As soon

53

as practicable after the execution of this Agreement, Accessity shall have prepared a document containing certain disclosures, risk factors and disclaimers in regard to Accessity, PEI, Kinergy and Reenergy and the Share Exchange for distribution to the Owners and the PEI Warrantholders prior to the Closing, subject to the review, comment and approval of each of PEI, Kinergy and Reenergy prior to such distribution (the "Owner Disclosure Document").

11.8 Cooperation. Each party shall use its best efforts to cause the transactions contemplated by this Agreement to be consummated, and without limiting the generality of the foregoing, to obtain all consents and authorizations of any Governmental Body and third parties listed on Schedules 4.7, 6.7, and 8.7, and to make all filings with and give all notices to government agencies and third parties which may be necessary or reasonably required in order to consummate the transactions contemplated by this Agreement. Each party shall give prompt notice to Accessity and the Acquired Companies, after receipt thereof by such party, of (i) any notice of, or other communication relating to, any default or event which, with notice or the lapse of time or both, would be reasonably likely to become a default under any indenture, instrument or agreement material to any of the Acquired Companies or Accessity or the operations, condition (financial or otherwise) of any of the Acquired Companies or Accessity, or to which any of the Acquired Companies or Accessity is a party or by which any of the Acquired Companies or Accessity or their respective assets or properties are bound and (ii) any notice or other communication from or to any third party alleging or stating that the consent of such third party is or may be required in connection with the transactions contemplated by this Agreement.

11.9 No Negotiations, etc. Prior to the Closing Date, neither Accessity

nor any of the Acquired Companies or the Owners shall, directly or indirectly, in any way contact, initiate, enter into or conduct any discussions or negotiations, or enter into any agreements, whether written or oral, with any Person with respect to the sale of all or any part of the assets of any of the Acquired Companies or Accessity or a merger or consolidation of any of the Acquired Companies or Accessity with any other person (other than in connection with the reincorporation of Accessity in Delaware), except, by the Board of Directors of Accessity or PEI or the managers of any of the Acquired Companies to the extent otherwise required in the exercise of its fiduciary duties to its shareholders or members, as the case may be, if it shall have received a Superior Proposal after the date hereof from a third party or parties. As used in this Agreement, the defined term "Superior Proposal" shall mean a bona fide unsolicited written proposal made by a third party which is (a)(i) for a sale, exchange, transfer or other disposition of more than 50% of the assets of the company, taken as a whole, in a single transaction or a series of related transactions, or (ii) for the acquisition, directly or indirectly, by such third party of beneficial ownership of more than 50% of the stock or limited liability company membership interests of the company, as the case may be, whether by merger, reorganization, consolidation, share exchange or purchase, business combination, recapitalization, liquidation, dissolution or similar transaction, and which and is (b) otherwise on terms which the Board of Directors or managers of the company, as the case may be, in good faith has concluded (after consultation with its financial advisors and legal counsel), taking into account, among other things, all legal, financial, regulatory and other aspects of the proposal and the third party making the proposal (x) that the proposal would, if consummated, result in a transaction that is more favorable to its shareholders (in their capacity as shareholders) or members (in their capacity as members), as the case may be, from a financial point of view, than the transactions contemplated by this Agreement and (y) that the proposal is reasonably capable of being consummated.

54

11.10 Delivery of Disclosure Schedules. The respective disclosure schedules relating to the representations and warranties of Accessity and each of the Acquired Companies in this Agreement shall be delivered as soon as practicable after such execution and, in any event, a reasonable time prior to the Closing.

11.11 Distribution to Owners and PEI Warrantholders. As soon as practicable after delivery of the respective disclosures schedules relating to the representations and warranties of the Acquired Companies in this Agreement and completion of the Owner Disclosure Document and the Proxy Statement, each of the Acquired Companies shall distribute to its respective Owners (and PEI will also distribute to the PEI Warrantholders) for execution a copy of this Agreement, together with a copy of such disclosure schedules, a copy of the Owner Disclosure Document and a copy of the Proxy Statement.

ARTICLE XII CONDITIONS TO ACCESSITY'S OBLIGATIONS

Each and every obligation of Accessity under this Agreement to be performed on or before the Closing Date shall be subject to the satisfaction, on or before the Closing Date, of each of the following conditions:

12.1 Representations and Warranties True. The representations and warranties of each of the Acquired Companies and the Owners contained herein, in the Schedules and Exhibits hereto and in all certificates and other documents delivered by each of the Acquired Companies and the Owners to Accessity pursuant hereto or in connection with the transactions contemplated hereby shall be true and accurate as of the date of this Agreement and as of the Closing Date with the same effect as if made on and as of the Closing Date.

12.2 Performance. Each of the Acquired Companies shall have performed and complied in all material respects with all agreements, obligations and conditions required by this Agreement to be performed or complied with by it on or prior to the Closing Date, including, without limitation, those referred to in Article XI above.

12.3 Permits. The Acquired Companies shall have obtained all such Permits (other than nondiscretionary Permits issued upon completion of construction of the proposed ethanol plant) from federal, state and local Governmental Bodies as are required by applicable law for the Acquired Companies to conduct each of the PEI Business, the Kinergy Business and the Reenergy Business (including, without limitation, the commencement of construction of a plant to produce up to 40 million gallons of ethanol per year), unless the failure to obtain any such Permit would not have an adverse effect on the assets, properties, business, condition (financial or otherwise) or prospects of Accessity or any of the Acquired Companies or the transactions contemplated hereby.

12.4 Governmental Consents. There shall have been obtained on or before the Closing such material permits or authorizations, and there shall have been taken such other action, as may be required to consummate the Share Exchange and the other transactions contemplated hereby by any Governmental Body having jurisdiction over the parties and the actions herein proposed to be taken, including but not limited to requirements under applicable federal and state securities laws and the compliance with, and expiration of any applicable waiting period for, the HSR Act.

55

12.5 Documents and Actions. Each of Acquired Companies shall have executed and delivered to Accessity this Agreement and the other agreements, documents and instruments and shall have taken the actions contemplated by Section 3.3 hereof.

12.6 Third Party Approvals and Consents. Each Acquired Company shall have received any and all consents, approvals, notices, filings or recordations of third parties required with respect to the execution and delivery of this Agreement or the transactions contemplated hereby or by any of the agreements, documents or instruments referred to herein.

12.7 No Legal Action. No temporary restraining order, preliminary injunction or permanent injunction or other order preventing the consummation of the Share Exchange or any of the transactions contemplated hereby shall have been issued by any Federal or state court and remain in effect, nor shall any proceeding initiated by any Governmental Body seeking any of the foregoing be pending. 12.8 No Material Adverse Change. There shall have been no material adverse change in the business, financial condition, operations or financial performance of any of the Acquired Companies since the date of this Agreement.

12.9 Consulting Agreements. Accessity shall have entered into a consulting agreement with Barry Siegel for advisory services, in form and substance mutually acceptable to the Acquired Companies, Accessity and Barry Siegel and Accessity shall have entered into a consulting agreement with Philip Kart for advisory services, in form and substance mutually acceptable to the Acquired Companies, Accessity and Philip Kart.

12.10 Proxy Statement. The Proxy Statement shall on the Closing Date not be subject to any proceedings commenced or threatened by the SEC.

12.11 Audited Financial Statements. Each of the Acquired Companies shall have delivered to Accessity the audited balance sheet of such Acquired Company as of December 31, 2003 and the related statements of income and changes in financial position or cash flows, as appropriate, for the period then ended.

12.12 Execution of Agreement by PEI Shareholders, PEI Warrantholders, Kinergy Members and Reenergy Members. All or substantially all (representing at least 95% of the issued and outstanding shares of PEI Stock as of the Closing) of the PEI Shareholders, all of the PEI Warrantholders, all of Kinergy Members and all of the Reenergy Members shall have executed and delivered to PEI, Kinergy or Reenergy, respectively, a copy of this Agreement and, in any event, a sufficient number of PEI Shareholders shall have executed a copy of this Agreement such that, after giving effect to the Share Exchange, the PEI Shareholders, the Kinergy Members and the Reenergy Members shall beneficially own in the aggregate at least 80% of the issued and outstanding common stock of Accessity.

12.13 Approval by Accessity Shareholders. The shareholders of Accessity shall have approved the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby (including, without limitation, the reincorporation of Accessity in the State of Delaware referred to in Section 13.6 below, the sale or other disposition of the Accessity Subsidiaries (as defined below) to Barry Siegel referred to in Section 13.13

56

form and substance reasonably acceptable to the Acquired Companies, and the adoption of a new stock option plan referred to in Section 13.16 below, in form and substance reasonably acceptable to the Acquired Companies)).

12.14 Limitation of Outstanding Capital Stock. As of the Closing Date, giving effect to the transactions contemplated hereby, Accessity shall not be obligated to issue to the PEI Shareholders, the Kinergy Members and the Reenergy Members or any other Person(s), more than 18,800,000 Accessity Exchange Shares in connection with the Share Exchange and the consummation of the transactions contemplated hereby on a fully-diluted basis (including shares of capital stock issuable upon exercise of any and all options, calls, warrants, claims, convertible debt and any other rights to acquire shares of capital stock of any of the Acquired Companies, whether accrued or contingent (including shares issuable upon exercise of the Accessity Replacement Warrants).

12.15 Completion of Due Diligence; Disclosure Schedules. Accessity shall have completed its financial and legal due diligence investigation of each of the Acquired Companies with results thereof satisfactory to Accessity in its sole discretion (including, without limitation, satisfaction with matters related to Security Markets (as defined in Section 14.4 below) and litigation and any and all disclosures contained in the respective disclosure schedules). In this regard, each of Accessity and the Acquired Companies acknowledge and agree that the respective disclosure schedules relating to the representations and warranties of Accessity and the Acquired Companies in this Agreement are not required to be delivered as of the time of execution of this Agreement by Accessity and the Acquired Companies, but are required to be delivered as soon as practicable after such execution and, in any event, a reasonable time prior to the Closing, to permit the parties to review, evaluate and approve the disclosures made therein as a part of their due diligence investigation. Notwithstanding the absence of such disclosure schedules as of the time of execution of this Agreement, each of Accessity and the Acquired Companies acknowledge and agree the representations and warranties made herein by Accessity and the Acquired Companies shall not be deemed false or misleading or deemed to contain untrue statements of material fact or to have omitted to state material facts solely because of the absence of such disclosure schedules as of the time of execution of this Agreement.

12.16 Fairness Opinion. Accessity shall have received a fairness opinion regarding the sale or otherwise disposition of its two subsidiaries, DriverShield CRM Corp., a Delaware corporation, and Sentaur Corp., a Florida corporation (collectively, the "Accessity Subsidiaries"), in consideration of the waiver by Barry Siegel of the change in control provisions set forth in the employment agreement between Accessity and Barry Siegel that expires on December 31, 2004.

Each and every obligation of the Acquired Companies, each of the PEI Warrantholders and each of the Owners under this Agreement to be performed on or before the Closing Date shall be subject to the satisfaction, on or before the Closing Date, of each of the following conditions:

57

13.1 Representations and Warranties True. The representations and warranties of Accessity contained herein and in all certificates and other documents delivered by Accessity to each of the Acquired Companies or the Owners pursuant hereto or in connection with the transactions contemplated hereby shall be true and accurate on and as of the Closing Date with the same effect as if made on and as of the Closing Date.

13.2 Performance. Accessity shall have performed and complied with all agreements, obligations and conditions required by this Agreement to be performed or complied with by it on or prior to the Closing Date, including,

without limitation, those referred to in Article XI above.

13.3 Documents and Actions. Accessity shall have executed and delivered to the each of the Acquired Companies and each of the Owners the agreements, documents and instruments and shall have taken the actions contemplated by Section 3.2.

13.4 Approval by Accessity Shareholders. The shareholders of Accessity shall have approved the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby (including, without limitation, the reincorporation of Accessity in the State of Delaware referred to in Section 13.6 below, the sale or other disposition of the Accessity Subsidiaries (as defined below) to Barry Siegel referred to in Section 13.11 below, the adoption of an Amended and Restated 1995 Incentive Stock Plan of Accessity, in form and substance reasonably acceptable to the Acquired Companies, and the adoption of a new stock option plan referred to in Section 13.16 below, in form and substance reasonably acceptable to the Acquired Companies)).

13.5 Resignations of Officers and Directors. At Closing, all of the directors of Accessity and the officers of Accessity identified in subsection (f) of Section 3.2 shall have resigned in writing from their positions as officers and employees of Accessity, effective upon the appointment of the individuals identified in subsection (d) of Section 3.2 above to the Board of Directors of Accessity.

13.6 Reincorporation in Delaware; Authorized Capital. Accessity shall have reincorporated, changed its name to Pacific Ethanol, Inc. and become duly organized, validly existing and in good standing under the laws of the State of Delaware in compliance with all applicable federal, state and applicable laws, rules and regulations and shall have sufficient shares of its capital stock authorized for issuance in order to complete the Share Exchange and the transactions contemplated hereby.

13.7 Valid Issuance of Accessity Exchange Shares. The shares of restricted Accessity common stock to be issued to the PEI Shareholders, the Kinergy Members and the Reenergy Members at Closing will be validly issued, fully paid and nonassessable under applicable law and will have been duly issued in a non-public offering in compliance with all applicable federal and state securities laws.

13.8 Third Party Approvals and Consents. Each Acquired Company shall have received any and all consents, approvals, notices, filings or recordations of third parties required with respect to the execution and delivery of this Agreement or the transactions contemplated hereby or by any of the agreements, documents or instruments referred to herein.

58

13.9 Cash Balance. As of the Closing Date, Accessity shall have at least the same Cash Balance as reported in its Quarterly Report on Form 10-QSB for the quarter ended March 31, 2004 filed with the SEC (approximately \$4,360,000), subject to adjustment as set forth on Schedule 13.9 hereto, and Accessity shall have delivered to the Acquired Companies a certificate of the president or chief executive officer of Accessity certifying to the foregoing. As used herein, the defined term "Cash Balance" shall mean the sum of (i) the amount of cash in Accessity's operating and disbursement bank accounts at JPMorgan-Chase Bank, (ii) the amount of Accessity funds invested in highly liquid investments in fixed income mutual funds at Solomon Smith Barney (with available liquidity on a next-day basis) and (iii) a \$300,000 restricted certificate of deposit with Bank Atlantic in Coral Springs, Florida established in connection with the lease agreement referred to in Section 13.11 below (which is to be released in \$100,000 increments in the 36th, 48th and 60th months of such lease agreement), excluding any cash received from the exercise of any outstanding options or warrants.

13.10 Limitation of Outstanding Capital Stock. As of the Closing Date, without giving effect to the transactions contemplated hereby, Accessity shall have no more than 2,638,081 of capital stock issued and outstanding on a fully-diluted basis (including shares of capital stock issuable upon exercise of any and all options, calls, warrants, claims and any other rights to acquire shares of capital stock of Accessity, whether accrued or contingent, other than an aggregate of 600,000 shares of common stock of Accessity to be issued and

beneficially owned by Barry Siegel and Philip Kart and up to 100,000 shares of common stock of Accessity issuable upon conversion of the issued and outstanding shares of Series A Convertible Preferred Stock of Accessity).

13.11 Disposition of Accessity Subsidiaries and Waiver of Change of Control Provisions by Barry Siegel. Prior to Closing, Accessity shall have sold or otherwise disposed of its two subsidiaries, DriverShield CRM Corp., a Delaware corporation, and Sentaur Corp., a Florida corporation (collectively, the "Accessity Subsidiaries"), pursuant to a written agreement between Accessity and Barry Siegel, in form and substance reasonably satisfactory to PEI, Kinergy and Reenergy, in consideration of the waiver by Barry Siegel of the change in control provisions set forth in the employment agreement between Accessity and Barry Siegel that expires on December 31, 2004, as the same would be applicable to the consummation of the transactions contemplated by this Agreement (including, but not limited to, the provisions that require Accessity to pay to Barry Siegel (i) a severance payment of 300% of his average annual salary for the past five years, less \$100; (ii) the cash value of his outstanding but unexercised stock options; and (iii) for any and all other perquisites in the event that he is terminated for various reasons specified in such agreement following a change of control (as defined in such agreement). Without in any way limiting the foregoing, as part of the disposition of the Accessity Subsidiaries to Barry Siegel, the facilities of Accessity located in Coral Springs, Florida shall have been duly subleased to Barry Siegel or an entity owned or controlled by Barry Siegel (which may be Sentaur Corp.) with the consent of the lessor under the existing lease agreement for such facilities, on terms and conditions reasonably satisfactory to the Acquired Companies. The parties acknowledge and agree that the personal property at the facilities of Accessity located in Coral Springs, Florida shall also be transferred to Barry Siegel or an entity owned or controlled by Barry Siegel (which may be Sentaur Corp.).

13.12 Governmental Consents. There shall have been obtained on or before the Closing such material permits or authorizations, and there shall have been taken such other action, as may be required to consummate the Share Exchange and the other transactions

59

contemplated hereby by any Governmental Body having jurisdiction over the parties and the actions herein proposed to be taken, including but not limited to requirements under applicable federal and state securities laws.

13.13 No Legal Action. No temporary restraining order, preliminary injunction or permanent injunction or other order preventing the consummation of the Share Exchange or any of the transactions contemplated hereby shall have been issued by any Federal or state court and remain in effect, nor shall any proceeding initiated by any Governmental Body seeking any of the foregoing be pending.

13.14 No Material Adverse Change. There shall have been no material adverse change in the business, financial condition, operations or financial performance of Accessity since the date of this Agreement.

13.15 Proxy Statement. The Proxy Statement shall on the Closing Date not be subject to any proceedings commenced or threatened by the SEC.

13.16 Stock Option Plan. Prior to Closing, Accessity shall have adopted a new stock option plan in the manner requested by PEI.

13.17 Retirement or Conversion of Preferred Stock of Accessity. Prior to Closing, Accessity shall have retired all of the issued and outstanding shares of Series A Convertible Preferred Stock of Accessity, on terms and conditions that are reasonably acceptable to PEI, Kinergy and Reenergy, or all of the issued and outstanding shares of Series A Convertible Preferred Stock of Accessity shall have been duly converted into shares of common stock of Accessity in accordance with the terms and provisions set forth in the Certificate of Incorporation of Accessity.

13.18 Execution of Agreement by PEI Shareholders, PEI Warrantholders, Kinergy Members and Reenergy Members. All or substantially all (representing at least 95% of the issued and outstanding shares of PEI Stock as of the Closing) of the PEI Shareholders, all of the PEI Warrantholders, all of Kinergy Members and all of the Reenergy Members shall have executed and delivered to PEI, Kinergy or Reenergy, respectively, a copy of this Agreement and, in any event, a sufficient number of PEI Shareholders shall have executed a copy of this Agreement such that, after giving effect to the Share Exchange, the PEI Shareholders, the Kinergy Members and the Reenergy Members shall beneficially own in the aggregate at least 80% of the issued and outstanding common stock of Accessity.

13.19 Establishment of Escrow Relating to Mercator Action. Prior to Closing, Accessity shall have established an escrow account with an escrow agent mutually acceptable to PEI and Accessity into which Accessity will deposit the net proceeds from a recovery from the current arbitration proceeding with Presidion Solutions, Inc., which shall thereafter be used solely to fund the legal fees, expenses and disbursements incurred in connection with the Mercator Action. After full and final settlement or other final resolution of the Mercator Action, Accessity shall cause the net proceeds from the Mercator Action received from the law firms representing Accessity in the Mercator Action (i.e., after retention by the law firms representing Accessity in this action, of the additional fees and expenses pursuant to the engagement agreement between the law firms and Accessity and an amount equal to twenty-five percent (25%) of the gross proceeds from the Mercator Action as full payment for these law firms'

60

representation of Accessity in the Mercator Action) to be deposited into such escrow account and thereafter to be distributed on a pro rata basis, to the fullest extent permitted under applicable law, to the holders of record of shares of Accessity common stock as of the date immediately prior to the Closing Date (for further distribution to the beneficial owners of shares of Accessity common stock as of such date, as applicable), after payment of any and all other fees and expenses in connection with the Mercator Action and with respect to such escrow arrangement, including, without limitation, a fee in an amount equal to one-third of the net proceeds from the Mercator Action received from the law firms representing Accessity in the Mercator Action which shall be paid to Accessity prior to such distribution.

13.20 Completion of Due Diligence; Disclosure Schedules. Each of the Acquired Companies shall have completed its financial and legal due diligence investigation of Accessity and each of the other Acquired Companies with results thereof satisfactory to such Acquired Company in its sole discretion (including, without limitation, satisfaction with matters related to Security Markets (as defined in Section 14.4 below), and litigation and any and all disclosures contained in the respective disclosure schedules). In this regard, each of Accessity and the Acquired Companies acknowledge and agree that the respective disclosure schedules relating to the representations and warranties of Accessity and the Acquired Companies in this Agreement are not required to be delivered as of the time of execution of this Agreement by Accessity and the Acquired Companies, but are required to be delivered as soon as practicable after such execution and, in any event, a reasonable time prior to the Closing, to permit the parties to review, evaluate and approve the disclosures made therein as a part of their due diligence investigation. Notwithstanding the absence of such disclosure schedules as of the time of execution of this Agreement, each of Accessity and the Acquired Companies acknowledge and agree that the representations and warranties made herein by Accessity and the Acquired Companies shall not be deemed false or misleading or deemed to contain untrue statements of material fact or to have omitted to state material facts solely because of the absence of such disclosure schedules as of the time of execution of this Agreement.

ARTICLE XIV POST-CLOSING COVENANTS

14.1 Certain Reporting Matters. Upon consummation of the Closing, Accessity shall timely file with the SEC a Current Report on Form 8-K with respect to the Share Exchange and the consummation of the transactions contemplated hereby.

14.2 Standard and Poors. Upon consummation of the Closing, to the extent required or otherwise deemed advisable by the Majority Parties, Accessity shall apply for listing with Standard and Poors Information Service and Blue Sky filings.

14.3 Books and Records. Unless otherwise consented to in writing by Accessity, neither the Acquired Companies nor any of the Owners shall destroy, alter or otherwise dispose of any original books or records of any of the

Acquired Companies prior to the Closing Date without first offering to surrender such books and records to Accessity and shall maintain such books and records in good condition in a reasonably accessible location.

14.4 Listing. If the shares of common stock of Accessity are listed on the Nasdaq SmallCap Market of the Nasdaq Stock Market, the Nasdaq National Market of the Nasdaq Stock Market, the American Stock Exchange or the New York Stock Exchange (these securities

61

markets collectively referred to as the "Security Markets" and, individually as a Security Market") as of the Closing Date, Accessity shall use its commercially reasonable efforts to maintain a listing on a Security Market for a period of one (1) year after the Closing Date.

14.5 Grant of Stock Options and Warrants. Upon consummation of the Closing and for a period of one (1) year after the Closing Date, Accessity shall not grant or issue any security, stock option or warrant (other than stock options contemplated to be issued pursuant to the new stock option plan referred to in Section 13.16 above, which would not be exercisable until one (1) year after the Closing Date) that may be sold in the Security Markets after having been registered under the Securities Act or traded pursuant to an exemption under the Securities Act, except (i) as otherwise contemplated by this Agreement; (ii) the sale of securities by Accessity pursuant to equity and/or debt financings to provide funds for working capital purposes; (iii) the Accessity Replacement Warrants; and (iv) the Accessity Exchange Shares which are subject to registration rights of certain PEI Shareholders pursuant to the terms of the various Registration Rights Agreement between PEI and such PEI Shareholders.

14.6 Board Seat. As provided in subsection (d) of Section 3.2 above, immediately prior to the Closing, the Board of Directors of Accessity shall have designated an individual to serve on the Board of Directors of Accessity as a Class II director (thereby holding such board seat until the annual meeting of Accessity shareholders to be held in the fourth calendar quarter of 2005).

14.7 Continuation and Prosecution of Lawsuit.

(a) Upon consummation of the Closing, Accessity shall continue to use its commercially reasonable efforts to vigorously prosecute the lawsuit previously filed against Mercator Group, LLC, Mercator Advisory Group, LLC, Mercator Momentum Fund, LP, Mercator Momentum Fund III, LP and Mercator Focus Fund, LP, Taurus Global, LLC and the action that was previously filed, dismissed without prejudice and may be re-filed against John W. Burcham, II, Craig A. Vanderburg, James E. Baiers and MediaBus Networks, Inc. n/k/a Presidion Corp., in the Circuit Court of the 11th Judicial Circuit in and for Miami-Dade County, Florida, Case No. 03-30243 CA 15, or as removed or re-filed in such other venue, as the case may be (the "Mercator Action"). The Mercator Action shall not be settled, to the fullest extent permitted by applicable law, without the express written consent (which consent shall not be unreasonably withheld or delayed) of: (i) the individual sitting on the Board of Directors and designated as the Accessity Board Member or (ii) should such Accessity Board Member no longer serve on the Board of Directors, then by the unanimous written consent of all the independent members of the Board of Directors of Accessity. The proceeds from any recovery or settlement in the Mercator Action shall be allocated as provided in Section 13.19 above.

(b) Upon consummation of the Closing, Accessity shall continue to use its commercially reasonable efforts to vigorously collect an arbitration award that it may have received from the American Arbitration Association against Presidion Solutions, Inc. The proceeds of any recovery from Presidion Solutions, Inc. shall be deposited into the escrow account pursuant to Section 13.19 above.

62

14.8 Reimbursement of Costs Paid by Reenergy Members. As soon as practicable after the Closing Date, PEI shall pay to the Reenergy Members or shall otherwise have caused the Reenergy Members to be paid an amount equal to \$150,000, as payment and reimbursement for the services, costs and expenses of the Reenergy Members in connection with the development of the Visalia project.

SURVIVAL OF REPRESENTATIONS AND WARRANTIES; INDEMNIFICATION

15.1 Survival of Representations and Warranties. Notwithstanding (a) the making of this Agreement or any related agreement, (b) any examination made by or on behalf of the parties hereto and (c) the Closing hereunder, (x) the representations and warranties of the Acquired Companies, the Owners and Accessity contained in this Agreement, or in any document delivered pursuant to the provisions of this Agreement, shall survive the Closing for a period of twenty-four (24) months, except for the representations and warranties made in Sections 4.11, 6.11, and 8.11 (Taxes and Tax Returns), Sections 4.13, 6.13, and 8.13 (Title to Properties; Encumbrances), Sections 4.15, 6.15, and 8.15(Intellectual Property), Sections 4.16, 6.16, and 8.16 (Compliance with Laws), and Sections 4.21, 6.21 and 8.21 (Environmental Matters), which in each case shall survive until the expiration of the applicable statute of limitations for the underlying cause of action plus six (6) months and (y) the covenants and agreements required to be performed after the Closing or pursuant to Article XI of this Agreement (unless noncompliance with those covenants contained in Article XI was waived in writing at the Closing) shall survive until fully performed or fulfilled. No action may be brought with respect thereto after such date; provided, however, that if, prior to such date, one party has notified the other party of a claim for indemnity under this Article XV (whether or not formal legal action shall have been commenced based upon such claim), such claim shall continue to be subject to indemnification in accordance with this Article XV.

15.2 Indemnification. The parties shall indemnify each other as set forth below:

(a) Subject to Section 15.1 above, from and after the Closing, (x) each of the Acquired Companies shall, severally and not jointly, indemnify and save harmless Accessity and its officers, directors, shareholders, successors and assigns from and against any loss, claim, liability, damage, complaint, action or causes of action, suits, proceedings, investigations, punitive damages, remedial costs, civil and criminal penalties or expenses, costs or other damages of any kind or nature, including reasonable attorneys' fees incurred in connection with any of the foregoing (collectively, the "Damages"), to the extent arising from, relating to or otherwise in respect of (i) the inaccuracy or breach of any representation or warranty of such Acquired Company contained in this Agreement (as of the date hereof, or as of the Closing Date) or of any representation, warranty or statement made in any schedule, certificate, document or instrument delivered by the Acquired Company, and (ii) the failure of such Acquired Company to perform any agreements or covenants of such Acquired Company contained in this Agreement; provided, however, that such Acquired Company shall not be responsible for any Damages with respect to any such matters until the cumulative aggregate amount of such Damages exceeds \$25,000, in which event such Acquired Company shall then be liable for all such cumulative aggregate Damages, including the first \$25,000; and (y) Accessity shall indemnify and save harmless each of the Owners, the PEI Warrantholders and the Acquired Companies and their respective officers, directors, shareholders, successors and assigns from and against any Damages to the extent arising from, relating to or otherwise in respect of (i) the inaccuracy or breach of any representation or warranty of Accessity contained in this Agreement (as of the date hereof, or as of the Closing Date) or of any representation, warranty or statement made in any schedule, certificate, document or instrument delivered by Accessity, and (ii) the

63

failure of Accessity to perform any agreements or covenants of Accessity contained in this Agreement; provided, however, that Accessity shall not be responsible for any Damages with respect to any such matters until the cumulative aggregate amount of such Damages exceeds \$25,000, in which event Accessity shall then be liable for all such cumulative aggregate Damages, including the first \$25,000. As used in this Section 15.2, any Person entitled to indemnification pursuant to the provisions of this Section 15.2 shall be referred to herein as an "Indemnified Party" and any Person required to indemnify any Indemnified Party pursuant to the provisions of this Section 15.2 shall be referred to herein as an "Indemnifying Party."

(b) The Indemnified Parties shall notify each of the Indemnifying Parties within a reasonable period of time after becoming aware of, and shall provide to each of the Indemnifying Parties as soon as practicable thereafter

all information and documentation necessary to support and verify, any matter which the Indemnified Party shall have determined has given rise to a claim for indemnification hereunder, and the Indemnifying Parties shall be given reasonable access to all books and records in the possession or under the control of the Indemnified Party which the Indemnifying Parties reasonably determine to be related to such claim. The failure by any Indemnified Party to so notify the Indemnifying Parties or any indemnification claim hereunder shall not relieve the relevant Indemnifying Party from any liability which such Indemnifying Party may have to such Indemnified Party under this Agreement, except to the extent that such claim for indemnification involves the claim of a third party against the Indemnified Party and the Indemnifying Party shall have been actually prejudiced by such failure. If any Indemnifying Party does not notify the Indemnified Parties within 30 calendar days following receipt by it of such notice that such Indemnifying Parties disputes its liability to the Indemnified Parties under this Agreement, such claim specified by such Indemnified Party in such notice shall be conclusively deemed a liability of such Indemnifying Parties under this Agreement and such Indemnifying Parties shall pay the amount of such liability to such Indemnified Party on demand or, in the case of any notice in which the amount of the claim (or any portion thereof) is estimated, on such later date when the amount of such claim (or portion thereof) becomes finally determined. If an Indemnifying Party has timely disputed its liability with respect to such claim, as provided above, such Indemnifying Party and the Indemnified Party or Parties shall proceed in good faith to negotiate a resolution of such dispute and, if not resolved through such negotiations, such dispute shall be resolved by litigation in accordance with the terms of this Agreement.

(c) All claims for indemnity under this Article XV shall be paid by the Indemnifying Parties on demand in immediately available funds in U.S. dollars.

(d) With respect to any third party claim or action that could give rise to indemnity under this Agreement, the Indemnifying Party shall be entitled to assume the defense thereof with counsel satisfactory to the Indemnified Party, provided, however, that upon the request of the Indemnified Party, the Indemnifying Party provide reasonable evidence of its ability to perform its obligations under this Section 15.2. After notice from the Indemnifying Party to the Indemnified Party of their election so to assume the defense thereof, the Indemnifying Party shall not be liable to the Indemnified Party under the foregoing indemnity agreement for any legal or other expenses subsequently incurred by the Indemnified Party in connection with the defense thereof other than (i) those relating to investigation or the furnishing of documents or witnesses and (ii) all reasonable fees and expenses of separate counsel retained by such Indemnified Party if (A) the Indemnifying Party and the Indemnified Party shall have agreed to the retention of such counsel or (B) counsel to the Indemnified Party shall have

64

concluded reasonably that the representation of the Indemnifying Party and the Indemnified Party by the same counsel would be inappropriate due to actual or potential differing interests between them in the conduct of the defense of such action.

(e) Whenever the Indemnifying Party controls the defense of a third party claim, the Indemnifying Party may only settle or compromise the matter subject to indemnification without the consent of the Indemnified Party if such settlement includes a complete release of all Indemnified Parties as to the matters in dispute and relates solely to money damages. An Indemnified Party will not unreasonably withhold consent to any settlement or compromise that requires its consent.

(f) In the event the Indemnifying Party fails to timely defend, contest or otherwise protect the Indemnified Party against any such claim or suit, the Indemnified Party may, but will not be obligated to, depend, contest or otherwise protect against the same, and make any compromise or settlement thereof, and in such event, or in the case where the Indemnified Party jointly controls such claim or suit, the Indemnified Party shall be entitled to recover its costs thereof from the Indemnifying Party, including without limitation, attorneys' fees, disbursements and all amounts paid as a result of such claim or the compromise or settlement thereof.

(g) Each Indemnified Party shall cooperate and provide such assistance as the Indemnifying Party may reasonably request in connection with the defense of the matter subject to indemnification and in connection with recovering from any third parties amounts that the Indemnifying Party may pay or be required to pay by way of indemnification hereunder.

ARTICLE XVI TERMINATION

16.1 Termination. This Agreement may be terminated at any time prior to the Closing Date (whether before or after approval of this Agreement and the consummation of the transactions contemplated hereby by the shareholders of Accessity):

(a) by the mutual consent of Accessity and the Acquired Companies;

(b) by either Accessity or any of the Acquired Companies, upon written notice, if there has been a material misrepresentation or any breach on the part of a party hereto in the representations, warranties or covenants contained in this Agreement which is not cured within ten (10) business days after such breaching party (and Accessity and the Acquired Companies) has been notified of the intent to terminate this Agreement pursuant to this subsection (b);

(c) by either Accessity or the Acquired Companies if the Closing has not occurred on or before July 30, 2004 (the "Final Date");

(d) by Accessity, upon written notice, if the shareholders of Accessity shall not have approved the Agreement and the consummation of the transactions contemplated hereby (including, without limitation, with respect to the approval by the shareholders of Accessity, the appointment of the individuals identified in subsection (d) of Section 3.2 above to the Board of Directors of Accessity, the reincorporation of Accessity in the State of Delaware

65

referred to in Section 13.6 above, the sale or other disposition of the Accessity Subsidiaries (as defined below) to Barry Siegel referred to in Section 13.13 above, the adoption of an Amended and Restated 1995 Incentive Stock Plan of Accessity, in form and substance reasonably acceptable to the Acquired Companies, and the adoption of a new stock option plan as referred to in Section 13.16 above, in form and substance reasonably acceptable to the Acquired Companies) prior to the Closing Date;

(e) by Accessity, if the Board of Directors of Accessity shall have received a Superior Proposal;

(f) by either Accessity or any of the Acquired Companies, upon written notice, if Accessity or such Acquired Company, as the case may be, shall have determined in good faith not to proceed with the Closing on the basis of the results of its financial and legal due diligence investigation of the other parties to this Agreement (including, without limitation, satisfaction with matters related to Security Markets (as defined in Section 14.4 above) and litigation);

(g) by either Accessity or any of the Acquired Companies, if all the conditions for Closing shall not have been satisfied or waived on or before the Final Date other than as a result of a breach of this Agreement by the terminating party; or

(h) by either Accessity or any of the Acquired Companies, if a permanent injunction or other order by any federal or state court which would make illegal or otherwise restrain or prohibit the consummation of the Share Exchange or the other transactions contemplated hereby shall have been issued and shall have become final and nonappealable.

16.2 Effect of Termination. If this Agreement is terminated as expressly permitted under Section 16.1, such termination shall be the sole remedy and this Agreement shall forthwith become void (except for Section 11.2 and Section 17.3) and there shall be no liability on the part of the Acquired Companies, the Owners, the PEI Warrantholders, or Accessity or any of their respective Affiliates; provided, however, that if such termination shall result from the breach by a party hereto of its obligations under this Agreement, such party shall be fully liable for any and all damages, costs and expenses sustained or incurred by the other parties as a result of such breach; and

provided, further, that if such termination shall result from the receipt by Accessity of a Superior Proposal, such party shall be liable for and shall pay to the other the termination fee provided for in Section 16.3 below. If this Agreement is terminated without a Closing, the Acquired Companies shall return promptly to Accessity all documents, work papers and other materials of Accessity furnished or made available to the Owners, the PEI Warrantholders, the Acquired Companies or their respective Advisors, and all copies thereof, and no such information, documents, work papers or other materials received by the Owners, the PEI Warrantholders or the Acquired Companies shall be revealed to any third party or used for the advantage of the Owners, the PEI Warrantholders or the Acquired Companies or any other party; and Accessity shall return promptly to each of the Acquired Companies, respectively, all documents, work papers and other material of the Acquired Companies furnished or made available to Accessity or its Advisors, and all copies thereof, and no such information, documents, work papers or other materials received by Accessity shall be revealed to any third party or used for the advantage of Accessity or any other party.

66

16.3 Termination Fee. If Accessity, prior to termination of this Agreement, receives any Superior Proposal, and this Agreement is thereafter terminated pursuant to Section 16.1(e) above as a result of the receipt of such Superior Proposal, then Accessity shall promptly pay to each of the Acquired Companies the reasonable fees and expenses of each such Acquired Company incurred by it prior to termination, up to a maximum of \$150,000 for all of the Acquired Companies (which, if the combined aggregate amount of such fees and expenses exceeds \$150,000 for all of the Acquired Companies, shall be allocated to each Acquired Company based on proportion that the aggregate number of Accessity Exchange Shares that the Owners of such Acquired Company would receive if the Share Exchange were consummated pursuant to the terms of this Agreement bears to the aggregate number of Accessity Exchange Shares that all of the Owners of all of the Acquired Companies would receive if the Share Exchange were consummated pursuant to the terms of this Agreement, provided, that if such allocation for any Acquired Company exceeds the amount of actual fees and expenses of such Acquired Company incurred by it prior to termination, the excess shall be allocated to the other Acquired Companies and divided based on the relative number of Accessity Exchange Shares that the Owners of the other Acquired Companies would receive if the Share Exchange were consummated pursuant to the terms of this Agreement). Notwithstanding anything to the contrary, the foregoing termination fee, if paid in the full amount set forth above, shall constitute liquidated damages and shall constitute the non-terminating party's sole remedy.

> ARTICLE XVII MISCELLANEOUS PROVISIONS

17.1 Public Announcements. Upon the execution of this Agreement by all parties, Accessity, PEI, Kinergy and Reenergy promptly will issue a joint press release approved by Accessity and the Acquired Companies announcing the execution of this Agreement. Thereafter, Accessity may issue such press releases, and make such other disclosures regarding the proposed Share Exchange, as it determines (after consultation with legal counsel and after having given PEI, Kinergy and Reenergy and their respective legal counsel the opportunity to review and comment on such disclosure) are required under applicable state and federal securities laws or the rules and regulations of the NASD. Subject to the foregoing, except as the Acquired Companies and Accessity shall authorize in writing, the parties hereto shall not, and shall cause their respective officers, directors, employees, Affiliates and Advisors not to, disclose any matter or matters relating to this transaction to any person not an officer, director, employee, Affiliate or Advisor of such party.

17.2 Amendment; Waiver. Neither this Agreement, nor any of the terms or provisions hereof, may be amended, modified, supplemented or waived, except by a written instrument signed by the parties hereto (or, in the case of a waiver, by the party granting such waiver). No waiver of any of the provisions of this Agreement shall be deemed or shall constitute a waiver of any other provision hereof (whether or not similar), nor shall such waiver constitute a continuing waiver. No failure of either party hereto to insist upon strict compliance by the other party with any obligation, covenant, agreement or condition contained in this Agreement shall operate as a waiver of, or estoppel with respect to, any subsequent or other failure. Whenever this Agreement requires or permits consent by or on behalf of any party hereto, such consent shall be given in a manner consistent with the requirements for a waiver of compliance as set forth in this Section 17.2. For purposes of this Section 17.2, each of the PEI Shareholders and each of the PEI Warrantholders hereby appoints Ryan Turner as his, her or its agent and attorney-in-fact to make and execute and any all such amendments, modifications,

67

supplements and waivers and hereby acknowledge and agree that a decision by Ryan Turner shall be final, binding and conclusive on such PEI Shareholder or PEI Warrantholder, as the case may be, and that Accessity, Kinergy, Reenergy and the other Owners may rely upon any act, decision, consent or instruction of Ryan Turner.

17.3 Fees and Expenses. Except as otherwise provided in this Agreement, each of the parties hereto shall bear and pay its own costs and expenses incurred in connection with the origin, preparation, negotiation, execution and delivery of this Agreement and the agreements, instruments, documents and transactions referred to in or contemplated by this Agreement (whether or not such transactions are consummated) including, without limitation, any fees, expenses or commissions of any of its Advisors, attorneys, accountants, agents, finders or brokers. Accessity shall indemnify the Owners and the PEI Warrantholders against any claims of third parties for any brokerage, finder's, agent's or similar fees or commissions in connection with the transactions contemplated hereby insofar as such claims are alleged to be based on arrangements or contacts made by, to or with Accessity or its Advisors or representatives. The Acquired Companies shall indemnify Accessity against all such claims insofar as they are alleged to be based on arrangements or contacts made by, to or with any of the Owners, the PEI Warrantholders or the Acquired Companies or their respective Advisors or representatives.

17.4 Notices.

(a) All notices, requests, demands and other communications required or permitted under this Agreement shall be in writing (including telefax, telegraphic, telex or cable communication) and mailed, telefaxed, telegraphed, telexed, cabled or delivered:

(i) If to the Owners or the PEI Warrantholders, to the respective address for each such Owner or PEI Warrantholder set forth on the signature pages hereto.

(ii) If to PEI, to:

Pacific Ethanol, Inc. 5711 N. West Avenue Fresno, CA 93711 Attn: Jeff Manternach Facsimile no.: (559) 435-1478

with a copy to:

Rutan & Tucker, LLP 611 Anton Blvd., 14th Floor Costa Mesa, California 92626 Attn: Larry A. Cerutti, Esq. Facsimile no.: (714) 546-9035

68

(iii) If to Kinergy, to:

Kinergy Marketing, LLC 1260 Lake Blvd., Suite 225 Davis, CA 95616 Attn: Neil Koehler Facsimile no.: (530) 750-3019

(iv) If to Reenergy, to:

Reenergy, LLC 225 SE 59th Avenue Portland, OR 97215 Attn: Tom Koehler Facsimile no.: (530) 226-7917

(v) If to Accessity, to:

Accessity Corp. 12514 West Atlantic Blvd. Coral Springs, FL 33071 Attn: Barry Siegel Facsimile no.: (954) 752-6544

with a copy to: Meritz & Muenz, LLP Three Hughes Place Dix Hills, NY 11746 Attn: Lawrence A. Muenz, Esq. Facsimile no.: (631) 242-6715

(b) All notices and other communications required or permitted under this Agreement which are addressed as provided in this Section 17.4 (i) if delivered personally against proper receipt or by confirmed telefax or telex, shall be effective upon delivery and (ii) if delivered (A) by certified or registered mail with postage prepaid, or (B) by Federal Express or similar courier service with courier fees paid by the sender upon receipt. Either party may from time to time change its address for the purpose of notices to that party by a similar notice specifying a new address, but no such change shall be deemed to have been given until it is actually received by the party sought to be charged with its contents.

17.5 Assignment. This Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns, but neither this Agreement nor any of the rights, interests or obligations hereunder may be assigned by any party hereto without the prior written consent of the other parties; provided, however, that Accessity may assign its rights and obligations under this Agreement to any entity who by merger, consolidation, purchase or sale subsequently becomes an Affiliate without the prior consent of the Owners or any of the Acquired Companies. Any assignment which contravenes this Section 17.5 shall be void ab initio.

69

17.6 Governing Law. This Agreement and the legal relations between the parties hereto shall be governed by and construed in accordance with (i) with respect to matters arising prior to the Closing Date, the internal laws of the State of New York, without giving effect to the conflicts of laws principles thereof, and (ii) with respect to matters arising on and after the Closing Date, the internal laws of the State of Delaware, without giving effect to the conflicts of laws principles thereof.

17.7 Counterparts. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, but all of which taken together shall constitute one and the same instrument. This Agreement will become binding when one or more counterparts hereof, individually or taken together, will bear the signatures of all the parties reflected hereon as signatories.

17.8 Headings. The headings contained in this Agreement are for convenience of reference only and shall not constitute a part hereof or define, limit or otherwise affect the meaning of any of the terms or provisions hereof.

17.9 Entire Agreement. This Agreement (which defined term includes the Schedules and Exhibits to this Agreement) embodies the entire agreement and understanding among the parties hereto with respect to the subject matter of this Agreement and supersedes all prior agreements, commitments, arrangements, negotiations or understandings, whether oral or written, between the parties with respect thereto. There are no agreements, covenants, undertakings, representations or warranties with respect to the subject matter of this Agreement other than those expressly set forth or referred to herein. This is an integrated agreement.

17.10 Severability. Each term and provision of this Agreement

constitutes a separate and distinct undertaking, covenant, term and/or provision hereof. In the event that any term or provision of this Agreement shall be determined to be unenforceable, invalid or illegal in any respect, such unenforceability, invalidity or illegality shall not affect any other term or provision of this Agreement, but this Agreement shall be construed as if such unenforceable, invalid or illegal term or provision had never been contained herein. Moreover, if any term or provision of this Agreement shall for any reason be held to be excessively broad as to time, duration, activity or subject, it shall be construed, by limiting and reducing it, so as to be enforceable to the extent permitted under applicable law as it shall then exist.

17.11 Other Remedies. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a party will be deemed cumulative with and not exclusive of any other remedy conferred hereby or by law on such party, and the exercise of any one remedy will not preclude the exercise of any other.

17.12 Absence of Third Party Rights. No provisions of this Agreement are intended, nor will be interpreted, to provide or create any third party beneficiary rights or any other rights of any kind in any client, customer, Affiliate, shareholders, or partner of any party hereto or any other person or entity unless specifically provided otherwise herein, and, except as so provided, all provisions hereof will be personal solely between the parties to this Agreement.

17.13 Construction of Agreement. The Agreement has been negotiated by the respective parties hereto and their attorneys and the language hereof will not be construed for or

70

against either party. A reference to a Section or an Exhibit will mean a Section in, or Exhibit to, this Agreement unless otherwise explicitly set forth herein.

17.14 Attorneys' Fees. In the event that any action or proceeding is commenced by any party hereto for the purpose of enforcing any provision of this Agreement, the parties to such action or proceeding may receive as part of any award, judgment, decision or other resolution of such action or proceeding their costs and reasonable attorneys' fees as determined by the person or body making such award, judgment, decision or resolution. Should any claim hereunder be settled short of the commencement of any such action or proceeding, the parties in such settlement shall be entitled to include as part of the damages alleged to have been incurred reasonable costs of attorneys or other professionals in investigation or counseling on such claim.

17.15 No Joint Venture. Nothing contained in this Agreement will be deemed or construed as creating a joint venture or partnership between any of the parties hereto. Except to the extent provided in Article XVI and Section 17.2 above and Section 17.16 below, no party is by virtue of this Agreement authorized as an agent, employee or legal representative of any other party and no party will have the power or authority to control the activities and operations of any other or to bind or commit any other. The status of the parties hereto is, and at all times will continue to be, that of independent contractors with respect to each other.

17.16 Further Assurances. On and after the Closing Date, the Owners shall take such other steps and actions as may be necessary to put Accessity in actual possession and operating control of PEI, Kinergy and Reenergy as may be reasonably requested by Accessity. Each party agrees to cooperate fully with the other parties and to execute such further instruments, documents and agreements and to give such further written assurances as may be reasonably requested by any other party to evidence and reflect the transactions described herein and contemplated hereby and to carry into effect the intents and purposes of this Agreement.

[SIGNATURES CONTAINED ON THE FOLLOWING PAGES]

 $$\rm IN$ WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the day and year first above written.

"ACCESSITY": ACCESSITY CORP. _____ By: _____ Print Name: _____ Title: _____ PACIFIC ETHANOL, INC. "ACQUIRED COMPANIES": _____ By: -----Print Name: _____ Title: _____ KINERGY MARKETING, LLC By: _____ Print Name: _____ Title: _____ REENERGY, LLC By: -----Print Name: _____ Title: _____ 72 PEI SHAREHOLDER SIGNATURE PAGE TO SHARE EXCHANGE AGREEMENT

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the day and year first above written.

By:		_
Name:		_
Title:		_
Name of PEI Shareholder	(if other than	individual):
		-

73

PEI WARRANTHOLDER SIGNATURE PAGE TO SHARE EXCHANGE AGREEMENT

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the day and year first above written.

Ву:_____

Name:_____

Title:_____

Name of PEI Warrantholder (if other than individual):

74

KINERGY MEMBER SIGNATURE PAGE TO SHARE EXCHANGE AGREEMENT

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the day and year first above written.

Neil Koehler

REENERGY MEMBER SIGNATURE PAGE TO SHARE EXCHANGE AGREEMENT

75

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the day and year first above written.

By:							-
Name:	:						-
Title	≥:						-
Name	of	Reenergy	Member	(if	other	than	individual):
							-

76

EXHIBIT A

Exchange Ratios

PEI Exchange Ratio:

1.0 (1-to-1)

Kinergy Exchange Ratio:

18,750 (18,750 shares per 1% Kinergy Interest)

Reenergy Exchange Ratio:

21,250 (21,250 shares per 1% Reenergy Interest)

EXHIBIT B

PEI Shareholders and Accessity Exchange Shares to be Received

Name of Shareholder	No. of Shares of PEI Stock Owned	4
William Jones	4,800,000	4,800,000
SC Fuels, Inc.	1,500,000	1,500,000
Ryan Turner	1,243,333	1,243,333
Jeannine Campos	6,667	6 , 667
Tony Campos	100,000	100,000
Cagan McAfee Capital Partners, LLC	1,000,000	1,000,000
Andrea Jones	1,350,000	1,350,000
Bradley Rotter	66,666.67	66 , 667
Turner Family Trust	20,000	20,000
Clark and Patti Abramson	20,000	20,000
Rogene Scott Turner, as Trustee for the Rogene Scott Turner Trust dated 9/10/91	3,000	3,000
W. Denman Zirkle	126,666	126,666
Luise Bettina Zirkle Garcia	16,667	16,667
Sigrid Anne Zirkle Carroll	16,667	16,667
William Wade Zirkle	16,667	16,667
Anne Pendleton Zirkle	3,333	3,333
Janet Dumper	5,000	5,000
Robert Dumper	5,000	5,000
B-	1	
Joseph Childrey	40,000	40,000
Micaela Zirkle Shaugnesy	20,000	20,000
Illiquid Assets Trust, FBO Peter H. Koehler (Jon W. Nickel, Trustee)	16,666.67	16 , 667
Roger Manternach	16,666.67	16,667
Barry Fay	35,000	35,000
James and Bernice Campbell	2,000	2,000
John Burke	5,000	5,000
McDonald Investments, Inc., FBO Michael Frangopoulos	10,000	10,000
Samuel Kozasky	2,000	2,000
Dermot Fallon	8,000	8,000

James Burkdoll	4,000	4,000
Howard Kaplan	10,000	10,000
Jay Scott	8,000	8,000
Gregg Mullery	10,000	10,000
Richard DeSousa	4,000	4,000
R. V. Edwards, Jr.	17,000	17,000
Louis Lyras	7,000	7,000
John and Anne Fallon	8,000	8,000
Venkata Kollipara (as Custodian for Priya Kollipara)	10,000	10,000
	B-2	
Venkata Kollipara (as Custodian for Puneet Kollipara)	10,000	10,000
Robert and Rosalie Dettle Living Trust (Robert E. Dettle, Trustee)	10,000	10,000
Michael Kemp	14,000	14,000
Daniel Yates	10,000	10,000
Alex and Lisa Jachno	10,000	10,000
Barbara LaCosse	3,400	3,400
Lakshmana Madala	13,400	13,400
Edward Muransky	7,000	7,000
Alex and Luba Jachno	2,000	2,000
Katharine Moore	7,000	7,000
Armen Arzoomanian	10,000	10,000
David DeSilva	35,000	35,000
Steve Elefter	4,000	4,000
Tom McFaul	3,400	3,400
Elizabeth Reed	17,000	17,000
Venkata Kollipara	10,000	10,000
Kennon White	7,000	7,000
Teixeira Investments, L.P.	34,000	34,000
David Jessen	4,000	4,000
Henry Mauz	8,000	8,000
Lyles Diversified, Inc.	1,000,000	1,000,000
	B-3	
Linden Growth Partners	250,000	250,000
Dan Hollis	250,000	250,000
TOTAL	12,252,200	12,252,200

 $\ensuremath{\mathsf{PEI}}$ Warrantholders and Warrants to Acquire Accessity Common Stock

B-4

Name of PE Warrantholder	No. of Shares Subject to Accessity Replacement Warrants	
Cagan-McAfee Capital Partners	14,167	\$1.50
Prima Capital Group, Inc.	28,320	\$1.50
Frank Siefert	1,000	\$1.50
Cagan-McAfee Capital Partners	50,000	\$2.00
Jeffrey Manternach	25,000	\$0.01
Liviakis Group	1,150,000	\$0.0001
TOTAL	1,268,487	\$0.0001-\$2.00

Convertible Debt

* Lyles Diversified, Inc. 1,000,000* 1,000,000*

* Lyles Diversified, Inc. ("LDI") may receive up to 1,000,000 shares of PEI Stock pursuant to the conversion of a portion of the currently outstanding debt owed by PEI to LDI, which conversion is at the option of LDI. Accordingly, the number of Accessity Exchange Shares to be received by LDI shall be equal to the product of (i) the number of shares of PEI Stock received upon conversion (at a conversion rate of \$1.50 per share) of such portion of such debt, multiplied by (ii) the PEI Exchange Ratio. The parties acknowledge and agree that Accessity Exchange Shares will be directly issued to LDI (without any preceding issuance of shares of PEI Stock) if such conversion occurs after the Closing Date.

B-5

	Kinergy	No. of Accessity
Name of Member	Percentage Interest	Exchange Shares
Neil Koehler	100%	1,875,000

C-1

EXHIBIT D

Reenergy Members and Accessity Exchange Shares to be Received

Name of Member	Reenergy Percentage Interest	No. of Accessity Exchange Shares
Kinergy Resources, LLC	23.5%	499,375
Kent Kaulfuss	23.5%	499,375
Flin-Mac, Inc.	23.5%	499,375
Tom Koehler	29.5%	626,875

D-1

EXHIBIT E

Form of Spousal Consent

The undersigned, as the spouse of ______, hereby signs and consents to the foregoing "Share Exchange Agreement" for the purpose of binding and consenting to the commitment of the marital community property of ______ and the undersigned as assets available for the satisfaction of the undersigned's obligations under the foregoing "Share Exchange Agreement" and any other documents, agreements or instruments referenced therein, and for the purpose of acknowledging and agreeing that (i) no consent of the undersigned shall be required for any future modification of the foregoing "Share Exchange Agreement" or any other document, agreement or instrument referenced therein, and (ii) any community property of

______ and the undersigned which shall hereafter be transmuted into separate property of the undersigned shall be available for satisfaction of obligations under the foregoing "Share Exchange Agreement" and any other documents, agreements or instruments referenced therein.

(Signature)

Name:_____

By:___

(Please print)

E-1

SCHEDULE 13.9 - CASH BALANCE

Reductions and adjustments to Cash Balance: (subject to further review, revision and finalization by Accessity and the Acquired Companies):

- For each month running from March 31, 2004, the Cash Balance shall fall by approximately \$135,000 due to current levels of operating losses.
- Re-purchase of Series A Convertible Preferred Stock for \$350,000.
 Payment Acceleration-Upon learning of the transaction certain vendors may insist on being paid more rapidly prior to the closing because the control of the funds will be in the hands of other individuals with whom they have no relationship, or because they may no longer be viewed as continuing vendors.
- 4. Funding/payment of the arbitration legal costs against Presidion Solutions, Inc. and Mercator in the months of March and April 2004. Should significant legal demands occur thereafter that are not currently anticipated, these will also reduce cash.
- 5. Increase in expenses- This transaction itself results in expenses that were not contemplated in budgeting process and are not the normal part of on-going monthly losses. These include, but are not limited to, legal costs, due diligence trips and searches, proxy development costs, printing and mailing proxy, ADP solicitation for a shareholder vote, the formality of a shareholder meeting and auditor's review of the proxy.
- 6. Use of estimates/trade debt- The amount of cash used is an estimate based on expectations of cash inflows and outflows which cannot be exactly projected under normal conditions and particularly in light of these circumstances, and accordingly variations are to be assumed for estimates in losses and demand of trade debt. Further, some of the invested funds are in fixed income mutual funds whose values change slightly on a daily basis.

Not withstanding the forgoing, Accessity agrees to use its best efforts to control its cash use and that expenditures will be in the ordinary course of business of its on-going operations, or for the purpose of concluding this transaction.

AMENDMENT NO. 1 TO SHARE EXCHANGE AGREEMENT

THIS AMENDMENT NO. 1 TO SHARE EXCHANGE AGREEMENT (this "Amendment") is made and entered into as of July 29, 2004, by and among Accessity Corp., a New York corporation ("Accessity"); Pacific Ethanol, Inc., a California corporation ("PEI"); Kinergy Marketing, LLC, an Oregon limited liability company ("Kinergy"); Reenergy, LLC, a California limited liability company ("Reenergy," and together with PEI and Kinergy, the "Acquired Companies"); each of the shareholders of PEI identified on the signature pages hereof (collectively, the "PEI Shareholders"); each of the holders of options or warrants to acquire shares of common stock of PEI identified on the signature pages hereof (collectively, the "PEI Warrantholders"); each of the limited liability company members of Kinergy identified on the signature pages hereof (collectively, the "Kinergy Members"); each of the limited liability company members of Reenergy identified on the signature pages hereof (collectively, the

WHEREAS, Accessity, PEI, Kinergy, and Reenergy have executed a Share Exchange Agreement dated as of May 14, 2004 (the "Exchange Agreement"); and

WHEREAS, Accessity, PEI, Kinergy and Reenergy desire to amend certain provisions of the Share Exchange Agreement.

NOW THEREFORE, in consideration of the foregoing premises and the respective promises and agreements of the parties set forth herein, the parties hereto agree as follows:

 $\ 1.$ DEFINITIONS. Capitalized terms used herein and not otherwise defined herein shall have the respective meanings ascribed thereto in the Exchange Agreement.

2. AMENDMENTS.

(a) Section 2.5 of the Exchange Agreement is hereby amended by deleting in its entirety the next-to-last sentence of said Section 2.5 which reads: "The parties shall not take a position on any tax return inconsistent with this Section 2.5." and inserting in its place the following new next-to-last sentence of Section 2.5 which shall read in its entirety as follows:

> "The parties shall not take a position on any tax return inconsistent with this Section 2.5 and shall not take any action or omit to take any action which will or which could reasonably be expected to result in the disqualification of the Share Exchange as part of a unified plan for the exchange of stock entitled to non-recognition treatment under Section 351 of the Code."

(b) Section 4.8 of the Exchange Agreement is hereby amended by inserting the phrase "Except as set forth on Schedule 4.8," at the beginning of Section 4.8. $1\,$

(c) Section 6.8 of the Exchange Agreement is hereby amended by inserting the word "no" before the word "Actions" in the first line of said Section 6.8, so that the first portion of the first sentence of Section 6.8 up to the first comma therein reads:

"There are no Actions to which Kinergy is a party,"

(d) Section 8.8 of the Exchange Agreement is hereby amended by inserting the word "no" before the word "Actions" in the first line of said Section 8.8, so that the first portion of the first sentence of Section 8.8 up to the first comma therein reads:

"There are no Actions to which Reenergy is a party,"

(e) Section 11.11 of the Exchange Agreement is hereby amended by deleting the period at the end of Section 11.11 and adding the following additional language at the end of Section 11.11 as so modified:

"; provided, that, in lieu of preparing (pursuant to Section 11.7 above) and distributing the Owner Disclosure Document, the documents

required to be furnished pursuant to Rule 502(b)(ii) of Regulation D promulgated under the rules and regulations of the Securities Act of 1933, as amended, may instead be distributed."

(f) Subsection (a) of Section 3.2 of the Exchange Agreement is hereby amended by deleting said subsection in its entirety and inserting in its place the following new subsection (a) of Section 3.2, which shall read in its entirety as follows:

"(a) to each PEI Warrantholder, an Accessity Replacement Warrant evidencing such PEI Warrantholder's right to acquire the number of shares of Accessity Common Stock set forth opposite his or her name as set forth on Exhibit B and otherwise providing for the same terms and conditions as provided for in the PEI Warrants of such PEI Warrantholder; and, if the transfer agent is able to provide such stock certificates by the Closing Date or, if not, as soon as the transfer agent is able to provide such stock certificates after the Closing Date, to each PEI Shareholder, a stock certificate evidencing his or her ownership of the number of Accessity Exchange Shares set forth opposite his or her name as set forth on Exhibit B; to each Kinergy Member, a stock certificate evidencing his or her ownership of the number of Accessity Exchange Shares set forth opposite his or her name as set forth on Exhibit C; and to each Reenergy Member, a stock certificate evidencing his or her ownership of the number of Accessity Exchange Shares set forth opposite his or her name as set forth on Exhibit D;"

(g) Subsection (c) of Section 16.1 of the Exchange Agreement is hereby amended by deleting said subsection in its entirety and inserting in its place the following new subsection (c), which shall read in its entirety as follows:

2

"(c) by either Accessity or the Acquired Companies if the Closing has not occurred on or before October 29, 2004 (the "Final Date");"

(h) Section 16.1 of the Exchange Agreement is hereby further amended by deleting the word "or" set forth at the end of subsection (g) of Section 16.1, by deleting the period and inserting the word "; or" at the end of subsection (h) of Section 16.1, and by adding a new subsection (i) to Section 16.1 which shall read in its entirety as follows:

> "(i) by either Accessity or any of the Acquired Companies, if holders of one percent (1%) or more of the common stock of Accessity exercise their dissenters' rights in connection with the proposed reincorporation of Accessity in the State of Delaware as contemplated by Section 13.6 above."

(i) Exhibit B to the Exchange Agreement is hereby amended by deleting the information in the last row of the table entitled "PEI Shareholders and Accessity Exchange Shares to be Received" on Exhibit B that reads: "Dan Hollis 250,000 250,000" and inserting in its place the following new information:

"Barry Uphoff	12,500	12,500
Andrew Hoffman	25,000	25,000
Stephen George	12,500	12,500
Bradley Rotter	150,000	150,000
Michael Peterson	37,500	37,500
James George	12,500	12,500"

(j) Exhibit B to the Exchange Agreement is hereby further amended by deleting the name "Liviakis Group" set forth immediately above the TOTAL line in the table entitled "PEI Warrantholders and Warrants to Acquire Accessity Common Stock" on Exhibit B and inserting in its place the name "Liviakis Financial Communications, Inc."

(k) Section 4.2 of the Exchange Agreement is hereby amended by deleting the number "20,000,000" and replacing it with "30,000,000".

3. MISCELLANEOUS. Except as modified and amended pursuant to this Amendment, the Exchange Agreement shall remain in full force and effect. This Amendment may be executed in one or more counterparts, each of which shall be deemed an original, but all of which taken together shall constitute one and the same instrument. This Amendment will become binding when one or more counterparts hereof, individually or taken together, will bear the signatures of all the parties reflected hereon as signatories.

[SIGNATURES CONTAINED ON THE FOLLOWING PAGE]

3

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed as of the day and year first above written.

"ACCESSITY":

ACCESSITY CORP.

By:

Barry Siegel, Chairman and CEO

"ACQUIRED COMPANIES":

PACIFIC ETHANOL, INC.

By:

Ryan Turner, Director and $\ensuremath{\mathsf{COO}}$

KINERGY MARKETING, LLC

By:

Neil M. Koehler, President

REENERGY, LLC

Ву:

Frank R. Lindbloom, Member/Owner

4 PEI SHAREHOLDER AND PEI WARRANTHOLDER SIGNATURE PAGE TO AMENDMENT NO. 1 TO SHARE EXCHANGE AGREEMENT

Pursuant to the authority granted to the undersigned in Section 17.2 of the Exchange Agreement, by execution of this Amendment below by the undersigned, the PEI Shareholders and PEI Warrantholders have executed this Amendment as of the day and year first above written.

By:

Ryan Turner, Attorney-in-Fact 5 KINERGY MEMBER SIGNATURE PAGE TO AMENDMENT NO. 1 TO SHARE EXCHANGE AGREEMENT

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed as of the day and year first above written.

Neil M. Koehler

6 REENERGY MEMBER SIGNATURE PAGE TO AMENDMENT NO. 1 TO SHARE EXCHANGE AGREEMENT

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed as of the day and year first above written.

KINERGY RESOURCES, LLC

By:

Neil M. Koehler, Member

FLIN-MAC,	INC.
-----------	------

By: Frank R. Lindbloom, President

----- Kent Kaulfuss

Tom Koehler

7

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002 PURSUANT TO REGULATION SS.240.15D-14 AS PROMULAGATED BY THE SECURITIES AND EXCHANGE COMMISSION

In connection with the Quarterly Report of Accessity Corp. (the "Company") on Form 10-QSB for the period ended June 30, 2004, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Barry Siegel, Chairman of the Board, President and Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 pursuant to Regulation ss.240.15d-14 as promulgated by the Securities and Exchange Commission, that:

(1) I have reviewed the Report being filed;

(2) Based on my knowledge, the Report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by the Report;

(3) Based on my knowledge, the financial statements, and other financial information included in the Report, fairly present in all material respects the financial condition, results of operations and cash flows of the issuer as of, and for, the periods presented in the Report;

(4) The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-14 and 15d-14) for the Registrant and have:

a) designed such disclosure controls and procedures to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;

b) evaluated the effectiveness of the registrant's disclosure controls and procedures as of June 30, 2004 (the "Evaluation Date"); and

c) presented in this quarterly report our conclusions about the effectiveness of the disclosure controls and procedures based on our evaluation as of the Evaluation Date;

(5) The small business issuer's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the Registrant's auditors and the audit committee of the Registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the Registrant's ability to record, process, summarize and report financial information; and

(b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the Registrant's internal control over financial reporting.

(6) The registrant's other certifying officers and I have indicated in this quarterly report whether or not there were significant changes in internal controls or in other factors that could significantly affect internal controls subsequent to the date of our most recent evaluation, including any corrective actions with regard to significant deficiencies and material weaknesses.

/s/ Barry Siegel

Barry Siegel Chairman of the Board,

President and Chief Executive Officer CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002 PURSUANT TO REGULATION SS.240.15D-14 AS PROMULAGATED BY THE SECURITIES AND EXCHANGE COMMISSION

In connection with the Quarterly Report of Accessity Corp. (the "Company") on Form 10-QSB for the period ended June 30, 2004, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Philip Kart, Senior Vice President, Treasurer, Secretary and Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 pursuant to Regulation ss.240.15d-14 as promulgated by the Securities and Exchange Commission, that:

(1) I have reviewed the Report being filed;

(2) Based on my knowledge, the Report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by the Report;

(3) Based on my knowledge, the financial statements, and other financial information included in the Report, fairly present in all material respects the financial condition, results of operations and cash flows of the issuer as of, and for, the periods presented in the Report;

(4) The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-14 and 15d-14) for the Registrant and have:

a) designed such disclosure controls and procedures to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;

b) evaluated the effectiveness of the registrant's disclosure controls and procedures as of June 30, 2004 (the "Evaluation Date"); and

c) presented in this quarterly report our conclusions about the effectiveness of the disclosure controls and procedures based on our evaluation as of the Evaluation Date;

(5) The small business issuer's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the Registrant's auditors and the audit committee of the Registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the Registrant's ability to record, process, summarize and report financial information; and

(b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the Registrant's internal control over financial reporting.

(6) The registrant's other certifying officers and I have indicated in this quarterly report whether or not there were significant changes in internal controls or in other factors that could significantly affect internal controls subsequent to the date of our most recent evaluation, including any corrective actions with regard to significant deficiencies and material weaknesses.

By /s/ Philip Kart

Philip Kart Senior Vice President,

Secretary, Treasurer, and Chief Financial Officer

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Accessity Corp. (the "Company") on Form 10-QSB for the period ended June 30, 2004 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Barry Siegel, Chairman of the Board, President and Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- The Report containing the financial statements for the Company fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- The information contained in the Report fairly presents, in all material respects the financial condition and results of operations of the Company.

By /s/ Barry Siegel

Barry Siegel Chairman of the Board, President and Chief Executive Officer

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Accessity Corp. (the "Company") on Form 10-QSB for the period ended June 30, 2004, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Philip Kart, Senior Vice President, Treasurer, Secretary and Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- The Report containing the financial statements for the Company fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- The information contained in the Report fairly presents, in all material respects the financial condition and results of operations of the Company.

By /s/ Philip Kart

Philip Kart Senior Vice President, Treasurer, Secretary and Chief Financial Officer